## Flygtningenævnets baggrundsmateriale

Bilagsnr.:	517
Land:	Gaza-Vestbredden
Kilde:	Ministerie van Buitenlandse Zaken, Nederlandene
Titel:	General Country of origin information report Palestinian Territories
Udgivet:	1. april 2022
Optaget på baggrundsmaterialet:	2. august 2022

# General Country of origin information report Palestinian Territories

Datum April 2022

(Jeneral ( Ountry	/ of origin	i intormation	report Palestinian	Lerritories	l Anril 202

## Publication details

Location The Hague Prepared by DAF/AB

Disclaimer: The Dutch version of this report is leading. The Ministry of Foreign Affairs of the Netherlands cannot be held accountable for misinterpretations based on the English version of the report.

## Contents

	Publication details	2
	Contents	3
1	Political and security situation	7
1.1	Political situation	
1.1.1	General	7
1.1.2	Political groups	10
1.2	Security situation	
1.2.1	General	
1.2.2	Security services and armed groups	14
1.3	Living conditions	
1.3.1	General	
1.3.2	Role of UNRWA	
1.3.3	Humanitarian situation and reconstruction in Gaza after May 2021	
1.4	Freedom of movement	
1.4.1	Travel within the country	
1.4.2	Settlement in the Palestinian Territories	
1.4.3	Palestinians with an Israeli work permit	
1.4.4	Travelling abroad	
1.7.7	Travelling abroad	
2	Identity, nationality and documents	
2.1	Obligation to provide identification	
2.2	Identifying documents	
2.2.1	Israeli authorities	
2.2.2	Palestinian authorities	
2.2.3	Jordanian authorities	33
2.2.4	UNRWA	
2.2.5	UNHCR	34
2.3	Nationality	34
2.4	Registration	38
2.5	Right of residence	40
3	Human rights	43
3.1	Compliance and violations	
3.1.1	Freedom of speech	
3.1.2	Freedom of association and assembly	
3.1.3	Freedom of religion and belief	
3.1.4	Evictions	
3.1.5	Recruitment	
3.2	Position of specific groups	
3.2.1	Journalists, human rights advocates and political activists	
3.2.2		
	Collaborators	
3.2.3	Groups receiving the special attention of the Israeli authorities	
3.2.4	Conscientious objectors and deserters	
3.2.5	Ethnic and religious groups	
3.2.6	LGBTI people	
3.2.7	Women	
3.2.8	(Unaccompanied) minors	
3.3	Oversight and legal protection	
3.3.1	Judicial process	67

3.3.2	Arrests, custody and detentions	68
3.3.3	The death penalty	
3.3.4	Ill-treatment and torture	
3.3.5	Disappearances and kidnappings	76
3.3.6	Extrajudicial executions and killings	
3.3.7	'Blood feuds and honour-related violence'	
3.3.8	Corruption	
4	Refugees and displaced persons	81
4.1	Refugee flows	81
4.2	Internally displaced persons	81
5	Palestinians returning to their homeland	84
5.1	Re-entry	84
6	Appendices	85
6.1	Public resources used	
6.2	List of abbreviations	
6.3	Map of the Palestinian Territories	

## Introduction

This general country of origin information report was prepared on the basis of the questions and points of focus in the *Terms of Reference* (ToR) drawn up by the Ministry of Justice and Security on 21 July 2021. Together with the country of origin report, the ToR is available on the website of the Dutch Government. This report describes the situation in the Palestinian Territories insofar as it is relevant for the assessment of asylum applications made by persons originating from those territories and for decisions related to the return of rejected Palestinian asylum seekers. This country of origin information report addresses the themes and questions posed by the Ministry of Justice and Security and is a summary of the factual information that was obtained. It is not a policy document and it does not reflect the government's vision or policy in relation to a country or region or contain conclusions about immigration policy.

This country of origin information report is the first general country of origin information report for the Palestinian Territories. The report covers the period from August 2019 to 22 April 2022. This country of origin information report is a factual, neutral and objective representation of the findings during the period under consideration and does not include any policy recommendations.

The report was created on the basis of public and confidential sources, using carefully selected, analysed and verified information. Information from non-governmental organisations (NGOs), specialist literature, media coverage and the relevant government agencies was used to compile the report. Unless stated otherwise or when the facts are generally undisputed, the content in this report is based on multiple sources. The public sources that were consulted are listed in the appendices.

Some of the confidential information used was obtained during digital interviews with experts. Confidential information originating from other confidential sources was also used. The information obtained confidentially has chiefly been used to support and supplement information that is based on publicly available information. The confidential sources are marked 'confidential source' in the footnotes and are dated.

Chapter 1 deals with the recent political and security situation in the Palestinian Territories and the living conditions there. Chapter 2 discusses nationality legislation and includes an overview of documents in the Palestinian Territories. Chapter 3 describes developments related to the observance of human rights in the Palestinian Territories. Chapter 4 examines the situation of refugees and internally displaced persons in the Palestinian Territories. Finally, Chapter 5 focuses on the possible risks for people returning to the Palestinian Territories.

## 1 Political and security situation

#### 1.1 Political situation

#### 1.1.1 General

The Palestinian Territories consist of the West Bank, including East Jerusalem, and Gaza (see also 6.3). There are sixteen provinces (*muhafazat*) in the Palestinian Territories: eleven in the West Bank and five in Gaza. The population of the West Bank in July 2021 was almost 3 million and of Gaza approximately 2.1 million. The Palestinian Authority (PA), also known as the Palestinian National Authority, governs parts of the West Bank (see also *Balance of power* below). Fatah is the dominant party in the PA and is headed by Mahmoud Abbas. There is a president who was elected in 2005 (Mahmoud Abbas), a prime minister (Mohammad Shtayyeh - Fatah) and a parliament (Palestinian Legislative Council) that was non-functioning during the reporting period. The last parliamentary elections were held in 2006, and Mahmoud Abbas's term of office expired in 2009.

In practice, the PA rules over parts of the West Bank (without East Jerusalem) and Hamas<sup>5</sup> de facto in Gaza, but not all government tasks are performed by Hamas (for example, the PA deals with the issues of identity documents and the funding of healthcare and education). After Hamas won the majority of parliamentary seats in the 2006 elections, the Palestinian parliament stopped functioning and Fatah has governed the West Bank on the basis of presidential decrees and Hamas in Gaza through its MPs that were elected in 2006.<sup>6</sup>

The PA was founded in 1993 as an interim administrative organisation after the Oslo Accords<sup>7</sup> were signed by the Israeli government and the Palestine Liberation Organisation (PLO). The PA operates from the city of Ramallah and falls administratively under the PLO. In practice, actual power rests with President Abbas, who heads the PA and chairs the PLO Executive Committee.<sup>8</sup> The PLO is an umbrella organisation that has represented Palestinians living in the Palestinian Territories and beyond since 1964. The PLO is made up of three parts: the Palestinian National Council (PNC), a parliamentary structure of 740 members who are theoretically elected by the Palestinian population, but in practice are elected by the Executive Committee; the Central Council, a decision-making body of 124 members that meets in practice instead of the PNC; the Executive Committee, a body of eighteen

<sup>&</sup>lt;sup>1</sup> The Netherlands and the EU support the peace process between Israeli and Palestinian authorities, which focuses on a two-state solution. The borders as they stood before 4 June 1967 (before the Six-Day War) are the starting point for this. The future Palestinian state will consist of the Gaza Strip and the West Bank, including East Jerusalem. Dutch Government, Nederland ondersteunt vredesproces Midden-Oosten, consulted on 13 April 2022.

<sup>&</sup>lt;sup>2</sup> ECFR, *Mapping Palestinian Politics: Institutions, Governorates*, consulted on 10 January 2022. See also **Error!**Hyperlink reference not valid.

<sup>&</sup>lt;sup>3</sup> CIA The World Factbook, *West Bank, Gaza Strip*, 16 November 2021, consulted on 29 November 2021.

<sup>&</sup>lt;sup>4</sup> A secular nationalist party.

<sup>&</sup>lt;sup>5</sup> A nationalist Islamic party that has been designated as a terrorist organisation by the US and the EU.

<sup>&</sup>lt;sup>6</sup> ECFR, *Mapping Palestinian Politics: Occupied Palestinian Territory (OPT)*, consulted on 25 February 2022. See also **Error! Hyperlink reference not valid.** 

<sup>&</sup>lt;sup>7</sup> The agreements concluded as part of the peace process between the PLO and Israel in 1993-1995.

<sup>&</sup>lt;sup>8</sup> ECFR, *Mapping Palestinian Politics: Occupied Palestinian Territory (OPT)*, consulted on 25 February 2022. See also **Error! Hyperlink reference not valid.** 

members that elects the chairman of the PLO. In theory, the members are elected by the PNC.9

## Balance of power

After its annexation in 1980, East Jerusalem was brought into the Jerusalem administrative district by Israel, giving the Israeli authorities de facto authority there. The PA considers it to be part of the PA Governorate of Jerusalem. The Netherlands, the EU and most of the international community regard East Jerusalem as an occupied territory under international law. 10 The West Bank is divided into three areas in accordance with the Oslo Accords: Area A (17.2%), which is under the authority of the PA, Area B (23.8%), which is under the civilian administration of the PA and Israeli security control, and Area C (59%), which is under full Israeli control (see also 6.3). 11 The PA and the Israeli authorities exchange information about security risks in the West Bank - for example about the activities of Hamas there. The PA has no de facto control over armed groups in refugee camps or armed groups involved in organised crime. 12 Israeli troops can enter the West Bank (see also 3.3.2) and have authority over checkpoints there (see also 1.4.1). Hamas - an EU-designated terrorist organisation - has held power in Gaza since the PA was expelled from Gaza by Hamas in 2007. Israel exercises de facto control over the Gaza-Israel border crossings, while Egypt controls the Rafah crossing with Egypt.

The following is an overview of the main political developments in the Palestinian Territories during the reporting period.

## Great March of Return

From 30 March 2018 up to and including November 2019, Palestinians in Gaza protested against the Israeli closure of Gaza (see also 1.3.1) and for the 'right of return' to their (former) homes and land in present-day Israel. The protests were suspended in late 2019. During these protests, which took place along the fence on the Gaza-Israel border, it is estimated that between 214 and 223 Palestinians, 46 of whom were minors, were killed and thousands were injured, including around eight thousand by live ammunition and around 5,400 by rubber bullets<sup>13</sup> or by being hit by tear gas cans. During the same period, 1 Israeli soldier was killed and 7 other Israeli soldiers were injured.<sup>14</sup> The Israeli authorities said they were investigating incidents in which Palestinians had been killed.<sup>15</sup> In June 2021, 91 cases were still under investigation.<sup>16</sup> In 143 other cases, the investigation was completed and

<sup>&</sup>lt;sup>9</sup> ECFR, *Mapping Palestinian Politics: About Institutaions, Palestine Liberation Organisation (PLO),* consulted on 22 February 2022. See also **Error! Hyperlink reference not valid.** 

<sup>&</sup>lt;sup>10</sup> The Netherlands and the EU consider settlements in the West Bank to be contrary to international law. The Netherlands considers the annexation of East Jerusalem to be contrary to international law. In resolution 478 (1980), the UN Security Council ruled that the application of Israeli law and measures that alter the character of the city in East Jerusalem is null and void and is not recognised. Ministry of Foreign Affairs, Letter to Parliament: Ontwikkelingen in Jeruzalem, Gaza en Israël, 17 May 2021.

<sup>&</sup>lt;sup>11</sup> ECFR, Mapping Palestinian Politics: Geography, Occupied Palestinian Territory, consulted on 10 January 2022. See also Error! Hyperlink reference not valid.

<sup>&</sup>lt;sup>12</sup> Confidential source, 7 February 2022.

<sup>&</sup>lt;sup>13</sup> Rubber bullets consist of a steel ball with rubber around it, so there is still a risk of injury if someone is hit. Confidential source, 7 February 2022.

<sup>&</sup>lt;sup>14</sup> UN, Two Years On: People injured and traumatized during the "Great March of Return" are still struggling, 6 April 2020, consulted on 13 March 2022; B'Tselem and Palestinian Centre for Human Rights, Unwilling and Unable: Israel's Whitewashed Investigations of the Great March of Return Protests, December 2021, p. 2; B'Tselem, Summary of joint report with PCHR, December 2021, consulted on 17 December 2021

<sup>&</sup>lt;sup>15</sup> UN, Two Years On: People injured and traumatized during the "Great March of Return" are still struggling, 6 April 2020, consulted on 13 March 2022; UN High Commissioner for Human Rights, Ensuring accountability and justice for all violations of international law in the Occupied Palestinian Territory, including East Jerusalem, 15 February 2021, p. 7-10.

<sup>&</sup>lt;sup>16</sup> Haaretz, Months later, Israeli military backlog holds up probe into dozens of Gaza border deaths, 29 June 2021, consulted on 13 March 2022.

referred to various government agencies. In 95 of these cases, it was decided to no longer conduct a criminal investigation. In one case, the death of a minor, a penalty was imposed.<sup>17</sup> There were no further updates on these investigations during the reporting period.<sup>18</sup>

#### US Embassy

In 2018, the US government decided to recognise Jerusalem as Israel's capital, and in 2019 the US embassy was moved from Tel Aviv to Jerusalem. The embassy was merged with the already existing US consulate in East Jerusalem. Due to this relocation, Palestinians were temporarily unable to access the US consulate during the reporting period (see also 1.4.4).

#### Israeli-Palestinian conflict in May 2021

In April and May 2021, tensions in East Jerusalem led to violent clashes between Palestinian protesters and Israeli police and Israeli settlers, which was followed by armed conflict between Israel and Hamas. The tensions were sparked, among other things, by the cordoning off of the Damascus Gate, a meeting place during Ramadan, by Israeli police, the beating of a Jewish boy by Palestinians, images of which were circulated on social media, Israeli approval for the construction of new houses in a settlement and the threatened evictions of Palestinians in the Sheikh Jarrah neighbourhood in East Jerusalem (in protest against the impending evictions, demonstrations and sit-ins were organised in Jerusalem in April 2021; see also 3.1.4). During Ramadan protests in May 2021, violence broke out on the Temple Mount/Haram al-Sharif, near the Al Aqsa mosque in Jerusalem, resulting in hundreds of injuries. The celebration of Jerusalem Day - when segments of Israeli society commemorate the conquest of East Jerusalem - also sparked violence in East Jerusalem. Hamas and PFLP increased tensions during this period by threatening violence. Hamas began shelling Jerusalem, and later several other locations in Israel, on the evening of 9 May. Israel responded the next day by carrying out air strikes on Gaza (for further escalation of violence, see 1.2).20

## Palestinian organisations

In October 2021, the Israeli authorities accused six Palestinian organisations of having links with the Popular Front for the Liberation of Palestine (PFLP), an organisation registered as terrorist by the EU (see also 1.1.2.). It concerned the following organisations: Al Haq, a Palestinian human rights organisation; the Palestinian branch of Defence for Children International (DCIP); the Bisan research centre; the Union of Agricultural Workers (UAWC); Addameer, an organisation that assists Palestinian detainees in Israeli prisons and the Union of Palestinian Women Committees. They were placed by Israel on Israel's list of terrorist organisations, which also includes the PFLP. There was a great deal of commotion in the international community and (international) media about this move. The US, the EU and other countries, including the Netherlands, asked Israel to share the information on which the allegations were based. Palestinian and Israeli human rights organisations and organisations such as *Human Rights Watch* (HRW) and *Amnesty* 

<sup>&</sup>lt;sup>17</sup> B'Tselem and Palestinian Centre for Human Rights , Unwilling and Unable: Israel's Whitewashed Investigations of the Great March of Return Protests, December 2021.

<sup>&</sup>lt;sup>18</sup> Confidential source, 14 April 2022.

<sup>&</sup>lt;sup>19</sup> Middle East Monitor, US backs down from reopening consulate in occupied East Jerusalem, 16 December 2021, consulted on 26 February 2022; Council on Foreign Relations, Global Conflict Tracker: Israeli-Palestinian Conflict, 7 January 2022, consulted on 13 January 2022.

<sup>&</sup>lt;sup>20</sup> Ministry of Foreign Affairs, Letter to Parliament: Ontwikkelingen in Jeruzalem, Gaza en Israël, 17 May 2021; Council on Foreign Relations, Global Conflict Tracker: Israeli-Palestinian Conflict, 7 January 2022, consulted on 13 January 2022.

International considered the Israeli listing to be an attack on Palestinian civil society.<sup>21</sup> HRW voiced its suspicion that the Israeli company NSO had monitored a number of people from these organisations with Pegasus software (see also 3.1.1).<sup>22</sup>

#### Rapprochement and peace process

The Israeli-Palestinian peace process had come to a standstill during the reporting period. There were, however, several meetings between the PA and Israeli ministers, such as between Israeli Defence Minister Gantz and Palestinian Prime Minister Abbas.<sup>23</sup>

#### Elections

In January 2021, President Abbas announced elections for the presidency and parliament of the PA. All elections were cancelled in April 2021.<sup>24</sup> At the end of 2021, local elections were held in 154 small municipalities. The second phase of these local elections took place on 26 March 2022 in the larger municipalities of the West Bank. Hamas said it was boycotting these elections, which means that there were no elections in Gaza.<sup>25</sup> At a meeting of the PLO Central Council in February 2022, decisions were taken about new leadership positions in the PLO. These appointments were criticised by Hamas, Islamic Jihad and the Popular Front for the Liberation of Palestine (PFLP). They claimed that the appointment of Abbas loyalists further strengthened the power of the president.<sup>26</sup>

#### 1.1.2 Political groups

Various political parties and groups exist in the Palestinian Territories, sometimes with their own military branches. They are listed in the following overview.

The PLO includes the following political parties:

- Fatah (National Liberation Movement): a secular nationalist party founded in the 1950s, which is the dominant political party in the PA and PLO. Since the death of Yasser Arafat, the party has been led by Mahmoud Abbas.
  - Democratic Reform Bloc: a 2016 branch of Fatah led by Mohammed Dahlan, who is currently living in exile.
  - National Democratic Assembly: a branch of Fatah that split off in 2021 and is led by Nasser Kidwa (Yasser Arafat's cousin).
- Popular Front for the Liberation of Palestine (PFLP): a revolutionary socialist party founded in 1967, which has been designated a terrorist organisation by the EU and the US.
  - Democratic Front for the Liberation of Palestine (DFLP): a branch of the PFLP that split off in 1969 and is led by Nayyef Hawatmeh, who is based in Syria.

NRC, Israël zet Palestijnse sociale organisaties op de terreurlijst, 25 October 2021, consulted on 26 October 2021.
 HRW, Spyware used to hack Palestinian Rights Defenders, 8 November 2021, consulted on 25 November 2021; AP News, Report: NSO spyware found on 6 Palestinians activists' phones, 8 November 2021, consulted on 1 March 2022.

<sup>&</sup>lt;sup>23</sup> The Independent, Israel approves measures for Palestinians after leaders meet, 29 December 2022, consulted on 30 December 2021; Haaretz, Bennett rejects Peace Process with Abbas, focuses on Palestinians' practical needs, 1 September 2021, consulted on 13 March 2022.

<sup>&</sup>lt;sup>24</sup> BBC Monitoring, Palestinian president opens Central Council session, 7 February 2022, consulted on 8 February 2022; ECFR, Mapping Palestinian Politics: Elections: About Elections, consulted on 26 February 2022. See also Error! Hyperlink reference not valid.

<sup>&</sup>lt;sup>25</sup> BBC Monitoring, *Palestinian local elections in West Bank*, 11 December 2021, consulted on 12 December 2022; The Independent, *Some 400,000 Palestinians vote in rare municipal elections*, 11 December 2021, consulted on 12 December 2022; BBC Monitoring, *Palestinians hold second round of local elections*, 27 March 2022, consulted on 28 March 2022.

<sup>&</sup>lt;sup>26</sup> Al Jazeera, Abbas accused of power grab after Palestinian appointments, 10 February 2022, consulted on 11 February 2022.

- Palestine Democratic Union (FIDA): a branch of the DFLP with a social democratic character that split off in 1990, led by Saleh Ra'fat.
- Palestinian People's Party (PPP): a traditionally communist party formed in 1982.
- Palestine Liberation Front (PLF): a party founded in 1977 and currently led by Wasel Abu Yousef and designated a terrorist organisation by the US.
- Palestine Popular Struggle Front (PPSF): led by Ahmad Majdalani, a close ally of Mahmoud Abbas.
- Popular Front for the Liberation of Palestine-General Command (PFLP-GC): a branch of the PFLP that split off in 2016, led by Ahmad Jibril from within Syria and Lebanon until July 2021 and designated a terrorist organisation by the US and EU.<sup>27</sup>

#### Outside of the PLO, the following parties exist:

- Hamas (Islamic Resistance Movement): a nationalist Islamic movement founded in 1987 by Palestinian members of the Muslim Brotherhood. Hamas is classified as a terrorist organisation by the US and EU, among others. After a power struggle with Fatah, Hamas has governed Gaza since 2007.
- Palestinian Islamic Jihad (PIJ): a party that was formed in 1982 through the Palestinian Muslim Brotherhood network and is funded by Iran. The PIJ is critical of the PA but focuses on military confrontation with Israel rather than on politics. The party is currently led by Ziyad al-Nakhalah and is classified as a terrorist organisation by the US and EU, among others.
- Popular Resistance Committees (PRC): a grouping of armed factions in Gaza led by Yasser Shashnyeh. The armed wing of the party are the al-Nasser Salah al-Din Brigades. The PRC is the second largest armed group in Gaza after Hamas and PIJ and is a close ally of these two parties. Former members of the PFLP are said to have joined the PRC. The PRC does not focus on politics but on military confrontation with Israel and sometimes with Fatah.
- Al-Mubadara (Palestinian National Initiative PNI): is a social democratic and non-violent movement that was formed in the 1990s in opposition to the Oslo Accords. Mustafa Barghouti is the current leader of the PNI.
- Third Way: a centrist party founded in 2006 that aims to provide an alternative to the two-party system of Fatah and Hamas, currently represented by Hanan Ashrawi.
- Hizb ut-Tahrir: a global Islamic movement that was founded in Jerusalem in 1953. Hizb ut-Tahrir does not involve itself in politics but seeks to rebuild an Islamic caliphate through non-violent action. It is particularly popular among Palestinian university students.<sup>28</sup>

#### 1.2 Security situation

#### 1.2.1 General

In the Palestinian Territories, there is both targeted and indiscriminate violence by various Palestinian armed factions, by the Palestinian security forces and by Israeli

<sup>&</sup>lt;sup>27</sup> ECFR, Mapping Palestinian Politics: About Political Parties, PLO Groups, consulted on 11 January 2022. See also Error! Hyperlink reference not valid.

<sup>&</sup>lt;sup>28</sup> ECFR, Mapping Palestinian Politics: About Political Parties, Non PLO Groups, consulted on 11 January 2022. See also Error! Hyperlink reference not valid.

troops. This violence is causing civilian casualties, including in the refugee camps where Palestinian militants often reside. During the reporting period (a period of over 2.5 years), in the Palestinian Territories a total of 465 Palestinian deaths (321 in Gaza and 140 in the West Bank) were registered by UN OCHA in connection with the Israeli-Palestinian conflict.<sup>29</sup> In the ACLED database, 588 deaths in the Palestinian Territories were recorded during the reporting period. The ACLED figures also include 128 civilian casualties among the total of 256 people killed during the May 2021 conflict (see below). They also include part of the total of 223 fatalities during the Great March of Return protests between March 2018 and November 2019 (see also 1.1.1).30 According to figures from B'Tselem, 12 Israeli civilians and 4 members of the Israeli security services were killed by Palestinians during the reporting period.<sup>31</sup> It is not clear which unit of the Israeli security services this involved. UN OCHA registered 21 Israeli casualties linked to the Israeli-Palestinian conflict during the reporting period, of which 14 were in Israel and 7 in the West Bank.32

In addition, there were deaths and injuries during the reporting period that were

- Israel drone and air strikes on Gaza<sup>33</sup> in response to Hamas rockets fired on Israel<sup>34</sup>
- Fighting between Palestinian armed groups (including Hamas and PIJ) and Israeli forces<sup>35</sup>;
- Fighting between armed Palestinian (clan) groups;
- Fighting between armed Palestinians in refugee camps (groups unknown)<sup>36</sup>;
- Use of violence against Palestinian civilians/protesters by the Palestinian security services and Israeli police/troops (see also 3.1.1);
- Explosives fired or accidentally detonated by Palestinian armed groups in Gaza;
- Violence by Israeli settlers against Palestinian civilians and violence by Palestinians against Israeli settlers in the West Bank.<sup>37</sup>

## Violent incidents in Gaza in May 2021<sup>38</sup>

The violent incidents and protests in Jerusalem and the West Bank on 9 May 2021 prompted Hamas and Palestinian Islamic Jihad (PIJ) to fire rockets at Israel. Israel responded the next day by launching air strikes on Gaza, after which more Hamas rockets were fired at Israel. Some of the rockets from Gaza did not reach Israel and fell in Gaza. The following week, Israel targeted Hamas and PIJ targets with air strikes but also hit homes, an apartment building with media offices and refugee and healthcare facilities. Meanwhile, protests in East Jerusalem spread to several cities in Israel and the Palestinian Territories. The rocket and air strikes continued until a

<sup>&</sup>lt;sup>29</sup> OCHA, Data on casualties, consulted on 21 April 2022. See also Error! Hyperlink reference not valid.

<sup>&</sup>lt;sup>30</sup> ACLED, Dashboard Gaza and West Bank, 2022, consulted on 21 April 2021.

<sup>&</sup>lt;sup>31</sup> B'Tselem, Fatalities: All data, consulted on 21 April 2022. See also Error! Hyperlink reference not valid.

<sup>32</sup> OCHA, Data on casualties, consulted on 21 April 2022. See also Error! Hyperlink reference not valid.

<sup>33</sup> ACLED, Dashboard Gaza and West Bank, 2022, consulted on 12 January 2022.

<sup>34</sup> Reuters, Israel-Gaza conflict rages on despite U.S. regional diplomacy, 17 May 2021, consulted on 13 January 2022; Council on Foreign Relations, Global Conflict Tracker: Israeli-Palestinian Conflict, 7 January 2022, consulted on 13 January 2022; ACLED, *Dashboard Gaza and West Bank*, 2022, consulted on 12 January 2022.

35 ACLED, *Dashboard Gaza and West Bank*, 2022, consulted on 12 January 2022.

<sup>&</sup>lt;sup>36</sup> ACLED, Dashboard Gaza and West Bank, 2022, consulted on 12 January 2022; Confidential source, 7 February 2022

<sup>&</sup>lt;sup>37</sup> ACLED, *Dashboard Gaza and West Bank*, 2022, consulted on 12 January 2022.

<sup>&</sup>lt;sup>38</sup> There were previous conflicts in Gaza in 2008-2009, 2012, and 2014. AP News, *The deepening costs of 4 Gaza* wars in 13 years, in 4 charts, 25 August 2021, consulted on 13 March 2022.

ceasefire was concluded on 21 May 2021. During this period, an estimated 253 Palestinians were killed, including 129 civilians, and around two thousand Palestinians were injured. In Israel, at least twelve people were killed and 710 injured.<sup>39</sup> Shooting incidents between Israeli troops and Hamas on the border between Israel and Gaza continued.<sup>40</sup>

#### Violent incidents in the West Bank and East Jerusalem

Hundreds of violent incidents between Jewish settlers and Palestinians have been recorded in the West Bank since early 2020 and continued to increase during the reporting period.<sup>41</sup> Violence by Jewish settlers includes, for example, the use of pepper spray, throwing stones, damaging cars and stealing and/or damaging olive trees and orchards, including crops and agricultural tools. 42 OCHA data states that 5 Palestinians were killed and 383 Palestinians injured (until 20 March 2022) by settlers in the West Bank between August 2019 and 15 March 2022. In the period August 2019 to 21 November 2021, 4 Israeli settlers were killed by Palestinians and 189 were injured by Palestinians (up to 20 March 2022).<sup>43</sup> According to one source, most Palestinians are injured by Israeli troops and not by the settlers themselves during settler incidents. Most of the Palestinians injured require medical attention after inhaling tear gas. 44 In the first ten months of 2021, 108 attacks on Palestinians and 302 attacks on Palestinian properties are said to have been carried out by settlers. In 2019, 335 cases were documented, while this involved 358 cases in 2020. In 2021, four Palestinians were killed by settlers. 45 There were also attacks against settlers. Some settlers were said to have been injured and dozens of Israeli cars damaged when Palestinians threw stones in November 2021.46 These examples are illustrative but can by no means be regarded as exhaustive.

In February 2022, there were almost daily clashes between Israeli police, Palestinians and Israeli settlers in the Sheikh Jarrah neighbourhood of East Jerusalem. This was due to tensions surrounding evictions and the opening of an Israeli parliamentarian's office near the home of a Palestinian family threatened with eviction.<sup>47</sup>

In March and April 2022, violence flared up between Palestinians and the Israeli security services. Palestinians and Arab nationals from Israel carried out terrorist attacks in Israel in which 21 people (including the perpetrators) were killed.<sup>48</sup> This

<sup>&</sup>lt;sup>39</sup> Reuters, Israel-Gaza conflict rages on despite U.S. regional diplomacy, 17 May 2021, consulted on 13 January 2022; Council on Foreign Relations, Global Conflict Tracker: Israeli-Palestinian Conflict, 7 January 2022, consulted on 13 January 2022.

<sup>&</sup>lt;sup>40</sup> BBC Monitoring, Three Palestinians wounded by Israeli tank fire on Gaza, 29 December 2021, consulted on 14 January 2022.

<sup>&</sup>lt;sup>41</sup> The Independent, Israel's security forces are 'complicit' in drastic surge in settler violence, report finds, 21 July 2021, consulted on 22 July 2021; Al Jazeera, 450 settler attacks on Palestinians in two years: Israeli NGO, 14 November 2021, consulted on 15 November 2021; OCHA oPt, Protection of Civilians Report | 22 February - 7 March 2022, 11 March 2022, consulted on 12 March 2022.

<sup>&</sup>lt;sup>42</sup> OCHA oPt, Protection of civilians report 5 – 8 October 2021, consulted on 22 October 2021; OCHA oPt, Protection of civilians report 9 October - 1 November 2021, consulted on 7 November 2021; The Independent, Palestinians say dozens of cars wrecked in settler vandalism, 9 November 2021, consulted on 10 November 2021; B'Tselem, State Business: Israel's misappropriation of land in the West Bank through settler violence, November 2021, consulted on 30 November 2021.

<sup>&</sup>lt;sup>43</sup> OCHA, *Data on casualties*, consulted on 6 April 2022. See also https://www.ochaopt.org/data/casualties .

<sup>&</sup>lt;sup>44</sup> Confidential source, 21 March 2022.

<sup>&</sup>lt;sup>45</sup> UN Human Rights Council, UN experts alarmed by rise in settler violence in occupied Palestinian territory, 10 November 2021, consulted on 13 March 2022.

<sup>&</sup>lt;sup>46</sup> OCHA oPt, Protection of civilians report 5 – 8 October 2021, consulted on 22 October 2021; OCHA oPt, Protection of civilians report 9 October - 1 November 2021, consulted on 7 November 2021; The Independent, Palestinians say dozens of cars wrecked in settler vandalism, 9 November 2021, consulted on 10 November 2021; B'Tselem, State Business: Israel's misappropriation of land in the West Bank through settler violence, November 2021, consulted on 30 November 2021.

<sup>&</sup>lt;sup>47</sup> OCHA oPt, *Protection of Civilians Report 8-21 February 2022*, 28 February 2022, consulted on 1 March 2022.

<sup>&</sup>lt;sup>48</sup> ACLED, *Dashboard Israel*, 2022, consulted on 21 April 2022.

was followed by a military response by Israel in the West Bank, especially around the city of Jenin. There were also several violent incidents and conflicts in East Jerusalem, including on the Temple Mount/Haram al-Sharif. More than 150 Palestinians were injured. There were stabbing incidents by Palestinians against Israeli police and civilians in East Jerusalem and the West Bank. The Israeli army took extra security measures and, among other things, restricted the freedom of movement of Palestinians in Jenin. Israeli troops made around 350 arrests in the West Bank. In addition, eighteen Palestinians – including minors – were also killed by Israeli forces in the first two weeks of Ramadan. <sup>49</sup> On 18 April 2022, a rocket was fired at Israel from Gaza. This rocket was intercepted by Israel. The rocket attack was not claimed by Hamas or any other armed group. This was followed by air strikes by Israel on Gaza, with no injuries. <sup>50</sup>

#### 1.2.2 Security services and armed groups

In the Palestinian Territories, the following are the most well-known security services: the General Intelligence Service (GIS), Preventive Security (PS) and Presidential Guard (PG) are affiliated with the PA, the National Security Forces (NSF) and Palestinian Civil Police (PC) are affiliated with the PA in the West Bank and with Hamas in Gaza, and the Security and Protection Unit and Internal Security Forces (ISF) are affiliated with Hamas. t1

The following armed groups (with their own power structures) exist in the Palestinian Territories:

- Izz al-Din al-Qassam Brigades (IQB): the military branch of Hamas, with Mohammed Deif as its leader, which was founded in 1991.
- Al-Quds Brigades (AQB): the military branch of Islamic Jihad, which was founded in 1992.

ECFR, Mapping Palestinian Politics: About Security Forces, consulted on 11 January 2022. See also Error!

Hyperlink reference not valid.; There are also the Preventative Security (PSO), the Military Intelligence (MI), the Civil Defense (CD), Customs Police, various committees and administrative bodies, the Organization and Administration Commission (OAC), Military Judiciary Commission (MJC), the Political and National Guidance Commission, the General Military Training Commission (GMTC), the Central Logistics Commission (LC), District Coordination Office (DCO), the Military Medical Services Directorate (MMS), the Central Armament Directorate (ARM), the Military Financial Administration (CFA); Confidential source, 11 March 2022.

<sup>&</sup>lt;sup>49</sup> BBC Monitoring, *Israeli police shoot Palestinian during suspected stabbing attack*, 19 March 2022, consulted on 20 March 2022; OCHA oPt, Highlights from the reporting period 18 - 21 maart 2022, 25 March 2022, consulted on 26 March 2022; BBC Monitoring, IS claims responsibility for deadly Israel attack, 28 March 2022, consulted on 29 March 2022; The New York Times, Palestinian Gunman Kills 5 in Israel's Fifth Attack in Recent Days, 29 March 2022, consulted on 30 March 2022; BBC Monitoring, Briefing: Surge in settler violence in West Bank follows deadly attack, 31 March 2022, consulted on 1 April 2022; BBC News, Two Palestinians killed in gunfight during Israeli West Bank raid, 1 April 2022, consulted on 2 April 2022; BBC Monitoring, Palestinian killed in Hebron as clashes erupt in West Bank, 1 April 2022, consulted on 2 April 2022; BBC Monitoring, Briefing: Tensions flare in East Jerusalem at start of Ramadan, 3 April 2022, consulted on 4 April 2022; BBC Monitoring, Briefing: Israel limits entry from West Bank for Ramadan prayers, 6 April 2022, consulted on 7 April 2022; The New York Times, Latest Attack In Israel Kills At Least Two In Tel Aviv, 8 April 2022, consulted on 9 April 2022; BBC Monitoring, Palestinian killed as Israeli forces raid Jenin refugee camp, 9 April 2022, consulted on 10 April 2022; BBC Monitoring, Briefing: Palestinian woman killed as surge of violence continues, 10 April 2022, consulted on 11 April 2022; OCHA oPt, Protection of Civilians Report 22 March - 4 April 2022, 11 April 2022, consulted on 12 April 2022; BBC Monitoring, Briefing: Israeli officer kills Palestinian after reported stabbing attempt, 12 April 2022, consulted on 13 April 2022; BBC Monitoring, Palestinian death toll rises to five in occupied West Bank, 14 April 2022, consulted on 14 April 2022; NOS, Geweld bij al-Aqsamoskee Oost-Jeruzalem: ruim 150 gewonden gemeld, 15 April 2022, consulted on 15 April 2022; BBC Monitoring, Israeli troops kill five Palestinians in West Bank clashes, 14 April 2022, consulted on 15 April, 2022; Confidential source, 5 April 2022; Confidential source, 15 April 2022; BBC Monitoring, Tensions flare as Jews visit Jerusalem's al-Aqsa mosque compound, 17 April 2022, consulted on 18 April 2022.

<sup>50</sup> NOS, Israël onderschept raket en voert bombardement uit op Gaza, 19 April 2022, consulted on 20 April 2022; BBC Monitoring, Briefing: Israel strikes Gaza after rocket fired amid high tension, 19 April 2022, consulted on 20 April 2022.

- Al-Aqsa Martyrs Brigades (AMB): the coalition group formed during the Second Intifada,<sup>52</sup> which is associated with, but not recognised by, Fatah. The AMB is fighting against Israel, but a number of AMB activists for example after an amnesty following the Second Intifada were incorporated into the PA security services or splintered into other groups active in Gaza (al-Aqsa Martyrs Brigades-Nidal al-Amoudi (AMB-NA), al-Aqsa Martyrs Brigades-Jaish al-Asifah, al-Aqsa Martyrs Brigades-Ayman Jawda, and the Popular Resistance Committees). There were also (violent) conflicts between the AMB and the PA security services.
- Al-Nasser Salah al-Deen Brigades (PRC): the military branch of the PRC, but in practice there is no difference between the military and political branches.
- Abu Ali Mustafa Brigades: the military branch of the PFLP active in the West Bank and Gaza.
- National Resistance Brigades: the military branch of the DFLP active in Gaza.
- Harakat al-Sabireen: founded by former members of Islamic Jihad in 2014 and active in the West Bank and Gaza.
- Salafi jihadists: a number of different groups (including the active *Jaysh al-Islam*, *Jaysh al-Umma*, *Jund Ansar Allah* and *al-Tawhid wal-Jihad*) that were formed after Hamas took power in Gaza in 2007. These groups are fighting against Israel but they are also in conflict with Hamas. Their presence is minimal compared to Hamas.<sup>53</sup>

#### 1.3 Living conditions

#### 1.3.1 General

In 2021, there was limited access in the West Bank and East Jerusalem to basic services, evictions, loss of land, destruction of real estate and restrictions in freedom of movement.<sup>54</sup> The situation in Gaza is described by UN agencies as a 'trajectory of de-development': the socio-economic position and living conditions in Gaza have deteriorated since Israel initiated the (land, air and sea) closure of Gaza due to the seizure of power by Hamas in 2007. This has had negative consequences for the economy, employment and education. The UN has called it a blockade<sup>55</sup> and stated that it is a violation of international humanitarian law and is leading to hardship for the people of Gaza. Around 2.1 million people live in Gaza, 1.4 million of whom are registered as refugees with the United Nations Relief and Works Agency (UNRWA). Gaza's eight refugee camps have one of the highest population densities in the world. 56 There is a housing shortage, partly due to the intense demographic pressure, the destruction of homes, Israeli air strikes as part of the violent conflict between Hamas and Israel and a shortage of building materials (see also 1.2.1).<sup>57</sup> Of the people in need identified by OCHA in the Palestinian Territories in 2021 (2.1 million), 947,000 were refugees outside the refugee camps and 464,000 were

 $<sup>^{52}</sup>$  Period of Palestinian resistance and escalating conflict with Israel from 2000 to 2005.

<sup>&</sup>lt;sup>53</sup> ECFR, Mapping Palestinian Politics: About Armed Groups, consulted on 11 January 2022. See also Error! Hyperlink reference not valid.

<sup>&</sup>lt;sup>54</sup> OCHA, *Humanitarian Response Plan: Working Document*, December 2021, p. 5.

 $<sup>^{\</sup>rm 55}$  The Netherlands has referred to it as the closure of land borders.

<sup>56</sup> UNRWA, Were We Work: Gaza Strip, consulted on 18 January 2022; Gisha, Gaza Up Close, 1 September 2021, consulted on 18 January 2022; UNHCR, Country of Origin Information on the Situation in the Gaza Strip, Including on Restrictions on Exit and Return, 2018, p. 4; Gisha, The illusion of disengagement, 12 September 2019, consulted on 12 April 2022

<sup>&</sup>lt;sup>57</sup> UNHCR, Country of Origin Information on the Situation in the Gaza Strip, Including on Restrictions on Exit and Return, 2018, p. 4.

refugees in refugee camps.<sup>58</sup> It is not clear how many of these numbers are UNRWA-registered. In March 2022, UNHCR said that the humanitarian and human rights situation in Gaza was serious and called on countries not to facilitate forced returns to Gaza.<sup>59</sup>

#### Work, food, water and electricity: West Bank

According to OCHA, in 2021 21% of West Bank residents could be regarded as people in severe and extreme need. 60 The unemployment rate in the West Bank was 16.9% in the second quarter of 2021. 61 In 2017, 20% of individuals and 13.7% of households in the West Bank were living below the poverty line. 62 In the West Bank, 40% of households are facing food insecurity. In Area C in particular, there is said to be major food insecurity. 63 Almost 670,000 people in the West Bank have limited access to water; this particularly applies to Area C residents, who often do not have their own water supply. Due to the high cost of alternative access to water in some areas, such as Area C, households are spending 15% of their income on water. The water infrastructure in Area C also runs the risk of being demolished if there are no building permits (see also 1.2.4). 64 Residents of the West Bank depend on Israeli supplies for the majority of their electricity. This supply was interrupted a number of times during the reporting period due to late payments and debts by the PA. 65

#### Work, food, water and electricity: Gaza

Three out of every five residents in Gaza are living below the poverty line. 66 Unemployment in Gaza was around 45% in June 2021. 35.7% of the employed were working in public service with an average income of USD 28 per day. The average income in the private sector was just under USD 10 per day. 67 Around 80% of the population in Gaza is dependent on international humanitarian aid – among other things due to high food prices. An estimated 39% of households are living with food insecurity and young children are at risk of chronic malnutrition. More than one million UNRWA refugees depend on UNRWA for food aid (see also 1.3.2). UNRWA issues micro-loans to Palestinian refugees, but also to the poor and other marginalised people living or working in the vicinity of refugees. Clean water is not accessible to 95% of the population. Since April 2021, access to electricity has been scaled up from four or five hours a day to fourteen hours a day. 68 According to another source, Gaza continued to experience power outages and residents of the area had access to electricity for only two to four hours a day. This supply would not be enough to run all the appliances in a household at the same time. 69 The fuel

 $<sup>^{\</sup>rm 58}$  OCHA, Humanitarian needs overview OPT, December 2021, p. 6.

<sup>&</sup>lt;sup>59</sup> UNHCR, UNHCR position on returns to Gaza, March 2022, p. 31-32.

<sup>60</sup> OCHA, *Humanitarian needs overview OPT*, December 2021, p. 6.

<sup>61</sup> World Bank Group, Economic Monitoring Report to the Ad Hoc Liaison Committee, 17 November 2021, p. 4.

<sup>&</sup>lt;sup>62</sup> UN General Assembly, Economic costs of the Israeli occupation for the Palestinian people: poverty in the West Bank between 2000 and 2019, 30 August 2021, p. 12-13.

<sup>&</sup>lt;sup>63</sup> Palestine Food Security Sector & Palestinian Central Bureau of Statistics, Socio-Economic & Food Security Survey 2020: State of Palestine, 2020, p. 1; OCHA, Palestinian strive to access water in the Jordan Valley, 22 June 2021, consulted on 2 March 2022.

<sup>&</sup>lt;sup>64</sup> West Bank Protection Consortium, Palestinian access to water& attacks on WASH structures in Area C, April 2021, p. 1.

<sup>&</sup>lt;sup>65</sup> Reuters, Israeli electric company ends power cuts to West Bank after Palestinians pay debt, 22 January 2020, consulted on 20 January 2022; The Times of Israel, Israeli electrical company to cut power to West Bank Palestinians over debts, 28 October 2021, consulted on 20 January 2022.

<sup>&</sup>lt;sup>66</sup> The Independent, *Defiant in war and isolation, Hamas plays long game in Gaza*, 20 December 2021, consulted on 14 January 2022.

<sup>&</sup>lt;sup>67</sup> Gisha, *Unemployment rate in Gaza at 44,7% in Q2 of 2021*, 21 October 2021, consulted on 14 January 2022.

<sup>&</sup>lt;sup>68</sup> UNRWA, Were we Work: Gaza Strip, consulted on 18 January 2022; Gisha, Gaza Up Close, 1 September 2021, consulted on 18 January 2022; UNHCR, Country of Origin Information on the Situation in the Gaza Strip, Including on Restrictions on Exit and Return, 2018, p. 10; UNRWA, What we do: Microfinance, consulted on 6 April 2022

<sup>&</sup>lt;sup>69</sup> Confidential source, 24 January 2022.

supply for the power plant in Gaza can be blocked by Israel, as happened in August 2020 and during the conflict in May 2021. The shortage of electricity is affecting industry and agriculture as well as access to essential services such as healthcare, water and sanitation.<sup>70</sup>

#### The Internet

The quality of the internet in the Palestinian Territories is poor and the costs are high. Israel's approval is required to build communication infrastructure in the Palestinian Territories and to obtain materials for this purpose. All of the communication infrastructure is said to be connected to Israel's infrastructure. Mobile internet in the West Bank is limited to the 3G network. Gaza residents only have access to 2G mobile internet.<sup>71</sup>

#### Education

The education system in the Palestinian Territories consists of public schools from the Ministry of Education, UNRWA schools (primary education for UNRWA registered refugees; see also 4.1) and schools in the private and non-profit sector. In the West Bank, including in Area C, schools are maintained by the PA. In East Jerusalem, these fall under the Israeli administration/government. The education system has to contend with early school leavers, partly as a result of poverty. Children with disabilities are more at risk of dropping out of school.<sup>72</sup>

Schools in Gaza have to deal with overcrowded classrooms, interruptions in student attendance and early school leavers. <sup>73</sup> Teachers in Gaza do not always receive their salaries. Extra financial contributions from parents to schools are normal and are a major source of income for schools. <sup>74</sup> The organisation *Al Mezan Center for Human Rights* states that access to education for children in Gaza is being detrimentally affected by the violent incidents. For example, in May 2021 186 schools (136 public, 13 private and 37 UNRWA schools) had been damaged by the conflict between Hamas and Israel. In addition, schools were not being used for education but for the reception of displaced persons. Of the schools used for shelter, 63 had suffered minor damage. <sup>75</sup>

#### Health care

The health system in the Palestinian Territories is facilitated by the Palestinian Ministry of Health, UNRWA, NGOs and private clinics/hospitals. According to thinktank *Al Shabaka*, <sup>76</sup> the quality of and access to healthcare differs for each facility depending on access to electricity and medical devices. There is said to be financial and physical barriers and uneven health insurance coverage. <sup>77</sup> The health sector in the West Bank and Gaza is struggling with shortages of medical equipment, drugs, infrastructure, staff and medical specialists. In the refugee camps there is also poor access to healthcare, including trauma and mental health care, especially for

<sup>&</sup>lt;sup>70</sup> UNRWA, Were we Work: Gaza Strip, consulted on 18 January 2022; Gisha, Gaza Up Close, 1 September 2021, consulted on 18 January 2022; UNHCR, Country of Origin Information on the Situation in the Gaza Strip, Including on Restrictions on Exit and Return, 2018, p. 10.

<sup>&</sup>lt;sup>71</sup> Open Democracy, Israel's digital apartheid is silencing Palestinians, 20 May 2021, consulted on 20 January 2022; The Jerusalem Post, World Bank: Israel must give Palestinians 4G so economy can thrive, 9 February 2022, consulted on 1 March 2022.

<sup>&</sup>lt;sup>72</sup> UNICEF, State of Palestine Country Report on out-of-school children, July 2018, p. 1-3, 8-11.

<sup>&</sup>lt;sup>73</sup> Al Mezan Center for Human Rights, Fact Sheet on Obstacles to Accessing Education in the Gaza Strip, August 2021, p. 6-7.

<sup>&</sup>lt;sup>74</sup> UNICEF, State of Palestine Country Report on out-of-school children, July 2018, p. 8-11.

<sup>&</sup>lt;sup>75</sup> Al Mezan Center for Human Rights, *Fact Sheet on Obstacles to Accessing Education in the Gaza Strip*, August 2021, p. 6-7.

<sup>&</sup>lt;sup>76</sup> Al Shabaka (The Palestinian Policy Network) is a non-profit think-tank that focuses on Palestinian human rights.

<sup>77</sup> Al Shabaka, *The Conditional Right to Health in Palestine*, 30 June 2019, consulted on 19 January 2022.

children. During the coronavirus pandemic in 2020, there were shortages of IC beds, medicines and tests in the West Bank.<sup>78</sup> Access to medical care in the West Bank for persons coming from Gaza is not guaranteed. The PA is said to have unpaid hospital bills at private hospitals for special cancer treatments that are not available in public hospitals.<sup>79</sup>

Healthcare in Gaza is worse than in the West Bank. During the coronavirus pandemic in 2020, there were also shortages of IC beds, medicines and tests in Gaza. For specialist care, persons from Gaza must travel to East Jerusalem, the West Bank, Israel or Jordan. Travel permits are often delayed or refused by the Israeli authorities (see also 1.4.1). 80 People in need of medical care have died as a result of delays in this application process. Due to the poor living conditions, the closure of borders and violent incidents during the conflict, there is also a need for trauma and mental healthcare, especially for children. Access to healthcare in the refugee camps in Gaza is poor. 81

In addition, closure of the borders is said to pose environmental, health and safety risks due to limited recycling and waste disposal options. 25 tonnes of old batteries are said to be stored in unsafe conditions at various locations in Gaza.<sup>82</sup>

#### Humanitarian aid

According to sources, Israel controls access to humanitarian aid in the Palestinian Territories.83 For the West Bank and East Jerusalem, a source says that in practice no permission is required from the Israeli authorities to import humanitarian aid.84 In Gaza, international organisations use the Kerem Shalom border crossing to deliver humanitarian aid. It is not clear whether this requires permission from the Israeli authorities in practice. According to one source, the Israeli authorities do not deny access to the providers of humanitarian aid. In May 2021, humanitarian aid to Gaza was temporarily not permitted due to the violent conflict (see also 1.2.1). Certain items, such as building materials, fuel, medical devices or sanitary infrastructure materials, are always subject to import restrictions. 85 According to one source, humanitarian aid from Arab countries enters Gaza via the Rafah border crossing with Egypt. For example, humanitarian aid during the escalation of violence in May 2021 and Covid-19 vaccines during the coronavirus pandemic came in through this crossing.86 The import of humanitarian aid was possible during the May 2021 conflict but was made more difficult in practice due to the temporary closures of the Erez and Kerem Shalom border crossings.87

## Access to work in Israel

<sup>&</sup>lt;sup>78</sup> Anera, *Health Care System in Palestine*, 21 September 2020, consulted on 18 January 2022.

<sup>&</sup>lt;sup>79</sup> Al Jazeera, Teen's death fuels Palestinian Authority corruption claims, 8 February 2022, consulted on 9 February 2022

<sup>80</sup> Anera, Health Care System in Palestine, 21 September 2020, consulted on 18 January 2022; Confidential source, 24 January 2022; The Guardian, Palestinian baby dies after treatment delayed by Israeli blockade of Gaza, 1 April 2022, consulted on 2 April 2022.

<sup>&</sup>lt;sup>81</sup> UNHCR, Country of Origin Information on the Situation in the Gaza Strip, Including on Restrictions on Exit and Return, 2018, p. 12 -13; Anera, Health Care System in Palestine, 21 September 2020, consulted on 18 January 2022; The Guardian, Palestinian baby dies after treatment delayed by Israeli blockade of Gaza, 1 April 2022, consulted on 2 April 2022.

<sup>82</sup> The Independent, Gaza's old battery pileups pose risk to health, environment, 22 December 2021, consulted on 27 December 2021.

 $<sup>^{\</sup>rm 83}$  Confidential source, 24 January 2022; Confidential source, 28 February 2022.

<sup>84</sup> Confidential source, 28 February 2022.

<sup>85</sup> Gisha, Gaza Up Close, 1 September 2021, consulted on 18 January 2022; Confidential source, 24 January 2022.

<sup>&</sup>lt;sup>86</sup> Confidential source, 24 January 2022.

<sup>87</sup> Gisha, Israel continues to restrict movement to and from Gaza, harming residents, 23 May 2021, consulted on 2 March 2022.

Palestinians from the West Bank and Gaza can, under certain conditions, work outside their own place of residence in Israel or in Israeli settlements (see also 1.4.3). To do this, they need a work permit that has been issued by the Israeli authorities. To obtain a work permit, workers must first have an Israeli employer. In December 2020, work permits in the construction sector were uncoupled from employers and arranged through intermediaries. Since March 2021, this has also been the case in industry and the services sector. Trade in work permits is prohibited under Israeli and Palestinian law.<sup>88</sup> Work permit applications are free<sup>89</sup> but were previously sold illegally by employers. Since then, work permits have been illegally sold by intermediaries. In 2020, a work permit obtained in this way cost approximately NIS 2,500 (EUR 710)<sup>90</sup>.

Besides undergoing a security check, a person must be over 22 years old and married in order to work in Israel and over 18 years old to work in Israeli settlements. These rules mean that some people get married in order to be eligible for a work permit. Work permits are only granted in sectors where Palestinian workers do not compete with Israelis. 91

Before the coronavirus pandemic, around 133,000 Palestinians worked in Israel, according to the *International Labour Organisation* (ILO). Of the estimated number of Palestinian workers in Israel and Israeli settlements, 94,000 had a work permit in 2019. 99% of the permit holders are male and most of them work in the construction sector. 92 In 2020, there were approximately 125,000 Palestinians working in Israel and Israeli settlements. In addition, there were around 24,000 Palestinians working without a work permit. The ILO did not specify whether Palestinians from East Jerusalem who have a Jerusalem identity card or an Israeli identity card required a work permit. 93

There are cases where employers buy up work permits and then withhold money from the salaries of their Palestinian employees. In addition, Palestinian workers with work permits are at risk of being paid below the minimum wage and not receiving employee benefits, such as access to healthcare, to which they are legally entitled. As a result, Palestinian workers may be exposed to bad working conditions, such as long working days, poor terms of employment and low pay. 94 There were cases where Israeli security forces withheld work permits as a collective punishment for Palestinian communities in which families were suspected of political activity or attacks against Israel. 95 The ILO says that around one third of the salary of Palestinian employees is spent on intermediaries who arrange work permits.

<sup>88</sup> International Trade Union Confederation (ITUC), Workers Rights in Crisis: Palestinian workers in Israel and the settlements, 12 April 2021, p. 9-10; International Labour Organisation, The situation of workers of the occupied Arab territories, Report of the Director-General – Appendix 2021, 2021, p. 11; KavLaoved, Policy brief – January 2022: Who is protecting the rights of Palestinian workers in Israel, January 2022, p. 1.

 $<sup>^{\</sup>rm 89}$  Confidential source, 25 March 2022.

<sup>&</sup>lt;sup>90</sup> Exchange rate on 6 April 2022.

<sup>&</sup>lt;sup>91</sup> International Trade Union Confederation (ITUC), Workers Rights in Crisis: Palestinian workers in Israel and the settlements, 12 April 2021, p. 9-10; International Labour Organisation, The situation of workers of the occupied Arab territories, Report of the Director-General – Appendix 2021, 2021, p. 11; KavLaoved, Policy brief – January 2022: Who is protecting the rights of Palestinian workers in Israel, January 2022, p. 1.

<sup>&</sup>lt;sup>92</sup> International Trade Union Confederation (ITUC), Workers Rights in Crisis: Palestinian workers in Israel and the settlements, 12 April 2021, p. 9.

<sup>&</sup>lt;sup>93</sup> International Labour Organisation, The situation of workers of the occupied Arab territories, Report of the Director-General – Appendix 2021, 2021, p. 11, 27, 33.

<sup>94</sup> The Jerusalem Post, Palestinian workers in Israel: Opportunity or exploitation?, 1 August 2021, consulted on 26 November 2021.

<sup>95</sup> International Trade Union Confederation (ITUC), Workers Rights in Crisis: Palestinian workers in Israel and the settlements, 12 April 2021, p. 10.

Palestinians earn up to four times as much per day in Israel as they do in Gaza and up to twice as much as Palestinians from the West Bank  $^{96}$ 

#### 1.3.2 Role of UNRWA

A separate UN agency (the United Nations Relief and Works Agency (UNRWA) was established in 1949 to assist the Palestinian refugees. UNRWA registers and provides aid to Palestinian refugees and four other categories of persons entitled to this relief. There are also a number of other categories of persons who are entitled to UNRWA relief but not entitled to UNRWA registration. UNRWA registration is therefore not always required to receive UNRWA relief.<sup>97</sup>

UNRWA is active in the Palestinian Territories and focuses on the following themes: primary education, primary healthcare, humanitarian and social services, infrastructure and camp improvement, financial support (micro-credit) and emergency relief (also in the case of armed conflict). Access to further education, specialist healthcare, the social safety net system and emergency relief is not universally granted and is based on certain criteria. UNRWA does not provide protection for refugees and has no control over the administration of refugee camps. <sup>98</sup>

UNRWA largely relies on voluntary donations from states. <sup>99</sup> UNRWA has been struggling for years with financial problems and budget deficits due to the growing refugee population and declining contributions from donors. In particular, the cessation of US funding under the Trump administration in 2018 has impacted UNRWA's finances. UNRWA was previously unable to pay the salaries of its staff, consisting mostly of Palestinian employees, for a number of months in 2020. <sup>100</sup> In the 2019-2020 academic year, UNRWA was unable to implement a food aid programme in Gaza without using a UN-CERF (emergency) fund and advances from its future budget. <sup>101</sup> While the US reintroduced the UNRWA contribution under the Biden administration, other donor countries such as the UK and the Gulf States reduced their contributions to UNRWA in 2021. <sup>102</sup> In 2021, UNRWA issued an emergency call for USD 231 million for humanitarian services in the Palestinian Territories. <sup>103</sup> In November 2021, it was announced that the salaries of 28,000 UNRWA employees could not be paid on time. According to one source, UNRWA was trying to reduce its workforce. <sup>104</sup> UNRWA is said to have received 48% of its 2022

<sup>&</sup>lt;sup>96</sup> International Labour Organisation, The situation of workers of the occupied Arab territories, Report of the Director-General – Appendix 2021, 2021, p. 11.

Ministry of Foreign Affairs, Thematisch Ambtsbericht Palestijnen in Libanon, January 2021, p. 7; Palestinian refugees (Palestinians from the former British Mandate who lost their homes and access to income due to the 1948 conflict), relatives of refugees who are not themselves refugees, frontier villagers, poor people from Jerusalem and Gaza and compromised cases are entitled to UNRWA registration and UNRWA relief. The remaining group consists of non-Palestinian persons or non-refugee family members who are entitled to UNRWA services because they were working/living in Palestine at the time. UNRWA, Eligibility and registration Q&A, 2020, p. 1-2; UNRWA Dashboard, UNRWA-mandaatgebied, consulted on 28 February 2022. See also Error! Hyperlink reference not valid.; Confidential source, 28 February 2022.

<sup>98</sup> UNRWA, Letter from UNRWA to UNHCR describing UNRWA's mandate and services, 22 September 2021, consulted on 7 April 2022; Confidential source, 28 February 2022.

<sup>&</sup>lt;sup>99</sup> Confidential source, 28 February 2022.

<sup>&</sup>lt;sup>100</sup> Ministry of Foreign Affairs, *Thematisch Ambtsbericht Palestijnen in Libanon*, January 2021, p. 40-41.

<sup>&</sup>lt;sup>101</sup> Confidential source, 28 February 2022.

<sup>&</sup>lt;sup>102</sup> The Guardian, UN Palestine refugee aid agency 'close to collapse' after funding cuts, 5 November 2021, consulted on 21 January 2022.

<sup>&</sup>lt;sup>103</sup> UNRWA, Emergency appeal, emergency appeal for the occupied Palestinian territory, consulted on 21 January 2022.

<sup>104</sup> Confidential source, 24 January 2022; UNWRA has more than 30,000 employees. UNRWA, Working at UNRWA, consulted on 22 February 2022. See also Error! Hyperlink reference not valid.

and 2023 budget from donations. 105 On 18 January 2022, Phillipe Lazzarini, the head of UNRWA, said that the organisation was suffering from such a chronic financial deficit that it might collapse. It would need USD 1.6 billion to be able to carry out its activities in 2022. 106

There are countless agencies and NGOs operating in the Palestinian Territories that provide relief to Palestinians who are registered as refugees. For example, UNRWA works together with other UN organisations and various larger and smaller (local) NGOs. 107 According to one source, like UNRWA most other organisations have budget deficits due to reduced donor funding. 108

1.3.3 Humanitarian situation and reconstruction in Gaza after May 2021 UN emergency aid agency OCHA has stated that during the conflict period in May 2021, 1,042 homes and commercial buildings were destroyed in Gaza, 769 homes and commercial buildings were seriously damaged, 58 educational facilities were damaged, nine hospitals were partially damaged, nineteen health clinics were damaged and 800,000 people (around one third of the population) were without normal access to drinking water in line with international standards. At the height of the escalation of violence in May 2021, 113,000 people were displaced in Gaza, with 71,000 of them using UNRWA schools as emergency shelters and 35,000 staying with host families or in informal shelters. In November 2021, approximately 8,250 of these people were still displaced (see also 4.2). 109

> Human Rights Watch has stated that USD 380 million worth of physical damage and USD 190 million worth of economic damage were caused in Gaza during the May 2021 conflict. 110 For emergency humanitarian aid and the reconstruction of damaged infrastructure in the Palestinian Territories, USD 95 million was requested in July 2021 from an alliance between several organisations and OCHA. By July 2021, half of this money had been raised. Besides rebuilding basic services such as healthcare and access to water, OCHA also made funds available for housing benefits for displaced persons whose houses had been destroyed. 111 Several countries supported humanitarian aid for Gaza - among other things through humanitarian partners, UN organisations, NGOs and the Red Cross/Red Crescent (International Committee of the Red Cross - ICRC). 112 A number of aid organisations were active in Gaza after the Israeli air strikes. UNRWA supported displaced persons with emergency shelters in UNRWA schools. Various organisations were active in providing food parcels, access to water and sanitation, non-food items such as toiletries and medicines and financial support, including credits for the purchase of food. 113

<sup>&</sup>lt;sup>105</sup> The Independent, *UN agency for Palestinian refugees faces funding crisis*, 30 November 2021, consulted on 21 January 2022

<sup>106</sup> UN News, UNRWA seeks \$1.6 bilion to support Palestinian refugees in 2022, 18 January 2021, consulted on 21 January 2022; The Guardian, UN Palestine refugee aid agency 'close to collapse' after funding cuts, 5 November 2021, consulted on 21 January 2022.

<sup>107</sup> UNRWA, *NGO partners*, consulted on 1 February 2022. See also **Error! Hyperlink reference not valid.**108 Confidential source, 24 January 2022.

<sup>109</sup> OCHA, GAZA: Internally Displaced people, 20 May 2021, p. 1; OCHA, Gaza Strip: Escalation of hostilities, 21 May 2021, p. 1; OCHA, Overview November 2021, 3 November 2021, consulted on 13 January 2022.

Human Rights Watch, *Israel and Palestine: Events of 2021*, consulted on 14 January 2022 111 OCHA, Response to the escalation in the oPt, Situation Report No. 6: 25 June - 1 Juli 2021, 1 July 2021, p. 1-2, 4.

<sup>112</sup> OCHA, Gaza Strip: Escalation of hostilities, 21 May 2021, p. 1; European Union, Palestine Factsheet, 23 September 2021, consulted on 14 January 2022; The Independent, Hamas: Qatar paying public salaries by sending fuel to Gaza, 30 November 2021, consulted on 14 January 2022.

<sup>113</sup> OCHA, Gaza Strip: Escalation of hostilities, 21 May 2021, p. 1; ACTED, Gaza: ACTED's Emergency Assistance for families to meet their needs, 23 December 2021, consulted on 13 January 2022.

According to sources, government assistance is inconsistent and families must try to survive for extended periods without government aid or must borrow money from relatives. 114 Hamas is said to have an estimated fifty thousand civil servants. 115 Most civil servants in Gaza are said to receive around 55% of their salaries. Qatar is supporting Gaza by supplying fuel to Hamas, which is allegedly selling it and using the proceeds to pay government salaries. 116

In September 2021, Israel allowed building materials into Gaza again after hundreds of Palestinians in Gaza protested along the border with Israel against the closure. 117 In addition, after the escalation of violence in May 2021, Qatar began to provide support through the UN for one hundred thousand poor families who consequently received USD 100 per month up to the end of 2021.118 In early December 2021, Egypt reopened the Rafah border crossing with Gaza - among other things to allow humanitarian aid and building materials to pass through. 119 Hamas threatened another escalation in December 2021, accusing Egypt and Israel of slowing down the reconstruction of Gaza by keeping Gaza closed. 120 Hamas was said to be taxing goods entering Gaza through the Erez and Rafah border crossings, resulting in estimated revenues of up to USD 30 million per month. Individuals who criticised Hamas's monopoly on fuel imports were at risk of detention. It is not clear how many cases this involved. 121 With the help of the United Nations Development Programme (UNDP), the rubble from destroyed homes and buildings in Gaza is being recycled in part for road construction as well as for the construction of cement and building blocks in the private sector. However, it has been reported that these building blocks do not comply with the applicable safety standards. <sup>122</sup> According to a news report, by January 2022 fifty of the 1,650 destroyed homes in Gaza had been restored. 123 One source said that at the end of January 2022 no reconstruction had taken place in Gaza. 124

#### 1.4 Freedom of movement

## 1.4.1 Travel within the country

The Gaza Strip

<sup>114</sup> ACTED, Gaza: ACTED's Emergency Assistance for families to meet their needs, 23 December 2021, consulted on 13 January 2022.

<sup>115</sup> The Independent, Defiant in war and isolation, Hamas plays long game in Gaza, 20 December 2021, consulted on 14 January 2022.

<sup>&</sup>lt;sup>116</sup> The Independent, Hamas: Qatar paying public salaries by sending fuel to Gaza, 30 November 2021, consulted on 14 January 2022.

<sup>&</sup>lt;sup>117</sup> Al Jazeera, Building materials allowed into Gaza after Israeli assault in May, 1 September 2021, consulted on 14 January 2022.

<sup>&</sup>lt;sup>118</sup> OCHA, Response to the escalation in the oPt Situation Report No. 10, September 2021, consulted on 18 January 2022.

<sup>&</sup>lt;sup>119</sup> BBC Monitoring, Egypt resumes opening Rafah crossing with Gaza, 6 December 2021, consulted on 14 January 2022.

BBC Monitoring, Hamas warns of 'escalation' in Gaza, blaming Egypt, 7 December 2021, consulted on 14 January 2022; Oxford Analytica, Palestinian Hamas may respond to Gaza aid delays, 8 December 2021, consulted on 14 January 2022.

<sup>&</sup>lt;sup>121</sup> The Independent, *Defiant in war and isolation, Hamas plays long game in Gaza*, 20 December 2021, consulted on 14 January 2022.

<sup>122</sup> The Independent, Rubble brings opportunity, and risk, in war-scarred Gaza, 13 January 2022, consulted on 14 January 2022.

<sup>123</sup> Reuters, With most Gaza homes wrecked by war still in ruins, smiles for the lucky few, 13 January 2022, consulted on 13 March 2022.

<sup>&</sup>lt;sup>124</sup> Confidential source, 24 January 2022.

People can move freely inside Gaza but there are sometimes temporary checkpoints that are managed by the local authorities. Freedom of movement is limited to the sea and land border areas. Israel announced in early December 2021 that it had completed the so-called 'iron wall', an underground and above-ground wall along the 65-kilometre border with Gaza. Sea Israeli troops use warning shots if people get too close to the border fences with Israel or if fishermen go too far out to sea. These individuals are at risk of arrest by Israeli troops. Iron February 2022, 25 cases were recorded in which Israeli forces had used these types of warning shots. During this period, seven fishermen, three of them children, were arrested at sea and their boat was seized by Israeli troops. Iron

#### The West Bank

In practice, freedom of movement in the West Bank is hampered by the numerous checkpoints, sealed-off areas, limited access to roads, Israeli settlements and Israeli military bases. <sup>129</sup> Israelis cannot enter the parts of the West Bank that are under PA control. Israelis use Israeli roads that bypass Palestinian towns and villages. <sup>130</sup> Palestinians cannot use certain roads reserved exclusively for Israeli settlers. Palestinians living in Hebron may not and cannot use certain roads into Hebron and in the centre of Hebron and risk arrest under Israeli military law if they do so. <sup>131</sup> The Israeli wall, of which 85% is located in the West Bank and not on the Green Line (the land borders according to 1949), is preventing thousands of Palestinians from accessing their (agricultural) lands and Palestinians on the other side of the wall from accessing the rest of the West Bank. <sup>132</sup>

During the coronavirus pandemic in 2020, travel options between cities were curtailed and curfews were imposed by the PA from March to May and in November. $^{133}$ 

#### Freedom of movement in refugee camps

Palestinians in refugee camps in Gaza and in the West Bank can travel into and out of the camps. For all intents and purposes, the camps are residential areas. <sup>134</sup> The various walls and checkpoints that exist in the West Bank make it difficult to travel into and out of various areas in this region. The same rules apply to Palestinians with a regular ID card who live in refugee camps as to Palestinians with a regular ID card who live outside the refugee camps. Residents of the Shu'fat refugee camp in East Jerusalem are at risk of losing their residence permits if they leave East Jerusalem for an extended period of time. This could happen, for example, if a

<sup>&</sup>lt;sup>125</sup> Confidential source, 24 January 2022.

<sup>&</sup>lt;sup>126</sup> Al Jazeera, Israel completes 'iron wall' underground Gaza barrier, 7 December 2021, consulted on 8 December 2022; NOS, Israëlische ondergrondse muur langs Gazastrook na bijna vier jaar voltooid, 7 December 2021, consulted on 8 December 2022.

<sup>&</sup>lt;sup>127</sup> OCHA oPt, *Protection of Civilians Report 7-20 December 2021*, 27 December 2021, consulted on 14 January 2022.

 <sup>128</sup> OCHA OPt, Protection of Civilians Report 8-21 February 2022, 28 February 2022, consulted on 1 March 2022.
 129 B'Tselem, Restrictions on Movement, 11 November 2017, consulted on 10 January 2022; OCHA, Longstanding access restrictions continue to undermine the living conditions of West Bank Palestinians, 8 June 2020, consulted on 13 March 2022; ECFR, Mapping Palestinian Politics: Geography, Occupied Palestinian Territory, consulted on 10 January 2022. See also Refugee Camps – Mapping Palestinian Politics – European Council on Foreign Relations

<sup>(</sup>ecfr.eu)

BBC Monitoring, Palestinian officers extricate Israelis attacked in Ramallah, 2 December 2021, consulted on 3

December 2021; OCHA, West Bank access restrictions, July 2020, consulted on 10 January 2022. See also Error!

Hyperlink reference not valid.

BTSelem, West Bank roads on which Israel forbids Palestinian vehicles, 31 January 2017, consulted on 10 January 2022. See also https://www.btselem.org/freedom\_of\_movement/forbidden\_roads;OCHA, Longstanding access restrictions continue to undermine the living conditions of West Bank Palestinians, 8 June 2020, consulted on 13 March 2022; Al Jazeera, Hebron: Israeli president visit overlooks violent settler control, 1 December 2021, consulted on 10 January 2022; Confidential source, 11 March 2022.

<sup>&</sup>lt;sup>132</sup> B'Tselem, The Separation Barrier, 11 November 2017, consulted on 14 January 2022; OCHA, Longstanding access restrictions continue to undermine the living conditions of West Bank Palestinians, 8 June 2020, consulted on 13 March 2022; Human Rights Watch, Israel and Palestine: Events of 2021, 2022, consulted on 14 January 2022.

<sup>&</sup>lt;sup>133</sup> Freedom House, Freedom in the World 2021: West Bank, 2021, consulted on 10 January 2022.

<sup>&</sup>lt;sup>134</sup> Confidential source, 24 January 2022; Confidential source, 7 February 2022.

person is working or studying abroad or if a person has not lived in the city for more than seven years, even if they have travelled back to the city in between. Due to the Israeli wall, this refugee camp, which officially belongs to Jerusalem, is located in the West Bank side. This makes it difficult for residents to travel to other parts of East Jerusalem. Many people from East Jerusalem are said to have moved to the camp because the accommodation costs and the cost of living in Jerusalem are so high. As a result, the camp is overcrowded.

#### Palestinians from other countries

Palestinians from other countries (that is, persons who were born outside the Palestinian Territories, have not previously lived in the Palestinian Territories and do not have a Palestinian ID number) who have settled in the Palestinian Territories can move freely inside Gaza and the West Bank provided they have a valid residence permit. 137 According to one source, the Israeli authorities have control over whether Palestinians from other countries are permitted to settle in the Palestinian Territories. It is not clear on which basis the Israeli authorities determine whether or not a person is permitted to settle in the Palestinian Territories. Palestinians with a foreign nationality, especially a European or American nationality, are eligible for a three-month travel permit that allows them to enter the Palestinian Territories via Jordan. This travel permit does not allow entry to Israel. Many of these Palestinians cannot re-enter the Palestinian Territories because the Israeli authorities do not allow this after the expiry date of a first travel permit. 138 Many people in this group are undocumented because they have entered illegally or on expired travel permits and do not have a valid residence permit. This limits their freedom of movement. Some of these people hardly ever leave their homes for fear of being arrested/deported (see also 2.5). According to sources, Palestinian refugees living outside the Palestinian Territories who have no identity documents or only a zero passport (see also 2.2) cannot enter Gaza. 139

## 1.4.2 Settlement in the Palestinian Territories

Israel monitors the registration of births, marriages, divorces, deaths and changes of address (see also 2.4). To settle in the other part of the Palestinian Territories (from Gaza to the West Bank and vice versa), permission from the Israeli authorities is required. People can apply for this permission by means of family reunification, for example. These types of permits to settle in the West Bank have been issued very rarely since 2003 (see also 2.3). Residents of Gaza, regardless of their registration, cannot settle in the West Bank. This is possible, however, the other way around. Travel between the West Bank and East Jerusalem or Israel also requires a permit from the Israeli authorities. These permits are only rarely granted to residents of Gaza. Applications from Gaza and the West Bank are said to be often refused and issued permits can be revoked without explanation. Travel permits can be cancelled when the Israeli military closes the borders, such as during Jewish holidays. 141

<sup>&</sup>lt;sup>135</sup> Haaretz, Who are East Jerusalems permanent residents, 10 April 2018, consulted on 23 February 2022.

<sup>136</sup> Francesca P. Albanese and Lex Takkenberg, *Palestinian refugees in international law*, 2020, p. 243.

<sup>&</sup>lt;sup>137</sup> Confidential source, 24 January 2022; Confidential source, 24 January 2022.

<sup>138</sup> Confidential source, 11 April 2022.

<sup>&</sup>lt;sup>139</sup> Confidential source, 24 January 2022; Confidential source, 24 January 2022.

<sup>&</sup>lt;sup>140</sup> B'Tselem, Restrictions on Movement, 11 November 2017, consulted on 10 January 2022; Confidential source, 24 January 2022; Confidential source, 20 April 2022.

<sup>&</sup>lt;sup>141</sup> B'Tselem, Restrictions on Movement, 11 November 2017, consulted on 10 January 2022.

Persons registered with UNRWA retain their right to benefits from UNRWA when they relocate inside UNRWA areas of operation (Palestinian Territories, Lebanon, Syria and Jordan). See 1.3.2, 2.2.4 and 4.1.

#### 1.4.3 Palestinians with an Israeli work permit

Palestinians from the West Bank can work in Israel and in Israeli settlements in the West Bank. To do this, they require a work permit that they obtain through their employer. To make the application, they need a (biometric) identity card and a security check from the Israeli Coordinator of Government Activities in the Territories (COGAT). Persons from Gaza were unable to work easily in Israel from 2007 to 2021, but since the end of 2021 a limited number of work permits have been issued again, but only to men. 142 In March 2022, it was indicated that the 12,000 available work permits would be increased to 20,000 permits. 143 Around 80% of people from Gaza with work permits work as agricultural labourers or construction workers in Israel. 144 While commuting, Palestinians must pass through Israeli security checkpoints before entering Israel or the settlements. International organisations describe these controls as invasive and humiliating. For example, women who regularly cross the border (without a work permit) are said to be at risk of sexual harassment by Israeli border personnel. Due to the long queues at the checkpoints, people must leave early in the morning, which means that their working day, including travel time, can last up to sixteen hours.

During the lockdowns imposed by Israel in the West Bank at the start of the coronavirus pandemic, Palestinian workers in the construction, healthcare and agriculture sectors were permitted to work in Israel. To be eligible for this, employees were obliged to spend two months in the location of their workplace without being able to travel back to their place of residence. Work permits for Gaza are usually valid for three months. Persons with this type of permit must be back before 7 p.m. in order to enter Gaza. According to one source, this rule is not enforced by the Israeli authorities but it means that people stay in Israel overnight (in makeshift accommodation) to avoid delays or closure of the border. 146

## 1.4.4 Travelling abroad

It is possible for residents of Gaza and the West Bank to travel abroad but this is subject to various conditions and obstacles. Inhabitants of Gaza and the West Bank require a PA travel document/passport to travel abroad. In addition, residents of Gaza require an exit permit from the Israeli authorities and usually Hamas (via the Erez border crossing) or the Egyptian authorities (via the Rafah border crossing). Hamas controls who is permitted to travel through the Rafah border crossing on the basis of Egyptian criteria. To travel abroad via Israel, a travel permit for Israel is required from the Israeli authorities. 147 According to one source, Palestinians from Gaza who are registered in Israel's Palestinian population register are considered to

<sup>&</sup>lt;sup>142</sup> Confidential source, 24 January 2022; Confidential source, 24 January 2022.

<sup>143</sup> The Times of Israel, Israel to authorize 2,000 more entry permits for Gaza workers, 10 March 2022, consulted on 13 March 2022.

<sup>&</sup>lt;sup>144</sup> Confidential source, 24 January 2022; Confidential source, 24 January 2022.

International Trade Union Confederation (ITUC), Workers Rights in Crisis: Palestinian workers in Israel and the settlements, 12 April 2021, p. 10, 14; UNHCR, Country of Origin Information on the Situation in the Gaza Strip, Including on Restrictions on Exit and Return, 2018, p. 9.

<sup>146</sup> Confidential source, 24 January 2022.

<sup>&</sup>lt;sup>147</sup> B'Tselem, Restrictions on Movement, 11 November 2017, consulted on 10 January 2022; Confidential source, 24 January 2022; Confidential source, 24 January 2022.

be residents of Gaza even if they have a different foreign nationality. <sup>148</sup> Blue ID cards issued by Hamas are not recognised. <sup>149</sup> According to one source, Palestinians from the West Bank have access to the US consulate in Jerusalem provided they have an appointment and a travel permit issued by the Israeli authorities. <sup>150</sup>

#### Erez border crossing

The Erez border crossing is monitored by the Israeli authorities and administered on the Palestinian side by the PA. Israel and the PA have checkpoints on this border. The Hamas authorities are said to also require people leaving Gaza via Erez to have an exit permit, but according to one source this does not apply in all cases. <sup>151</sup> According to another source, this does not apply to foreigners and only local residents require an exit permit from Hamas. <sup>152</sup> Palestinians can apply for an exit permit through the Palestinian Civil Affairs Committee (PCAC) of the PA. This application then goes through the Israeli COGAT to the Coordination Liaison Administration (CLA) for applications from persons from Gaza. <sup>153</sup> It is very difficult to obtain these travel permits and most applications are refused. It is not clear how many of the applications are refused. <sup>154</sup> According to one source, Palestinians face long waiting times that exceed the official seventy days for an application. In addition, applications remain unanswered or are blocked for security reasons that are said to be arbitrarily applied, which means that applications are denied and no reason is given to applicants. <sup>155</sup>

The border crossing via Erez is said to be mainly reserved for authorised foreigners, business people and humanitarian cases such as medical emergencies. While travelling to the West Bank via Israel, Gaza residents may only take a metal-free bag with no wheels. Particular items, such as electrical appliances and chargers, food, cosmetics and toothbrushes, are not permitted. Factoring to one source, people can take only one pair of extra shoes. Every Between March 2020 and October 2021, travel via Erez was limited to six percent of the former number of trips due to coronavirus measures. During the reporting period, multiple cases were reported of people who were refused a travel permit by the Israeli authorities because their relatives from Gaza are living outside Gaza on (expired) travel permits issued by the Israeli authorities. Another source said that the travel permit issued by the Israeli authorities is valid for three months. According to another source, however, for the West Bank this is usually six months and for Gaza it is usually three months for diplomatic applications and two months for humanitarian applications.

## Transit via Jordan

```
149 Confidential source, 24 January 2022.
150 Confidential source, 13 April 2022.
151 Confidential source, 24 January 2022.
152 Confidential source, 20 April 2022.
153 Confidential source, 24 January 2022.
154 People must comply with the conditions stipulated in the Status of Authorizations, see also Error! Hyperlink reference not valid.; Confidential source, 24 January 2022; B'Tselem, Restrictions on Movement, 11 November 2017, consulted on 10 January 2022; Confidential source, 24 January 2022.
155 Confidential source, 24 January 2022.
156 B'Tselem, Restrictions on Movement, 11 November 2017, consulted on 10 January 2022; Confidential source, 24 January 2022.
156 B'Tselem, Restrictions on Movement, 11 November 2017, consulted on 10 January 2022; Confidential source, 24 January 2022.
```

<sup>148</sup> Confidential source, 24 January 2022.

 <sup>157</sup> Confidential source, 24 January 2022.
 158 Gisha, Exits by Palestinians via Erez Crossing to Israel, the West Bank, and abroad, 30 June 2021, consulted on 9 February 2022; Confidential source, 24 January 2022.

<sup>&</sup>lt;sup>159</sup> Confidential source, 24 January 2022.

 $<sup>^{160}</sup>$  Confidential source, 11 April 2022.

<sup>&</sup>lt;sup>161</sup> Confidential source, 20 April 2022.

Persons from Gaza travelling abroad through Jordan must first enter Israel via the Erez border crossing, then travel via the West Bank and via the King Hussein/Allenby Bridge border crossing. This does not require a Jordanian transit visa but does require a 'no objection letter' (adam mumana) from the Jordanian authorities. According to one source, it is difficult to obtain a 'no objection letter' in time. 162 This letter must be requested through a private company that has an office in Gaza and in the West Bank. 163 The application period usually takes three to four weeks but the application must be officially granted within fourteen days. 164 Persons travelling on this route must undergo an Israeli security check at the Erez and Allenby border crossings. Persons entering Jordan from this border crossing use a shuttle bus, operated by the PA, from the Erez border crossing to the Allenby Bridge border crossing. 165 Persons who make this journey in the regular way cannot stop in between in Israel or the West Bank. 166

## Rafah border crossing

The Rafah border crossing is controlled by Egypt and administered/governed by Hamas. The Egyptian authorities can refuse to allow people to cross the border for security reasons. 167 The exit permit application process for the Rafah border crossing can take months and people are put on a Hamas Interior Ministry waiting list. 168 The procedures and fees for entry and exit via Rafah fluctuate and are unpredictable. Besides this official registration, there are also other routes, an 'unofficial' route and a VIP route for people who do not meet the regular criteria or who want to travel faster. Influential individuals and certain Palestinian officials, such as some Hamas members, can use a separate procedure to coordinate their crossing. The normal route has no application fee and persons pay 300 USD for a one-way shuttle bus transfer between Rafah and Egypt. The costs for the unofficial route, in addition to the cost of the regular shuttle bus, are between 350 and 500 USD but can be as high as 4,000 USD. The application period for a travel permit via this route can be between ten days and three weeks. Men from the age of forty and some families are permitted to make the journey from Rafah to Egypt independently and do not have to use the shuttle bus. The VIP route is coordinated by the Egyptian company Hala and two travel agencies in the Gaza Strip. It costs around USD 750 for the outward journey and around USD 600 for the return journey to Rafah. The application time is between 48-72 hours and VIP travellers can use a VIP zone at the border crossing. VIP travellers do not travel by shuttle bus but are transported in cars. 169 According to one source, the travel time to cross the border at Rafah using the VIP route takes almost half a day and another half day to reach Cairo. 170

Due to the coronavirus pandemic, the Rafah border crossing was closed until February 2021.<sup>171</sup> Due to the poor security situation in the Egyptian province of North Sinai in the recent past, travellers were at risk of violent attacks around Rafah (such as bombing and shooting incidents, murder, kidnapping) by extremist groups,

<sup>&</sup>lt;sup>162</sup> Confidential source, 24 January 2022. 163 The Representative Office of Norway to the Palestinian Authority, Information to Gaza Citizens, consulted on 23 February 2022. See also Error! Hyperlink reference not valid. 164 Confidential source, 20 April 2022.

<sup>&</sup>lt;sup>165</sup> Confidential source, 24 January 2022.

<sup>&</sup>lt;sup>166</sup> Confidential source, 20 April 2022.

<sup>&</sup>lt;sup>167</sup> Confidential source, 24 January 2022.

 <sup>168</sup> Confidential source, 24 January 2022; Confidential source, 24 January 2022.
 169 Confidential source, 24 January 2022.

<sup>&</sup>lt;sup>170</sup> Confidential source, 24 January 2022.

<sup>&</sup>lt;sup>171</sup> Gisha, One year of "coronavirus closure" at Erez crossing, 18 March 2021, consulted on 10 January 2022; Confidential source, 24 January 2022.

among others. 172 According to one source, it is not possible for foreigners to enter Gaza through Rafah unless they are married to, a child of, or a resident of Gaza themselves. 173 Persons from Gaza do not require an Egyptian visa to enter or leave Egypt as a transit country to or from Gaza. They do require a visa, however, if Egypt is the destination country. To apply for an Egyptian visa, men from Gaza who are between the ages of 18 and 40 must obtain approval from the Egyptian authorities to apply for a visa. It is not clear whether this approval is already included with the regular visa application. 174 One source says that the Egyptian authorities generally do not grant visas to Palestinian men under the age of forty. 175

#### Checkpoints

Hamas has three checkpoints in Gaza on the road after the Erez border crossing managed by the Israeli authorities. It is not clear whether all of these checkpoints are in operation. 176 In the West Bank, there are 71 permanent checkpoints and 108 checkpoints that are sometimes manned. These checkpoints are controlled by the Israeli authorities and managed by military personnel, border police or private security personnel. The checkpoints are located both on the border with Israel and inside the West Bank. Around the perimeter of East Jerusalem, there are several checkpoints that control entry into and exit from the city. There are also checkpoints between East and West Jerusalem. In Hebron, there are five checkpoints on the boundaries of the part of the city centre that is militarily sealed off and one within this closed area. 177

<sup>&</sup>lt;sup>172</sup> UNHCR, Country of Origin Information on the Situation in the Gaza Strip, Including on Restrictions on Exit and Return, 2018, p. 27-28.

<sup>&</sup>lt;sup>173</sup> Confidential source, 24 January 2022.

<sup>&</sup>lt;sup>174</sup> Confidential source, 24 January 2022.

<sup>&</sup>lt;sup>175</sup> Confidential source, 20 April 2022.

OCHA, Gaza Strip Access and Movement, October 2020, consulted on 10 January 2022. See also Error!
 Hyperlink reference not valid.; Confidential source, 24 January 2022.

 B'Tselem, Restrictions on Movement, 11 November 2017, consulted on 10 January 2022; OCHA, West Bank

access restrictions, July 2020, consulted on 10 January 2022. See also Error! Hyperlink reference not valid.

## 2 Identity, nationality and documents

## 2.1 Obligation to provide identification

It is not clear whether there is a legal identification requirement in Gaza and in the West Bank. In the West Bank, however, Palestinians from the age of sixteen must identify themselves by means of an identity card on the street and at the various Israeli checkpoints located in this region (see also 1.4). In Gaza, Palestinians identify themselves with a PA identity card or an identity card issued by Hamas (see also 2.2.2). Palestinians must also be able to identify themselves inside Israel with a PA identity card or an identity card from East Jerusalem.<sup>178</sup>

## 2.2 Identifying documents

#### 2.2.1 Israeli authorities

#### East Jerusalem

Palestinian residents from East Jerusalem are issued a blue identity card by the Israeli authorities that is valid as a residence permit. The 'permanent' status is not guaranteed. They can apply for an Israeli travel document (laissez-passer) that is valid for three years but they are not eligible for an Israeli passport. <sup>179</sup> Applications for a travel document are made at the Population Registry offices of the Israeli Ministry of Interior. Before 2018, Palestinians from East Jerusalem could only submit this application at the office of the Israeli Population, Immigration and Border Authority in East Jerusalem. Since then, it has been possible for persons with a biometric travel document, which has been issued since June 2017, to make a new application at the other offices of this Israeli government service. <sup>180</sup> According to one source, people who have made an appointment at another office are usually referred back to make an appointment at the East Jerusalem office. <sup>181</sup> An Israeli travel document is issued one month after application.

## 2.2.2 Palestinian authorities

#### West Bank and Gaza

Since 1995, Palestinians from the West Bank and Gaza have been able to apply to the PA for identity cards (*hawiyeh*) and <sup>183</sup> passports. These documents are controlled by the Israeli Population Registry in the Palestinian Territories (see also 2.4). <sup>184</sup> Some West Bank residents are entitled to a temporary Jordanian passport

<sup>&</sup>lt;sup>178</sup> Confidential source, 24 January 2022; Confidential source, 20 April 2022

<sup>&</sup>lt;sup>179</sup> Mutaz M. Qafisheh, Who has the right to become a Palestinian Citizen? An International law analysis, Yearbook of Islamic and Middle Eastern Law Online, June 2017, p. 138; Francesca P. Albanese en Lex Takkenberg, Palestinian refugees in international law, 2020, p. 239.

<sup>180</sup> Haaretz, East Jerusalem residents will be able to renew travel documents at any other Interior Ministry offices, 14 May 2018, consulted on 2 March 2022.

<sup>&</sup>lt;sup>181</sup> Confidential source, 20 April 2022.

<sup>&</sup>lt;sup>182</sup> Immigration and Refugee Board of Canada, Query response on Israel and Palestine: Travel documents issued by the Israeli government to residents of the West Bank and Gaza Strip (2017 - March 2020), 27 April 2020, consulted on 2 March 2022.

<sup>&</sup>lt;sup>183</sup> In the Netherlands, this is a travel document and not a passport. Confidential source, 13 April 2022.

<sup>&</sup>lt;sup>184</sup> Francesca P. Albanese en Lex Takkenberg, Palestinian refugees in international law, 2020, p. 238; Home Office, Report of a Home Office Fact-Finding Mission Occupied Palestinian Territories: freedom of movement, security and human rights situation, March 2020, p. 9.

(see also *Documents issued by other authorities*). According to one source, men and women from the age of eighteen can independently apply to the Palestinian Ministry of the Interior for identity documents and travel documents.<sup>185</sup>

#### The identity card

The identity cards for Palestinians are almost identical to Israeli identity cards. However, they are green and have a green cover with PA insignia, while the first language on the card is Arabic and the second Hebrew. The identity card is issued to persons from the age of sixteen. It has a passport photo in the top left corner and bears the following information: place of birth, place of issue, whether the holder is from the West Bank or Gaza, the name of the father, mother, paternal grandfather and, on older identity cards, the holder's religion. The identity card has no expiry date. <sup>186</sup> Identity cards have a national ID number, consisting of nine digits starting with the number '9'. <sup>187</sup> Identity cards of persons born in Gaza have a number starting with the number '8'. Sometimes a number starts with the number '4' when the person concerned became a resident after the Oslo Accords. <sup>188</sup>

Children of persons with an identity card must be added to the identity cards of their father and mother before the age of five. Children up to the age of fifteen of a person with a Palestinian identity card residing in North America must be added to this identity card. <sup>189</sup> The names and identity numbers of children remain on the identity cards of the parents even after these children have been issued their own identity cards. <sup>190</sup> It is not possible to apply for identity cards from abroad. <sup>191</sup> Identity cards are issued by the Palestinian Ministry of Interior on the day of application. <sup>192</sup>

For the identity card issued by Hamas that is only valid in Gaza, see also the heading *Undocumented in Gaza*.

## Passport

The PA also issues a travel document/passport to Palestinian residents of the West Bank and Gaza and Palestinians living outside the Palestinian Territories. 193 This

- <sup>185</sup> Confidential source, 24 January 2022.
- <sup>186</sup> Immigration and Refugee Board of Canada, Israel and Palestine: Whether a Palestinian who has lived outside of the Occupied Palestinian Territory (OPT) for a number of years can return to reside in the OPT; whether the returnee requires an Israeli-issued identity card and whether it is possible to obtain the card from abroad; whether someone born in Gaza faces particular challenges returning to the OPT, 26 April 2016, consulted on 1 February 2022; Al Jazeera, The colour coded ID system for Palestinians, 18 November 2017, consulted on 6 January 2022; Confidential source, 11 April 2022.
- <sup>187</sup> Francesca P. Albanese en Lex Takkenberg, Palestinian refugees in international law, 2020, p. 238; Home Office, Report of a Home Office Fact-Finding Mission Occupied Palestinian Territories: freedom of movement, security and human rights situation, March 2020, p.
- <sup>188</sup> Norwegian Refugee Council, Undocumented and Stateless: The Palestinian Population Registry and Access to Residency and Identity Documents in the Gaza Strip, January 2012, p. 43.
- <sup>189</sup> Immigration and Refugee Board of Canada, Israel and Palestine: Whether a Palestinian who has lived outside of the Occupied Palestinian Territory (OPT) for a number of years can return to reside in the OPT; whether the returnee requires an Israeli-issued identity card and whether it is possible to obtain the card from abroad; whether someone born in Gaza faces particular challenges returning to the OPT, 26 April 2016, consulted on 1 February 2022.
- <sup>190</sup> Confidential source, 11 April 2022.
- <sup>191</sup> Immigration and Refugee Board of Canada, Israel and Palestine: Whether a Palestinian who has lived outside of the Occupied Palestinian Territory (OPT) for a number of years can return to reside in the OPT; whether the returnee requires an Israeli-issued identity card and whether it is possible to obtain the card from abroad; whether someone born in Gaza faces particular challenges returning to the OPT, 26 April 2016, consulted on 1 February 2022.
- <sup>192</sup> Home Office, Report of a Home Office Fact-Finding Mission Occupied Palestinian Territories: freedom of movement, security and human rights situation, March 2020, p. 38.
- 193 Immigration and Refugee Board of Canada, Palestine: Procedures to obtain a Palestinian passport in the West Bank; the documents to be submitted and whether the Palestinian Authority requires original identity documents or accepts photocopies; whether the passport applicant has to appear in person to apply for or receive a

passport is issued after approval has been granted by the Israeli authorities. 194 Children who were born abroad but have a Palestinian father can also apply for this passport. Since 2009, this passport has been black instead of green and is valid for five years. The passport can be obtained from any age at the passport office of the Palestinian Ministry of Interior in the district where this person lives. The applicant must be present in person. The following documents are required for the application: an identity card, four passport photos, a birth certificate, a proof of work (optional if persons want to include their profession in the passport), an application form and any previous passports. All official documents must be originals or certified copies. Applicants receive a proof of application. 195 One source says that passports are issued within three days. 196 The Immigration and Refugee Board of Canada stated in 2010 that there were no 'fraudulent passports' in the West Bank. 197

#### Identification documents for minors

According to sources, a personal identity card is issued to Palestinian minors from the age of sixteen. 198 In exceptional cases, minors from Gaza may leave Gaza unaccompanied with the consent of their parents or quardian. 199 According to one source, this does not happen often.<sup>200</sup>

#### Biometric data

The Israeli Population Registry (see also 2.4) in the Palestinian Territories and the Ministry of Civil Affairs of the PA keep records of the population registration data.<sup>201</sup> Identity cards for Palestinians from the Palestinian Territories that are issued by the PA do not have biometric data.<sup>202</sup> The PA is said to be preparing to issue biometric passports, but this has not yet been implemented.<sup>203</sup>

#### Undocumented migrants in Gaza

Since 2007, the Interior Ministry in Gaza under the control of Hamas has issued a paper identity card with a blue cover to persons in Gaza. This document does not have a registration number linked to the Israeli authorities and is not recognised by Israel.<sup>204</sup> These ID cards are in Arabic and English and have a number starting with

passport; the features of a passport; the name of the signature on page three of the passport; the duration between the application for and the reception of a passport: whether a person may have two valid passports that were issued on different dates; the prevalence of fraudulent Palestinian passports (January 2008 - April 2010), 30 April 2010, consulted on 6 January 2022.

- <sup>194</sup> Confidential source, 11 April 2022.
- <sup>195</sup> Immigration and Refugee Board of Canada, Palestine: Procedures to obtain a Palestinian passport in the West Bank; the documents to be submitted and whether the Palestinian Authority requires original identity documents or accepts photocopies; whether the passport applicant has to appear in person to apply for or receive a passport; the features of a passport; the name of the signature on page three of the passport; the duration between the application for and the reception of a passport; whether a person may have two valid passports that were issued on different dates; the prevalence of fraudulent Palestinian passports (January 2008 - April 2010), 30 April 2010, consulted on 6 January 2022.
- <sup>196</sup> Confidential source, 20 April 2022.
- <sup>197</sup> Immigration and Refugee Board of Canada, Palestine: Procedures to obtain a Palestinian passport in the West Bank; the documents to be submitted and whether the Palestinian Authority requires original identity documents or accepts photocopies; whether the passport applicant has to appear in person to apply for or receive a passport; the features of a passport; the name of the signature on page three of the passport; the duration between the application for and the reception of a passport; whether a person may have two valid passports that were issued on different dates; the prevalence of fraudulent Palestinian passports (January 2008 - April 2010), 30 April 2010, consulted on 6 January 2022.
- 198 Confidential source, 27 January 2022; Confidential source, 10 February 2022.
   199 Confidential source, 24 January 2022; Confidential source, 10 February 2022.
- <sup>200</sup> Confidential source, 24 January 2022.
- <sup>201</sup> Al Jazeera, *Not a life: Israel keeps many Palestinians without legal status*, 26 October 2021, consulted on 6 January 2022.
- <sup>202</sup> Confidential source, 20 April 2022.
- <sup>203</sup> Biometricupdate, *Biometrics shipment to Palestinian Authority mysteriously released after being confiscated*, 22 January 2021, consulted on 6 January 2022.
- Daraj News, Mustafa Ibrahim, Gaza: Yellow Passports, Blue Cards and Lost Rights, 7 October 2021, consulted on 9 February 2022.

the number '7'.<sup>205</sup> It is estimated that more than thirty thousand of these ID cards have been issued. Persons who are not registered with the Israeli Civil Administration use this document to identify themselves in Gaza – for example for the purpose of healthcare or education. This is said to mainly concern persons who entered Gaza via the Rafah border crossing on a temporary residence permit and Palestinians of other nationalities or foreigners who do not have a family reunification permit from the Israeli authorities.<sup>206</sup>

#### Refugees in the West Bank and Gaza

Certain groups of Palestinian refugees are not included in the Israeli Population Registry and therefore do not have access to ID cards and passports issued by the PA. For example, there is a group of Palestinian refugees in Gaza who have travelled to Gaza from other countries in the Middle East and have stayed there with a residence permit that has since elapsed/expired. Some persons in this group do have access to an identification card issued by Hamas, but they do not have a national number issued by the Israeli authorities. Since 2011, approximately 850 Palestinian UNRWA-registered refugees from Syria have also been staying in Gaza. They have access to UNRWA services and are registered with UNRWA and the General Administration for Palestine Arab Refugees (GAPAR) in Syria. There is also a group of refugees (estimated at between 25,000 and 50,000 persons) in the West Bank who have PA identity cards from Gaza. Israel does not consider this group to be legal residents of the West Bank. The group is said to be at risk of expulsion to Gaza, but this does not apply to Palestinians from Gaza who were already in the West Bank before September 2005.<sup>207</sup>

In October 2021, four thousand new family reunification applications were approved by the Israeli authorities. These involved undocumented foreigners married to Palestinians or undocumented relatives of Palestinians, but most of the allocations have gone to Palestinians from Gaza who are living in the West Bank and want to change their addresses. It was not clear whether this involved persons at risk of deportation to Gaza or whether those persons had been in the West Bank before September 2005. Foreign partners of Palestinians could enter the Palestinian Territories with a visa but could not renew it or obtain permanent residence permits, leaving them undocumented in the Palestinian Territories. <sup>208</sup> After 1967, people with a West Bank identity card who had been abroad for more than six years lost their residence permits. This concerns a group of approximately 140,000 persons who were entitled to this type of identity card again after the Oslo Accords in 1993. The application process for an identity card for this group goes through the Israeli authorities and can take years. <sup>209</sup> According to one source, around 3,400 applications had been approved by January 2022. <sup>210</sup>

#### Zero passports and other documents issued by the PA

Through its foreign representatives, the PA issues Palestinian travel documents to Palestinians living outside the Palestinian Territories. These are called *zero* 

<sup>&</sup>lt;sup>205</sup> Norwegian Refugee Council, Undocumented and Stateless: The Palestinian Population Registry and Access to Residency and Identity Documents in the Gaza Strip, January 2012, p. 43.

<sup>&</sup>lt;sup>206</sup> Daraj News, Mustafa Ibrahim, Gaza: Yellow Passports, Blue Cards and Lost Rights, 7 October 2021, consulted on 9 February 2022.

<sup>&</sup>lt;sup>207</sup> Francesca P. Albanese en Lex Takkenberg, *Palestinian refugees in international law*, 2020, p. 238, 240. <sup>208</sup> Al Jazeera, *Not a life: Israel keeps many Palestinians without legal status*, 26 October 2021, consulted on 6

January 2022.

209 The New York Times, *Palestinian American died handcuffed in Israeli custody, witnesses say*, 20 January 2022, consulted on 21 January 2022.

<sup>&</sup>lt;sup>210</sup> Confidential source, 24 January 2022.

passports. In some cases, this travel document can be used for travel and the issue of visas. The Netherlands recognises this travel document for the issue of visas, but not all foreign authorities recognise it as a valid travel document. Persons can use documents issued by the PA to identify themselves as Palestinian refugees – for example at UNRWA in order to receive some of their services. <sup>211</sup> According to one source, undocumented Palestinians from Gaza can also be issued a *zero passport*, but because Israel, Jordan and Egypt do not recognise this travel document, people from Gaza cannot in practice use it to enter or leave Gaza. <sup>212</sup>

Besides the zero passports, the PA is said to also issue proof of nationality to Palestinians. This document states that the person is Palestinian. The document can be used as a domestic ID card in Lebanon, for example, where unregistered Palestinians have proof of nationality issued by the PA. The zero passport travel document may possibly be used to travel abroad or to apply for a temporary residence permit. In Lebanon, for example, these documents do not grant a right of residence or access to services such as opening a bank account or civil registration with the Lebanese authorities. In Lebanon, Palestinians are said to have documents issued by the Lebanese authorities as well as the PA. This information has not been confirmed by other sources. <sup>213</sup>

#### 2.2.3 Jordanian authorities

Palestinians from the West Bank who lost their Jordanian citizenship in 1988 when Jordanian sovereignty over the area came to an end can apply for a temporary Jordanian passport that is valid for five years. This passport is regarded as a travel document by Jordan and most other countries and is different from the Jordanian passport, which has a national identity number. There are Palestinians from the West Bank who, in addition to this temporary passport, also have a Palestinian passport. Persons with a temporary Jordanian passport have access to a green card that permits them to stay in Jordan for up to two months. In addition, persons with this green card do not need to receive an entry and exit stamp from the Israeli authorities in their passport or travel document when they enter and leave Jordan via the West Bank. Persons from East Jerusalem who do not have access to a PA passport are said to be more likely to use the temporary Jordanian travel document. <sup>215</sup>

#### 2.2.4 UNRWA

UNRWA issues a family registration card (*bitaqa tashjil al ayla*) to registered Palestinian refugees. This document serves as proof of registration with UNRWA and grants access to UNRWA services. UNRWA registration provides access only to UNRWA services and not to legal residence, refugee status or (physical) protection in the country where a person is registered. The following documents can also be issued on request: a family registration certificate (*shahada tashjil al ayla*) and a family composition print-out. These are documents that specify which UNRWA

<sup>&</sup>lt;sup>211</sup> Ministry of Foreign Affairs, *Thematisch Ambtsbericht Palestijnen in Libanon*, January 2021, p. 29 – 30; Confidential source, 13 April 2022.

<sup>&</sup>lt;sup>212</sup> Euro-Mediterranean, Human Rights Monitor, *The Gaza Strip: Undocumented Citizens*, March 2021, p. 24.

<sup>&</sup>lt;sup>213</sup> Ministry of Foreign Affairs, *Thematisch Ambtsbericht Palestijnen in Libanon*, January 2021, p. 26.

<sup>214</sup> Mutaz M. Qafisheh, Who has the right to become a Palestinian Citizen? An International law analysis, Yearbook of Islamic and Middle Eastern Law Online, June 2017, p. 120.

<sup>&</sup>lt;sup>215</sup> Danish National ID Centre, Jordan: Issuance of Jordanian documents to stateless Palestinians, 16 April 2020, p. 2, 4.

 $<sup>^{216}</sup>$  Confidential source, 28 February 2022.

services have been received by a person and where a registered person was born, is living or (temporarily) staying if this is no longer the original place of residence. According to one source, a family registration certificate basically contains the same information as a family registration card. These documents are only issued if the person in question has given their approval. The data is based on information contained in the UNRWA Refugee Registration Information System (RRIS). New documents cannot be requested online. It is possible to contact UNRWA from abroad and apply for a new family registration card. This can also be done by granting a power of attorney to a family member or a lawyer; the person with the power of attorney will then receive the new family registration card. UNRWA documents are not issued to the group of Palestinians who were internally or externally displaced in or after 1967. UNRWA can also issue individual ID cards for persons working for UNRWA, including Palestinian refugees.<sup>217</sup>

## Family Registration Card

The family registration card is an A5-format card with no expiry date. It contains the following information in Arabic and English: name of the head of the family, family registration number, former family registration number, place of origin<sup>218</sup>, UNRWA location of registration, place of residence in the country or territory and residential address. The following information is provided for each individual family member: individual registration number, name, date of birth, gender, relationship to the head of the family, marital status and registration category in UNRWA.<sup>219</sup> The head of the family is issued a family registration card. A new family registration card is issued when a new registration is made or when a new family is formed from a previous family, with children being given their own card when they marry and start a family. Some orphans or minors have their own family registration cards.<sup>220</sup>

#### 2.2.5 UNHCR

UNHCR does not issue documents to Palestinians that come under the UNRWA areas of operation, including the Palestinian Territories (Gaza and West Bank), Lebanon, Jordan and Syria. Palestinian refugees registered with UNHCR in other countries do not have access to UNHCR services such as shelter and financial or medical assistance in the UNRWA areas of operations.<sup>221</sup>

## 2.3 Nationality

#### Palestinian Territories

- <sup>217</sup> The Danish Immigration Service, *Palestinian Refugees: Access to registration and UNRWA services, document and entry to Jordan*, June 2020, p. 21; Ministry of Foreign Affairs, *Thematisch Ambtsbericht Palestijnen in Libanon*, January 2021, p. 24 25.
- <sup>218</sup> According to UNRWA, Palestinian refugees are those persons and their descendants who lost their places of residence in British Mandate Palestine during the period from 1 June 1946 to 15 May 1948. UNRWA, *Consolidated Eligibility and Registration Instructions (CERI)*, 2006, p. 3.
- There are different categories of persons who can register with UNRWA. Some persons registered with UNRWA do not belong to the refugee group but are entitled to registration and assistance from UNRWA. Palestinian refugees, relatives of refugees who are not themselves refugees, frontier villagers, the poor from Jerusalem and Gaza and compromised cases are entitled to UNRWA registration and UNRWA relief efforts. UNRWA, *Eligibility and registration Q&A*, 2020, p. 1-2; Palestinians displaced as a result of the 1967 conflict, persons that fall under the UNRWA mandate for humanitarian or other policy reasons, persons eligible for the UNRWA Microfinance and Microenterprise Programme (MMP), persons employed by UNWRA and their family members and persons living in refugee camps and communities are entitled to UNRWA assistance but not to UNRWA registration. UNRWA, *Eligibility and registration Q&A*, 2020, p. 2; UNRWA, *Consolidated Eligibility and Registration Instructions (CERI)*, 2006, p. 3-4.
- <sup>220</sup> Ministry of Foreign Affairs, *Thematisch Ambtsbericht Palestijnen in Libanon*, January 2021, p. 25.
- <sup>221</sup> Ministry of Foreign Affairs, *Thematisch Ambtsbericht Palestijnen in Libanon*, January 2021, p. 25.

The PA issues identification documents that show that the holder has the Palestinian nationality. This nationality is not recognised by the Netherlands. In the absence of a widely recognised Palestinian state and associated nationality law, Palestinians living under the PA in the West Bank and Gaza do not have an internationally recognised certificate of nationality for the Palestinian Territories. <sup>222</sup> Existing nationality law is based on former Ottoman, British, Jordanian and Egyptian law. According to the legal framework (Article 27 of the 2007 General Elections Act), a person is Palestinian if he or she:

- Was born in the British Mandate of Palestine or was entitled to Palestinian nationality according to the laws of that period (1920-1948). Descendants of this person are also entitled to Palestinian nationality regardless of where they were born<sup>223</sup>;
- Born in Gaza, the West Bank or East Jerusalem;
- He or she is the spouse of a person recognised as Palestinian according to the above rules.<sup>224</sup>

There is a 1995 Palestinian Draft Nationality Law and a 2011 PLO Draft Nationality Law. The latter stipulates that everyone who has a Palestinian ID card from the PA can become a Palestinian citizen of a sovereign Palestinian state. According to this law, Palestinians from East Jerusalem, refugees from the 1948 and 1967 conflicts and Palestinian nationals of Israel would also be entitled to Palestinian nationality.<sup>225</sup>

Article 12 of the Jordanian Nationality Law of 1954 stipulates that Jordanian citizenship is granted to foreigners after four years of residence in Jordan provided they can demonstrate a fluent command of the Arabic language and good conduct and reputation. <sup>226</sup> In practice, the Jordanian law of 1954 and the Palestinian Citizenship Order of 1925 have not been enacted since 1967 due to Israeli regulations. <sup>227</sup>

## Losing and regaining Palestinian citizenship

There are currently no regulations for the loss of citizenship.<sup>228</sup> According to former Jordanian law (in this case Article 18 of the Nationality Law of 1954), persons lose their Palestinian nationality if they have completed military service for another country without the consent of the (Palestinian) ministerial cabinet and refuse to terminate it, work as a civil servant for another country and refuse to serve as an

<sup>&</sup>lt;sup>222</sup> Although a person's nationality (as ethnicity or background) may be separate from his or her citizenship, in this official country of origin report nationality also means citizenship.

<sup>223</sup> In practice it is not possible for Palestinian women to pass on their nationality to their children. See also Palestinians from abroad and other foreigners.

OHCR, The Contribution of the State of Palestine to the Special Rapporteur's Thematic Report on Racism, Racial Discrimination, and Xenophobia in the Contest of Laws, Policies and Practices Relating to Citizenship, Nationality and Immigration, p. 2.

<sup>&</sup>lt;sup>225</sup> UNDP, State of Palestine: Gender Justice & the Law, 2018, p. 21.

<sup>&</sup>lt;sup>226</sup> OHCR, The Contribution of the State of Palestine to the Special Rapporteur's Thematic Report on Racism, Racial Discrimination, and Xenophobia in the Contest of Laws, Policies and Practices Relating to Citizenship, Nationality and Immigration, p. 2.

<sup>227</sup> According to Article 2 of the Immigration Act of 1941 (British Mandate period), no nationality is barred from immigration, entry or residence in the Palestinian State. According to Article 5 of this law, entry or immigration to the country may be refused if this person has no legal capacity, has been convicted of murder, or threatens the peace and order of Palestine. OHCR, The Contribution of the State of Palestine to the Special Rapporteur's Thematic Report on Racism, Racial Discrimination, and Xenophobia in the Contest of Laws, Policies and Practices Relating to Citizenship, Nationality and Immigration, p. 3; Mutaz M. Qafisheh, Who has the right to become a Palestinian Citizen? An International law analysis, Yearbook of Islamic and Middle Eastern Law Online, June 2017, p. 130-131.

<sup>&</sup>lt;sup>228</sup> Mutaz M. Qafisheh, Who has the right to become a Palestinian Citizen? An International law analysis, Yearbook of Islamic and Middle Eastern Law Online, June 2017, p. 130-131.

civil servant for the (Palestinian) government or are serving an enemy state. 229 In practice, this legislation is ineffective and this registration of citizenship is under the control of the Israeli authorities. <sup>230</sup> The conditions for regaining Palestinian nationality are not clear.

#### Israel

Israeli nationality can be obtained in two ways.

Firstly, based on the following:

- A person was born in Israel and has a father or mother who is an Israeli
- A person was born abroad and has a father or mother who is an Israeli citizen.<sup>231</sup> According to one source, if this parent was also born abroad, the application would be easier to obtain on the basis of the Law of Return<sup>232</sup>;
- A person is stateless in Israel. This person can apply for Israeli citizenship between his/her eighteenth and twenty-fifth birthday and must have resided in Israel for five consecutive years prior to the application. <sup>233</sup> According to one source, there were no known cases of persons having acquired nationality using this option<sup>234</sup>;
- Based on the Law of Return for a person who was born to a Jewish mother or who has converted to Judaism and is not a member of any other religion. This also includes a child or grandchild of a Jewish person and the spouse of the child or grandchild of a Jewish person. This right may be denied if a person engages in activities against the Jewish population, endangers Israel's public health or security or has a criminal record<sup>235</sup>;
- Former citizens of the British Mandate of Palestine who resided in Israel when the state was established in 1948 and the entry into force of the nationality law in 1952.<sup>236</sup> According to one source, this does not apply to Palestinian refugees.<sup>237</sup>
- Minors adopted by an Israeli citizen.<sup>238</sup>

Secondly, following the approval of the Israeli Ministry of the Interior. Israeli nationality can be obtained using this procedure in the following two ways:

Through naturalisation. This person must speak Hebrew and have resided in Israel for five years prior to the application and must be entitled to permanent residence in Israel and have the intention to reside in Israel and have renounced other nationalities when assuming Israeli citizenship. 239 According to one source, these are mainly non-Jewish spouses of Israeli

<sup>&</sup>lt;sup>229</sup> OHCR, The Contribution of the State of Palestine to the Special Rapporteur's Thematic Report on Racism, Racial Discrimination, and Xenophobia in the Contest of Laws, Policies and Practices Relating to Citizenship, Nationality and Immigration, p. 2.

<sup>&</sup>lt;sup>230</sup> Mutaz M. Qafishen, *Who has the right to become a Palestinian Citizen? An International law analysis*, Yearbook of Islamic and Middle Eastern Law Online, June 2017, p. 131.

<sup>&</sup>lt;sup>231</sup> Israel Ministry of Foreign Affairs, Acquisition of Israeli Nationality, 1 January 2010, consulted on 4 January 2022; Confidential source, 26 January 2022. <sup>232</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>233</sup> Israel Ministry of Foreign Affairs, Acquisition of Israeli Nationality, 1 January 2010, consulted on 4 January 2022; Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>234</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>235</sup> Israel Ministry of Foreign Affairs, *Acquisition of Israeli Nationality*, 1 January 2010, consulted on 4 January 2022; Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>236</sup> Israel Ministry of Foreign Affairs, Acquisition of Israeli Nationality, 1 January 2010, consulted on 4 January 2022; Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>237</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>238</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>239</sup> Israel Ministry of Foreign Affairs, Acquisition of Israeli Nationality, 1 January 2010, consulted on 4 January 2022; Confidential source, 26 January 2022.

- persons who are granted Israeli nationality in this way; not many other persons could be granted the Israeli nationality<sup>240</sup>;
- In special cases, the Ministry of the Interior can grant Israeli nationality for example, to foreign athletes.<sup>241</sup>

Displaced Palestinians whose former residence is inside Israel's current borders are not eligible for the Israeli nationality. Residents of the Palestinian Territories are not eligible for the Israeli nationality or an Israeli residence permit, not even through family reunification. The temporary *Citizenship Law* of 2003 that prevented family reunification expired in July 2021. Although new applications for family reunification have been submitted, it is not clear whether these have actually been granted by the Israeli Ministry of Interior. According to one source, these applications were still being handled according to legislation that had expired. During the reporting period, there was a new process aimed at getting the same Citizenship Law adopted by the Israeli parliament. The new law was adopted at the beginning of March 2022.

## Losing and regaining Israeli citizenship

Israeli citizenship is annulled if a person leaves Israel illegally for particular countries or adopts the nationality of one of those countries, commits an act contrary to loyalty to Israel or has acquired Israeli citizenship on a false basis.<sup>246</sup> It is possible to renounce the Israeli nationality if permission to do so is granted by the Ministry of Interior. If a person is not Jewish and has another nationality alongside the Israeli nationality, a source says that this should not be a problem - with the exception of certain countries. If a person is Jewish or is obliged to complete military service in Israel, it would be more difficult to renounce Israeli nationality. This person must then prove his/her residence status in another country. 247 It is also possible to obtain a second nationality (or multiple nationalities), but only after obtaining the Israeli nationality. However, the foreign spouses of Israeli citizens are said to be able to retain their own nationality.<sup>248</sup> The conditions for regaining the Israeli nationality are not clear. According to one source, a Jewish person can return to Israel under the Law of Return (see the sections above under the heading Israel). When this option cannot be used and many years have passed since the renunciation of Israeli nationality, the source says that it is difficult to regain that nationality. The source is aware of one case from twenty years ago, when an Arab citizen of Israel who returned from the West Bank with her children did manage to regain the Israeli nationality.<sup>249</sup>

#### Statelessness

<sup>&</sup>lt;sup>240</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>241</sup> Israel Ministry of Foreign Affairs, Acquisition of Israeli Nationality, 1 January 2010, consulted on 4 January 2022; Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>242</sup> OHCR, The Contribution of the State of Palestine to the Special Rapporteur's Thematic Report on Racism, Racial Discrimination, and Xenophobia in the Contest of Laws, Policies and Practices Relating to Citizenship, Nationality and Immigration, p. 4-5.

<sup>&</sup>lt;sup>243</sup> Middle East Institute, After 18 years, Palestinian families can finally live together: The end of Israel's 2003 citizenship law, 10 August 2021, consulted on 5 January 2022.

<sup>&</sup>lt;sup>244</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>245</sup> The Times of Israel, New 'Citizenship Law' advances, months after ban on Palestinian spouses lapsed, 9 January 2022, consulted on 7 March 2022; BBC Monitoring, Briefing: Israel votes to re-ban Palestinian family unification, 11 March 2022, consulted on 12 March 2022.

<sup>&</sup>lt;sup>246</sup> The list relates to non-Jewish persons from Iran, Iraq, Syria and Lebanon. USDoS, Israel (Includes West Bank and Gaza) 2020 International Religious Freedom Report, p. 8; Israel Nationality Law 5712-1952, 1952, p. 5-8; Confidential source, 26 January 2022.

 <sup>&</sup>lt;sup>247</sup> Confidential source, 26 January 2022.
 <sup>248</sup> Israel Nationality Law 5712-1952, 1952, p. 5-8; Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>249</sup> Confidential source, 26 January 2022.

The PA regards Palestinians in the Palestinian Territories as Palestinians of Palestinian nationality. This also applies to Palestinians with multiple nationalities. The Israeli authorities regard Palestinians living in Israel as Arab nationals with Israeli nationality and Palestinians from East Jerusalem as 'permanent' residents of Jerusalem (see also 2.4 and 2.5). According to one source, the Palestinian nationality is not recognised by the Israeli authorities. International organisations have different descriptions for Palestinians. UNRWA does not refer to nationality and calls refugees from the British Mandate of Palestine 'Palestine refugees' (see also 1.2.3).

### Arab citizens of Israel

According to one source, it is not known whether Arab citizens of Israel or persons of Israeli nationality can also hold Palestinian citizenship of the PA. There are no known examples of this, according to the source. It is not clear whether these persons can register with the PA because this registration is controlled by the Israeli authorities (see also 2.4).<sup>253</sup>

## Palestinians from abroad and other foreigners

Under Article 12 of the Jordanian Nationality Law of 1954, which has no practical effect in the Palestinian Territories, Jordanian citizenship is granted to foreigners after four years of residence provided they can demonstrate a fluent command of the Arabic language and good conduct and reputation. According to this article, in case of marriage citizenship is granted regardless of gender.<sup>254</sup> In practice, this citizenship registration is under the control of the Israeli authorities<sup>255</sup>

Since 2010, Palestinian women married to persons who are not Palestinian have been able to pass on their Palestinian nationality to their children under the age of sixteen, based on an order issued by the Palestinian Ministry of the Interior. A foreign partner can apply for Palestinian nationality in the West Bank by means of an application for family reunification. In practice, this registration is controlled by Israel and it is practically impossible for a foreigner with a Palestinian partner to obtain a visa or residence permit for the Palestinian Territories. In Gaza, it is not possible for women to pass on their nationality to their foreign husband or to their children. Only men can pass on their Palestinian nationality to their wives and children.

## 2.4 Registration

<sup>250</sup> Mutaz M. Qafisheh, Who has the right to become a Palestinian Citizen? An International law analysis, Yearbook of Islamic and Middle Eastern Law Online, June 2017, p. 131.

<sup>&</sup>lt;sup>251</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>252</sup> Confidential source, 28 February 2022. <sup>253</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>254</sup> OHCR, The Contribution of the State of Palestine to the Special Rapporteur's Thematic Report on Racism, Racial Discrimination, and Xenophobia in the Contest of Laws, Policies and Practices Relating to Citizenship, Nationality and Immigration, p. 2.

<sup>255</sup> Mutaz M. Qafisheh, Who has the right to become a Palestinian Citizen? An International law analysis, Yearbook of Islamic and Middle Eastern Law Online, June 2017, p. 131.

UNDP, State of Palestine: Gender Justice & the Law, 2018, p. 21.

<sup>&</sup>lt;sup>257</sup> Confidential source, 11 April 2022.

<sup>258</sup> OHCR, The Contribution of the State of Palestine to the Special Rapporteur's Thematic Report on Racism, Racial Discrimination, and Xenophobia in the Contest of Laws, Policies and Practices Relating to Citizenship, Nationality and Immigration, p. 2, 5-6.

<sup>&</sup>lt;sup>259</sup> UNRWA, Regional Information Relating to Women's Nationality, consulted on 7 March 2022. See also Error! Hyperlink reference not valid.; Confidential source, 24 January 2022.

The population of the West Bank was almost 3 million in July 2021 (including around 432,000 Israeli settlers in 2019) and the population of Gaza was approximately 2.1 million. <sup>260</sup> In 2019, the population of East Jerusalem included approximately 350,000 Palestinians. <sup>261</sup> Palestinians have been included in the Israeli Population Registry for the Palestinian Territories since 1967. They are subdivided into Jerusalem, the West Bank and Gaza. Persons registered as residents of the West Bank or Gaza are entitled to an identity card and passport issued by the PA. <sup>262</sup> Palestinians registered in East Jerusalem fall under the control of Israel's Interior Ministry. The population registry of the Ministry of the Interior (PA) is a digital database based on the Israeli population registry for Palestinians. Adjustments and new registrations in this system can be viewed by the Israeli authorities. <sup>263</sup> For UNRWA registrants, see also 2.2 and Chapter 4.

### Registration of children

Children of a person with a residence permit and an identity card from Jerusalem are not automatically eligible for this residence permit. For this to be possible, an application must be made to the Israeli Ministry of the Interior before a child reaches the age of fourteen. After this, it is no longer possible to submit an application. After the age of eighteen, it is not possible to be processed for registration or family reunification. The registration process for children can take many years and the renewal of a residence permit can take up to one year. As a result, there are said to be around ten thousand unregistered children in East Jerusalem.<sup>264</sup>

## UNRWA registration

UNRWA registers persons and their descendants, through the male line, who meet the criteria of the Consolidated Eligibility and Registration Instructions (CERI). New applications for this registration are still being processed. Any Palestinians are registered as refugees with UNRWA. At the end of 2020, there were 871,537 registered refugees in the West Bank and 1,476,706 registered refugees in Gaza in the Palestinian Territories (see also Chapter 4). There are also approximately 379,000 other individuals registered with UNRWA who are entitled just to UNRWA services.

Palestinians registered with UNRWA in another country can formally transfer their registration to another territory<sup>267</sup> under the mandate of UNRWA. This does not happen in practice, however, because re-registration requires permission from the local authorities in the area where the person de facto lives and wants to move his/her registration.<sup>268</sup> According to one source, Palestinians – including Palestinian refugees – can only legally reside in the Palestinian Territories if they are registered in the Israeli Population Registry.<sup>269</sup> Persons who are registered with UNRWA can gain access to UNRWA services without relocating their registration to the new area. These persons therefore remain registered with UNRWA in the other country. It is

<sup>&</sup>lt;sup>260</sup> CIA The World Factbook, *West Bank, Gaza Strip*, 16 November 2021, consulted on 29 November 2021.

 <sup>&</sup>lt;sup>261</sup> B'Tselem, East Jerusalem, 27 January 2019, consulted on 11 February 2022.
 <sup>262</sup>Home Office, Report of a Home Office Fact-Finding Mission Occupied Palestinian Territories: freedom of movement,

security and human rights situation, March 2020, p. 90; Confidential source, 24 January 2022.
 Home Office, Report of a Home Office Fact-Finding Mission Occupied Palestinian Territories: freedom of movement, security and human rights situation, March 2020, p. 32-34, 90.

<sup>&</sup>lt;sup>264</sup> Al Jazeera, *Israeli law tears Palestinian families apart*, 19 July 2013, consulted on 6 January 2022.

<sup>&</sup>lt;sup>265</sup> Confidential source, 28 February 2022.

<sup>&</sup>lt;sup>266</sup> Confidential source, 28 February 2022.

 $<sup>^{\</sup>rm 267}$  UNWRA is active in the Palestinian Territories, Lebanon, Jordan and Syria.

<sup>&</sup>lt;sup>268</sup> Ministry of Foreign Affairs, *Thematisch Ambtsbericht Palestijnen in Libanon*, January 2021, p. 13.

<sup>&</sup>lt;sup>269</sup> Confidential source, 28 February 2022.

not possible to have multiple UNRWA registrations in different areas. When registered persons move house, they usually stick to the area of registration where they or their parents/grandparents registered with UNRWA. That means that their registration information does not always reflect the location where they are currently residing.<sup>270</sup> If persons obtain a (different) nationality, it does not affect their right to register with UNRWA.<sup>271</sup>

#### 2.5 Right of residence

#### East Jerusalem

Palestinian residents of East Jerusalem are regarded by the Israeli authorities as foreigners and are 'permanent residents' of the city. They only have access to documents that are specific to residents of East Jerusalem and they are not citizens of Israel (see also 2.2).<sup>272</sup> Right of residence is registration-dependent; these registrations have been included since the Israeli census of 1969. Some of this group have been granted Israeli citizenship on the basis of naturalisation.<sup>273</sup> Palestinians in East Jerusalem run the risk of losing their residence permits if they leave Jerusalem and Israel for a longer period of time (see also 1.4.4). This also applies when they want to settle in the West Bank. If they lose this residence permit, they run the risk of becoming stateless because they do not have Israeli nationality or access to a Palestinian residence permit.<sup>274</sup> According to one source, they also lose their residence permit when they acquire another nationality.<sup>275</sup>

It is said that the PA does not want to issue Palestinian identity cards to Palestinians from East Jerusalem because it wants to prevent the Israeli authorities from deporting Palestinians from East Jerusalem to the West Bank. However, it does treat this group as Palestinian citizens - except when it involves the jurisdiction of the police and the court.<sup>276</sup> More than 14,000 Palestinians are believed to have lost their residence permits for East Jerusalem between 1967 and 2011. Due to restrictive family reunification legislation, many families are living apart, between East Jerusalem and other parts of the Palestinian Territories.<sup>277</sup> Women with an identity card from Gaza or the West Bank who have a husband from East Jerusalem are said to be restricted in their freedom of movement. They have no access to work and no opportunity to obtain a driving licence and often stay at home for fear of being deported from the city.278

West Bank and Gaza

<sup>&</sup>lt;sup>270</sup> Ministry of Foreign Affairs, *Thematisch Ambtsbericht Palestijnen in Libanon*, January 2021, p. 13; Confidential source, 28 February 2022.

<sup>&</sup>lt;sup>271</sup> Confidential source, 28 February 2022.

<sup>&</sup>lt;sup>272</sup> OHCR, The Contribution of the State of Palestine to the Special Rapporteur's Thematic Report on Racism, Racial Discrimination, and Xenophobia in the Contest of Laws, Policies and Practices Relating to Citizenship, Nationality and Immigration, p. 5; Francesca P. Albanese and Lex Takkenberg, Palestinian refugees in international law, 2020, p. 239.

<sup>&</sup>lt;sup>273</sup> Mutaz M. Qafisheh, Who has the right to become a Palestinian Citizen? An International law analysis, Yearbook of Islamic and Middle Eastern Law Online, June 2017, p. 137-138.

Mutaz M. Qafisheh, Who has the right to become a Palestinian Citizen? An International law analysis, Yearbook of Islamic and Middle Eastern Law Online, June 2017, p. 138; Francesca P. Albanese en Lex Takkenberg, Palestinian refugees in international law, 2020, p. 239.

<sup>&</sup>lt;sup>275</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>276</sup> Mutaz M. Qafisheh, Who has the right to become a Palestinian Citizen? An International law analysis, Yearbook of Islamic and Middle Eastern Law Online, June 2017, p. 139; Francesca P. Albanese and Lex Takkenberg, Palestinian refugees in international law, 2020, p. 239.

<sup>&</sup>lt;sup>277</sup> Al Jazeera, *Israeli law tears Palestinian families apart*, 19 July 2013, consulted on 6 January 2022. <sup>278</sup> Luna Saadeh, *The Israeli Occupation and Palestinians' Right to Choice in Marriage*, Kohl Journal, 2018, consulted on 6 January 2022.

According to the Israeli authorities, Palestinians in Gaza and in the West Bank who took part in the Israeli census in 1967 are entitled to residency in the Palestinian Territories. Between 1967 and 1994, hundreds of thousands of Palestinians lost their residence permits because they were not included in this census and had left the Palestinian Territories (for too long) according to their residence permits. As a result, many Palestinians live illegally in the West Bank and Gaza. <sup>279</sup> In January 2021, it was announced that Israel would grant residence permits to approximately 13,500 Palestinians from the West Bank and Gaza. <sup>280</sup> It is not clear whether these permits were actually issued.

# Foreigners in the Palestinian Territories

The Israeli authorities do not grant right of residence to non-Jewish foreigners in the Palestinian Territories. Up to 2007, the Israeli authorities extended tourist visas for persons who were born in the Palestinian Territories and hold foreign passports. After this date, persons with foreign passports who had lived in the Palestinian Territories for many years could not re-enter the Palestinian Territories. For example, foreigners who come to visit family lose their right to re-enter the Palestinian Territories when they leave the Palestinian Territories. <sup>281</sup>

#### Israel

Arab citizens of Israel are entitled to Israeli citizenship; these are mostly Palestinians who remained in Israel after 1948. Palestinians from the Palestinian Territories can apply for a residence permit in Israel by means of the family reunification process. From 2003 onward, these applications were made almost impossible by the Citizenship Law (see also 2.3).

## Marriage of a Palestinian to a foreigner

Because there is no Palestinian nationality law, Jordanian nationality law applies. However, this is not applied in practice (see also 2.3). Under Article 12 of the Jordanian Nationality Law of 1954, Jordanian citizenship is granted to foreigners after four years of residence. In practice, these applications are controlled by Israel and it is practically impossible for a foreigner with a Palestinian partner to obtain a visa or residence permit for the Palestinian Territories. <sup>282</sup>

In Israel, it is not possible to marry a person who belongs to another religious group or to enter into a civil marriage. (Interreligious) marriages that are concluded abroad are, however, recognised in Israel.<sup>283</sup> Palestinians from the Palestinian Territories were often unable to obtain an Israeli residence permit or nationality in East Jerusalem or Israel even if they had a partner with an Israeli residence permit

<sup>&</sup>lt;sup>279</sup> Francesca P. Albanese and Lex Takkenberg, *Palestinian refugees in international law*, 2020, p. 23; The Times of Israel, *Some Palestinians get legal status after years in Gaza limbo*, 10 January 2022, consulted on 7 March 2022.

<sup>&</sup>lt;sup>280</sup> The Independent, Some Palestinians get legal status after years in Gaza limbo, 10 January 2022, consulted on 10 January 2022.

OHCR, The Contribution of the State of Palestine to the Special Rapporteur's Thematic Report on Racism, Racial Discrimination, and Xenophobia in the Contest of Laws, Policies and Practices Relating to Citizenship, Nationality and Immigration, p. 4.

<sup>&</sup>lt;sup>282</sup> OHCR, The Contribution of the State of Palestine to the Special Rapporteur's Thematic Report on Racism, Racial Discrimination, and Xenophobia in the Contest of Laws, Policies and Practices Relating to Citizenship, Nationality and Immigration, p. 2, 5-6; Mutaz M. Qafisheh, Who has the right to become a Palestinian Citizen? An International law analysis, Yearbook of Islamic and Middle Eastern Law Online, June 2017, p. 134; Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>283</sup> U.S. Embassy in Israel, *Marriage*, consulted on 7 March 2022. See also **Error! Hyperlink reference not valid.**; It is possible to enter into a civil marriage if both partners do not belong to a religion, but is said to be a rare occurrence. Haaretz, *Who would you be allowed to marry in Israel today?*, 10 April 2018, consulted on 13 March 2022.

or nationality. This is due to the Citizenship Law (see also 2.3).<sup>284</sup> According to one source, Palestinians from the Palestinian Territories can marry an Israeli and are entitled to an Israeli residence permit - but not to Israeli citizenship - if they reside in Israel. This residence permit must be renewed annually; sometimes these persons have access to temporary residence permits that are valid for a longer period of time. To marry an Israeli, Palestinian women must be over 25 years old and Palestinian men over 35 years old. They and their family members must not pose a security risk from an Israeli perspective. The presence of children does not matter; children from this marriage become Israeli citizens or receive a residence permit through their Israeli parent. Persons from Gaza cannot obtain an Israeli residence permit, not even through marriage. This is apparently only possible on humanitarian grounds, but according to one source it does not happen in practice.<sup>285</sup>

#### Undocumented Palestinians

Undocumented persons residing in Israel or residents of East Jerusalem who violate their right to a residence permit can be expelled from the country by the Israeli authorities. According to one source, persons who reside in Israel on an expired visa due to the presence of their family are still entitled to apply for a residence permit. To do this, they must provide specific documents from their country of origin and it can be difficult to obtain these types of documents. It is difficult for Palestinians who are afraid to return to the Palestinian Territories or who are afraid of causing problems with the PA to provide these documents. It is not clear which documents this involved.<sup>286</sup>

<sup>&</sup>lt;sup>284</sup> Middle East Institute, After 18 years, Palestinian families can finally live together: The end of Israel's 2003 citizenship law, 10 August 2021, consulted on 5 January 2022.

285 Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>286</sup> Confidential source, 26 January 2022.

# 3 Human rights

Due to the complexity of the legal system and the legislation in the Palestinian Territories, before discussing the observance and violations of human rights it is important to first describe the context. Legislation in the Palestinian Territories consists of a combination of British, Jordanian, Egyptian, Ottoman and Palestinian laws, including Islamic law (*Sharia*) and Israeli military orders (orders). Due to the division of power between Gaza and the West Bank, different laws are adopted by the PA and by Hamas.<sup>287</sup> In Gaza, the Palestinian, British and Egyptian penal codes apply. These laws can all be applied, but if they contradict each other, the most recent law will prevail.<sup>288</sup> In the West Bank, both the Palestinian and Jordanian penal codes apply, but presidential decrees also have to be taken into account.<sup>289</sup>

Palestinians in the West Bank come under both Palestinian jurisdiction and the Israeli military judiciary. In Area C, the Israeli army has exclusive control. Palestinians arrested by Israeli forces are tried under Israeli military law. According to Freedom House, the two legal systems do not operate independently. The Israeli civil courts have jurisdiction over Israeli settlers in the West Bank.<sup>290</sup> Military courts have jurisdiction over offences related to security and public order, especially traffic violations, but also criminal offences not defined as security violations.<sup>291</sup> In Gaza, civilians are tried in civil and religious courts and also (unlawfully) by Hamas in military courts.<sup>292</sup> Palestinians in East Jerusalem fall under the Israeli civil justice system, including Palestinians from Gaza who are arrested by the Israeli authorities. Palestinians in the West Bank suspected of crimes against other Palestinians are usually tried by Palestinian courts.<sup>293</sup>

## 3.1 Compliance and violations

## 3.1.1 Freedom of speech

Freedom House says that political and civil liberties are severely restricted in Gaza.<sup>294</sup> While there is some scope for openly political discussions in the West Bank, according to Freedom House, the PA rules in an authoritarian way. Journalists, critics and human rights activists who criticise the PA run the risk of being intimidated and repressed (see also 3.2.1).<sup>295</sup> According to the Arab Center for the Advancement of Social Media (*7amleh*), between 6 and 19 May 2021 more than five hundred cases

<sup>&</sup>lt;sup>287</sup> DCAF, Palestinian Women and Penal Law: Policy Brief Ramallah and Geneva, May 2012, p. 1; UNDP, State of Palestine: Gender Justice & the Law, 2018, p. 10;

<sup>&</sup>lt;sup>288</sup> Women's Learning Partnership, *Palestine (Gaza) Penal code, British Mandate Criminal code*, consulted on 3 March 2022; EASO, *COI Query Palestine*, 10 January 2020, p. 3-4.

<sup>&</sup>lt;sup>289</sup> DCAF, *Palestinian Women and Penal Law: Policy Brief Ramallah and Geneva*, May 2012, p. 1.

<sup>&</sup>lt;sup>290</sup> Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021; The New York Times, Critic's Death Puts Focus on Palestinian Authority's Authoritarianism, 7 July 2021, consulted on 27 July 2021.

<sup>&</sup>lt;sup>291</sup> B'Tselem, *The Military Courts*, 11 November 2017, consulted on 7 April 2022.

<sup>&</sup>lt;sup>292</sup> Human Rights Watch, *Abusive System: Failures of Criminal Justice in Gaza*, September 2012, p. 13.

<sup>&</sup>lt;sup>293</sup> DCIP, *No way to treat a child: Palestinian children in the Israeli military detention system*, April 2016, p. 10. <sup>294</sup> Freedom House, *Gaza Strip: Freedom in the World 2020 Country Report*, consulted on 25 February 2022.

<sup>&</sup>lt;sup>295</sup> Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021; Al Jazeera, Why is the Palestinian Authority cracking down on opposition?, 25 January 2022, consulted on 26 January 2022.

were documented in which Palestinian freedom of expression was censored online or accounts were temporarily removed by online (social media) platforms.<sup>296</sup>

#### Demonstrations in Gaza

According to one source, there is said to be zero tolerance for most demonstrations in Gaza. People who demonstrate are at risk of violence, intimidation and detention by the authorities. Protests involving subjects that are sensitive for Hamas or the humanitarian situation in Gaza, such as access to electricity, are not permitted. The source says that certain topics, such as demonstrations against Israel or gender-based violence, are tolerated because they do not specifically target the authorities. Although it is not a legal requirement, protesters that are planning a demonstration must apply to Hamas for approval. According to this source, it is mainly political parties that obtain approval for demonstrations. <sup>297</sup>

#### Demonstrations in the West Bank

Permits from the PA are required to hold demonstrations in the West Bank. Demonstrations criticising the policy of the PA have been broken up by PA security forces. In 2020, it was prohibited to hold meetings because of the coronavirus pandemic. The Palestinian organisation Al Haq registered nine peaceful gatherings in 2020 that were attacked or banned in the Palestinian Territories. For example, in September 2020 PA security services attacked protesters in Ramallah with tear gas and stun grenades. <sup>298</sup>

According to Israeli military order number 101 dating from 1967, protesters in the West Bank require a permit if the demonstration involves more than ten people. In addition, there is an Israeli military order, number 1615 from 2009, that states that persons who disturb public order or cause incitement must be punished. The Israeli forces use this legislation to prosecute protesters and they use tear gas, rubber bullets or live ammunition to break up demonstrations. <sup>299</sup>

The following examples from 2021 and 2022 are illustrative but by no means exhaustive.

In September 2021, one hundred Palestinian protesters in the West Bank were injured in clashes with Israeli soldiers. The protesters gathered in solidarity with six Palestinians who had earlier escaped from an Israeli prison. Palestinian demonstrators near Ramallah fired at Israeli soldiers but nobody was injured. Israeli troops fired on the crowd and also used tear gas and rubber bullets. Demonstrations were held in Ramallah, Nablus, Bethlehem and Hebron, among others.<sup>300</sup>

Journalists who report on demonstrations run the risk of excessive use of force by the Israeli police. For example, during demonstrations in the Sheikh Jarrah neighbourhood in East Jerusalem on 17 December 2021, Associated Press (AP) photojournalist Mahmoud Illean was beaten by Israeli police and stun grenades were fired at him. Other journalists are also said to have been attacked during these

<sup>&</sup>lt;sup>296</sup> 7amleh, The Arab Center for the Advancement of Social Media, *The Attacks on Palestinian Digital Rights, Progress report, May* 6-19, 2021, p. 2-3.

<sup>&</sup>lt;sup>297</sup> Confidential source, 3 February 2022.

<sup>&</sup>lt;sup>298</sup> Al Haq, Field Report on Human Rights Violations in 2020, 2 March 2021, consulted on 1 March 2022; Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021.

<sup>&</sup>lt;sup>299</sup> Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021.
<sup>300</sup> Haaretz, Shots fired at Israeli forces during West Bank protest in support of escaped prisoners, 8 September 2021, consulted on 12 April 2022; Al Jazeera, Israel attacks solidarity protests for escaped Palestinians, 9 September 2021, consulted on 13 January 2022.

demonstrations.  $^{301}$  Israeli activists who wanted to demonstrate against the Israeli president's visit to the Ibrahimi Mosque in Hebron (West Bank) in November 2021 were stopped from entering the area by the Israeli army.  $^{302}$ 

Secondary school students demonstrating in Tubas (West Bank) in early February 2022 were shot at by an undercover PA agent. Two students were injured. The agent was suspended. An internal investigation into the agent is said to be ongoing. <sup>303</sup>

Violence between Palestinians and Israeli forces in the West Bank flared up again in February 2022 – among other things during several demonstrations against Israeli settlements and in protest against the killing of five Palestinians that same month by Israeli forces (see also 3.3.6). During the protests, 442 Palestinians were injured in the confrontation with Israeli troops.<sup>304</sup>

### Online monitoring

PA security forces monitor online activity and social media posts and arrest individuals who incite or criticise the PA. The 2017 Electronic Crimes Law (ECL) of the PA, also known as the Cybercrime Law, punishes a number of online offences with fines and jail terms, including the publishing of material that is critical of the PA, that disrupts public or national order or is contrary to family and religious values. In 2020, several individuals were arrested and questioned by the PA security service for material they posted on social media, including planning a demonstration that was critical of how the PA was handling the coronavirus pandemic.<sup>305</sup>

According to one source, Hamas also monitors mobile phones, but there is no evidence of this. Although internet providers are independent, a source says it is not impossible that the Palestinian telecommunications company Paltel is maintaining contact with the PA, which means that the PA may have access to customer data. It is not known whether private users or providers are obliged to place an internet filter. In criminal cases, private providers are required to provide customer phone records and messages to the PA. According to the source, the use of internet cafés is outdated as most people use the internet at home and on their mobile phones. Social media are also monitored by Hamas and the PA. Political opponents and demonstrators therefore run the risk of being questioned by the security services. They can be ordered to divulge the passwords of their mobile phones and their social media accounts. In such cases, the authorities can confiscate telephones for days in order to examine the contents. The source says that when entering and leaving the Palestinian Territories, people are not asked to provide passwords from their phones or social media by default. Persons who post certain information on the internet from abroad may be at risk of interrogation, intimidation or smear campaigns by the Palestinian authorities. According to the source, this differs per case and it is not clear which cases will be prosecuted. To obtain a SIM card, mobile phone users must register by name with telephone companies.<sup>306</sup>

<sup>&</sup>lt;sup>301</sup> The Independent, *AP photographer beaten by Israeli police in east Jerusalem*, 17 December 2021, consulted on 20 December 2021.

<sup>302</sup> Al Jazeera, Hebron: Israeli president visit overlooks violent settler control, 1 December 2021, consulted on 2 December 2021.

<sup>303</sup> Al Jazeera, Palestinians demand answers for PA shooting at student protest, 3 February 2022, consulted on 9 February 2022.

<sup>&</sup>lt;sup>304</sup> OCHA OPt, *Protection of Civilians Report 8-21 February 2022*, 28 February 2022, consulted on 1 March 2022.

<sup>305</sup> Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021.

<sup>&</sup>lt;sup>306</sup> Confidential source, 3 February 2022.

### Monitoring by the Israeli authorities

The Israeli authorities are said to digitally track Palestinians and collect and monitor personal data. According to sources, this is done using methods such as security cameras and facial recognition technology, spyware on mobile phones and the monitoring of Palestinian social media and telephone traffic and conversations in the West Bank and Gaza. <sup>307</sup> For example, in May 2021, Palestinian visitors to the Al Aqsa mosque received text messages signed by 'Israeli intelligence' saying that they had engaged in 'violent work' and that they would be held responsible for it. In addition, at the start of the coronavirus pandemic in April 2020 the Israeli authorities are said to have drafted new regulations that permit mobile phones to be monitored. For example, Palestinians were asked to download the *Coordinator* or *Monasiq* phone app in order to receive permits to travel to work. <sup>308</sup> According to one source, the use of this app is questionable. The source says that this could lead to self-censorship and a curtailment of freedom because Palestinians are afraid of being monitored, particularly in their own homes. <sup>309</sup>

A number of human rights organisations claim that the company NSO Pegasus has used spyware to hack and monitor the mobile phones of six Palestinian (human rights) organisations. It is not clear why and for whom this information was monitored.  $^{310}$ 

### 3.1.2 Freedom of association and assembly

According to one source, organisations can operate openly in the Palestinian Territories provided they are not critical of Hamas or the PA. Organisations – including those from Gaza – must have their funding approved by the PA. According to one source, these funding applications take a long time and must be approved in the bureaucratic system by various ministries, security services, the cabinet and the office of the prime minister. The authorities are said to influence the work of organisations in this way. <sup>311</sup> Activists who are critical of the PA leadership are at risk of intimidation and ill-treatment by the security forces. In addition, the activities of Islamic groups are regularly shut down by the PA or Israeli troops. <sup>312</sup>

Although there are a wide range of NGOs active in the West Bank, their freedom of movement is restricted by Israeli forces, according to Freedom House. Under Israeli legislation passed in 2017, foreigners who openly support a boycott of Israel or of settlements in the West Bank can be barred from entering the country. Among other things, this law was used to prevent two US congressmen from entering Israel and the West Bank in August 2019. The same law was applied in November 2019 to order American Omar Shakir, the Human Rights Watch director for Israel and the Palestinian Territories, to leave the country.<sup>313</sup> In July 2020, the Israeli authorities

<sup>&</sup>lt;sup>307</sup> Al Shabaka: The Palestine Policy Network, Rethinking Palestine Podcast, Palestine under Surveillance with Mona Shtaya, 25 November 2021, consulted on 25 November 2021; The Washington Post, Israel escalates surveillance of Palestinians with facial recognition program in West Bank, 8 November 2021, consulted on 1 March 2022.

<sup>308</sup> Middle East Eye, 'The Coordinator': Israel instructs Palestinians to download app that tracks their phones, 8 April 2020, consulted on 12 March 2022; Middle East Eye, Israel: Lawyers call on Shit Bet to halt text message threats targeting Palestinians, 13 May 2021, consulted on 23 February 2022; The Independent, Israel upholds use of surveillance technology on protesters, 2 February 2022, consulted on 3 February 2022.

<sup>&</sup>lt;sup>309</sup> Al Shabaka: The Palestine Policy Network, Rethinking Palestine Podcast, Palestine under Surveillance with Mona Shtaya, 25 November 2021, consulted on 25 November 2021.

<sup>310</sup> Human Rights Watch, Spyware used to hack Palestinian Rights Defenders, 8 November 2021, consulted on 25 November 2021; AP News, Report: NSO spyware found on 6 Palestinians activists' phones, 8 November 2021, consulted on 1 March 2022.

<sup>311</sup> Confidential source, 3 February 2022.

<sup>&</sup>lt;sup>312</sup> Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021.

<sup>313</sup> Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021.

are said to have raided the offices of three arts and culture organisations (Edward Said National Conservatory of Music, Yabous Cultural Center and the Shafaq Cultural Network) in East Jerusalem. Documents were seized and the directors of the organisations were detained.<sup>314</sup> The Israeli authorities said that it involved an investigation into financial offences, including money laundering and tax evasion.<sup>315</sup>

#### Trades unions

Trades unions can be formed without government supervision, but it is said that labour legislation is not properly applied in practice. To organise a strike, an application must be made to the Palestinian Ministry of Labour Affairs. This process is made difficult by certain regulations. Palestinian workers in Jerusalem are subject to Israeli law. In October 2020, teachers in the West Bank were arrested after taking part in a strike over unpaid salaries. The PA is facing financial difficulties due to economic stagnation, inflation and dwindling foreign aid and was unable to pay full government salaries at certain times during the reporting period. One source says that a number of trade unions are active in Gaza but cannot play an active role in civil society due to the influence of Hamas organisations.

## 3.1.3 Freedom of religion and belief

According to a survey by the organisation PAX, the majority of respondents in the West Bank say they enjoy a high level of religious freedom. In addition, 73% of those surveyed say they view religion as an individual choice and respect other beliefs. However, 18% of the respondents claimed that they had experienced religious discrimination. The PA and private citizens are said to interfere in the exercise of religion. For example, it is said that the authorities persecute people who do not fast during Ramadan and religious leaders stigmatise people who do not comply with their religious precepts. It was not clear which type of persecution this involved.<sup>319</sup>

According to Freedom House, due to the security situation Jewish, Muslim and Christian communities in the West Bank face varying degrees of restrictions when visiting places of worship. For example, Palestinian Muslims are regularly prevented from entering Jerusalem to pray and young Muslim men are not permitted to enter the Temple Mount on Fridays (see explanation in footnote). 320 According to one source, Gazans (both Christians and Muslims) also face restrictions to their movements when practising their faith and visiting religious sites in Jerusalem,

<sup>&</sup>lt;sup>314</sup> Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021.

<sup>315</sup> Palestinian Centre for Human Rights, IOF Raid Yabous Cultural Centre and Edward Said National Conservatory of Music in Occupied East Jerusalem, 23 July 202, consulted on 12 April 2022.

Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021.
 Al Jazeera, Palestinian Authority struggles to pay public employees, 2 December 2021, consulted on 3 December 2021; Reuters, Palestinian Authority cut wages in prisoner, tax dispute with Israel, 6 December 2021, consulted on 7 March 2022.

 <sup>318</sup> Confidential source, 3 February 2022.
 319 The PAX survey included 850 respondents of various ages and political and religious affiliations. PAX, Executive Summary Discrimination Combating Mechanisms: Freedom of Religion and Belief in Palestine, 2020, p. 4.

The religious buildings on the Temple Mount in Jerusalem are administered by the Jordanian *Waqf* (religious foundation). The Jordanian *Ministry of Islamic Affairs and Holy Places* supports the maintenance and salaries of Waqf employees in Jerusalem. The Israeli police are responsible for the security of the Temple Mount. In January and February 2020, Palestinian protests on the Temple Mount were broken up by Israeli forces. Since November 2020, Israeli authorities are said to check visitors to the old city (Temple Mount) for identity cards. People with identity cards from the West Bank are said to be refused entry. In addition, prominent Muslims are said to be sometimes barred from entry by the Israeli authorities and some Muslims are refused entry on Islamic holidays on the basis of their age and/or gender. Gazans have not been allowed to enter the Temple Mount since 2017. Muslims who are Israeli citizens and Palestinian residents of Jerusalem do not require a permit to visit the Temple Mount. USDoS, *Israel* (*includes West Bank and Gaza*) 2020 International Religious Freedom Report, 2021, p. 15-17, 20; Freedom House, *Westbank: Freedom in the World 2021*, 2021, consulted on 29 November 2021.

Bethlehem and Hebron.<sup>321</sup> In addition, Jews and Muslims are kept apart when entering the Ibrahimi Mosque/*Tomb of the Patriarchs* (part mosque, part synagogue) in Hebron, located in the West Bank. This was first instituted after an attack in 1994 in which 29 Muslims were killed.<sup>322</sup> Palestinians must go through a military checkpoint in order to enter the mosque. Palestinians are also not allowed to enter the mosque during the night or on the Jewish Sabbath and Jewish holidays.<sup>323</sup>

A USDoS report states that access to many religious sites is being restricted by the Israeli authorities. <sup>324</sup> During the coronavirus pandemic, places of worship were partially closed and gatherings were restricted in the Palestinian Territories. In late November 2021, it was announced that the Israeli authorities would grant permits to five hundred Christians from Gaza to travel to family and holy sites in Israel and the West Bank during the Christmas season. The freedom of movement to and from Gaza is very limited (see also 1.4). <sup>325</sup> It is not clear whether these permits were actually granted. In April 2022, it was announced that Israel would allow 722 Christians from Gaza, consisting of entire families, access to Jerusalem for Easter. In 2019, only three hundred Christians over the age of 55 received a visa. <sup>326</sup>

#### 3.1.4 Evictions

Palestinians in East Jerusalem and the West Bank are at risk of eviction. UNRWA has stated that the demolition of homes, infrastructure and livelihoods in East Jerusalem and Area C by the Israeli authorities is destroying Palestinian families and communities. Between 28 and 46 percent of Palestinian homes in East Jerusalem and Area C are at risk of demolition. In East Jerusalem and in Area C of the West Bank, it is said to be difficult to obtain an Israeli building permit. Between 2009 and 2018, only two percent of applications (98 out of 4,422) were granted. Palestinians who build in this area without a permit are at risk of demolition and eviction. According to one source, people who make minor repairs to their homes are also at risk of being accused of not having a building permit. Israeli settlers can, however, build in these areas and can obtain permits from the Israeli authorities.

The *Norwegian Refugee Council* (NRC) reported that 218 Palestinian households were facing eviction court cases in November 2021.<sup>332</sup> Human Rights Watch has stated that at least 46 Palestinian communities in the West Bank and more than one hundred thousand Palestinians in East Jerusalem are at risk of eviction. The organisation *Save the Children* says that more than fifty childcare locations and primary schools used by more than 5,000 Palestinian children are at risk of

<sup>321</sup> Confidential source, 3 February 2022.

<sup>322</sup> BBC Monitoring, Israeli president's visit to West Bank shrine draws protest, 29 November 2021, consulted on 30 November 2021.

<sup>323</sup> Al Jazeera, Hebron: Israeli president visit overlooks violent settler control, 1 December 2021, consulted on 2 December 2021.

 <sup>324</sup> USDoS, Israel (includes West Bank and Gaza) 2020 International Religious Freedom Report, 2021, p. 70-71.
 325 The Independent, Israel to give Gaza Christians permits for Christmas holiday, 24 November 2021, consulted on 25 November 2021

<sup>326</sup> Nederlands Dagblad, *Meer christenen uit Gaza welkom in Jeruzalem,* 15 April 2022, consulted on 15 April 2022.

<sup>327</sup> UNRWA, Demolition Watch, consulted on 14 January 2022. See also Error! Hyperlink reference not valid.

<sup>&</sup>lt;sup>328</sup> OCHA, *Most Palestinian plans to build in Area C not approved*, 22 June 2021, consulted on 1 March 2022.

<sup>329</sup> UNRWA, Demolition Watch, consulted on 14 January 2022. See also Error! Hyperlink reference not valid.

<sup>&</sup>lt;sup>330</sup> Confidential source, 2 February 2022.

<sup>331</sup> OCHA, Most Palestinian plans to build in Area C not approved, 22 June 2021, consulted on 1 March 2022.

<sup>332</sup> Norwegian Refugee Council, 10 things you should know about the evictions in East Jerusalem, 11 November 2021, consulted on 1 March 2022.

demolition.<sup>333</sup> The *Palestinian Independent Commission for Human Rights* (ICHR) has said that the Israeli authorities demolished 869 Palestinian houses and buildings in 2020. 242 of the 248 homes were demolished because it was claimed they had no building permit. This mainly concerned houses in East Jerusalem and Area C. In addition, 8,830 *dunums*<sup>334</sup> in the West Bank were seized for Israeli settlements, 11,200 dunums were set aside for nature reserves, and olive and almond trees were uprooted on 1,500 dunums of farmland.<sup>335</sup> OCHA has stated that 2,416 buildings were destroyed and more than three thousand people were displaced from 2019 to January 2022.<sup>336</sup> The demolition is set to continue in 2022. To illustrate, in February 2022, 29 Palestinian homes and other buildings were demolished, including 24 in Area C and five in East Jerusalem. 23 people were displaced as a result.<sup>337</sup>

According to one source, displaced persons end up on the streets of the West Bank and East Jerusalem. People often stay with family first or move to another location in the West Bank. In the West Bank, there is an international organisation that offers emergency aid and/or temporary accommodation in temporary structures or caravans. After that, displaced persons often leave for other locations in the West Bank, <sup>338</sup>

#### East Jerusalem

Palestinians in East Jerusalem have been living for decades with the threat of a possible eviction. The Sheikh Jarrah and Silwan neighbourhoods have attracted particular attention because of the recent court cases over Palestinian home ownership in East Jerusalem. This mainly concerns Palestinians and their descendants who were displaced from West Jerusalem in 1948 and fled to East Jerusalem. In the early 1950s, 28 Palestinian refugee families received funding from UNRWA to build houses on land in the Sheikh Jarrah neighbourhood. This land was made available by the Jordanian government, which at the time had control over East Jerusalem. The families had to give up their right to food aid and were granted ownership of the land and homes after three years of paying rent. 339

Israeli settler organisations are said to use Israeli laws that permit the reclamation of land in East Jerusalem that had been Jewish-owned before 1948. However, Palestinians are not entitled to reclaim land that was in Palestinian possession before 1948. In 2008, around three hundred people in East Jerusalem were at risk of eviction. Israel did not recognise these individuals as owners of the homes and of the land on which the homes were built. In 2022, this number was said to have increased to 970 people. At least 317 people are believed to have been displaced by evictions in East Jerusalem in 2021. According to one source, not only individual

<sup>333</sup> oPt Education Cluster, Background on schools under threat of demolition in West Bank, December 2020, p. 1; Save the Children, Hope under the rubble: The impact of Israel's home demolition policy on Palestinian children and their families, 2021, p. 4; Human Rights Watch, Israel and Palestine: Events of 2021, 2022, consulted on 14 January 2022.

 $<sup>^{\</sup>rm 334}$  A metric measure in which 1 dunum equals 1,000 m².

<sup>335</sup> ICHR, The Status of Human Rights in Palestine Executive Summary 2020, 2021, p. 16.

<sup>&</sup>lt;sup>336</sup> OCHA, Data on demolition and displacement in the West Bank, 12 January 2021, consulted on 14 January 2022.

<sup>337</sup> OCHA oPt, Protection of Civilians Report 8-21 February 2022, 28 February 2022, consulted on 1 March 2022.

<sup>338</sup> Confidential source, 2 February 2022.

 <sup>&</sup>lt;sup>339</sup> Francesca P. Albanese and Lex Takkenberg, *Palestinian refugees in international law*, 2020, p. 242- 243.
 <sup>340</sup> The Association for Civil Rights in Israel, *Sheikh Jarrah: Discrimination, Evictions and Dispossession under the auspices of the law*, June 2021, consulted on 19 January 2022; BBC News, *Sheikh Jarrah: Why could Palestinians lose their homes in Jerusalem*, 3 August 2021, consulted on 17 January 2022.

<sup>341</sup> Francesca P. Albanese and Lex Takkenberg, Palestinian refugees in international law, 2020, p. 242- 243.

<sup>342</sup> BBC News, Sheikh Jarrah: Why could Palestinians lose their homes in Jerusalem, 3 August 2021, consulted on 17 January 2022; Al Jazeera, Sheikh Jarrah: Palestinian family faces forced displacement, 5 January 2022, consulted on 6 January 2022.

houses are affected but also entire streets and neighbourhoods in certain parts of East Jerusalem – for example in Silwan, Batn Al Hawa and Sheikh Jarrah. 343

### Sheikh Jarrah

In October 2020, it was announced that a number of Palestinian families in Sheikh Jarrah would be evicted from their homes in May 2021. In February 2021, those families lodged an appeal that led to demonstrations in Jerusalem and a further escalation of violence (see also 1.2.1). In November 2021, four families turned down a deal that placed their homes under temporary ownership of Israeli settler organisations. As part of this deal, they would have been able to stay in their homes as protected tenants for fifteen years. According to one source, there have been thousands of eviction court cases pending during the reporting period but the chances of Palestinians winning them are slim. Also On 1 March 2022, Israel's Supreme Court decided to postpone the evictions of seven Palestinian families in Sheikh Jarrah. The families can remain in the homes as tenants until ownership of the land in question has been established.

In January 2022, several families in East Jerusalem were evicted and their homes were demolished. The residents often demolish their own homes because otherwise they must pay the municipality of Jerusalem for the costs of the demolition.<sup>347</sup> According to one source, relief organisations are only permitted to provide emergency relief in East Jerusalem and not sustainable aid.<sup>348</sup> According to another source, evicted people sometimes stay with relatives but most of them end up in the part of East Jerusalem that is on the other side of the wall around the West Bank or in the West Bank (see also 4.2). For example, one family was temporarily housed in a Red Cross tent after being evicted.<sup>349</sup>

Pending a further decision by the Israeli Supreme Court and further evictions, tensions in the neighbourhood remained high. There were several violent incidents such as stabbings and violence between Palestinian residents and settlers and attacks on demonstrators and journalists, including by Israeli police. There were also restrictions in freedom of movement and continued demonstrations.<sup>350</sup>

## 3.1.5 Recruitment

It is unclear whether militias are forcibly recruiting or enlisting Palestinians. According to one source, there is no forced recruitment for the PA security services. There was no known forced recruitment by militias.<sup>351</sup> Defence for Children

<sup>343</sup> Confidential source, 2 February 2022.

<sup>&</sup>lt;sup>344</sup> Al Jazeera, *Sheikh Jarrah families reject 'unjust' deal with Israeli settlers*, 2 November 2021, consulted on 3 November 2021.

<sup>345</sup> Confidential source, 2 February 2022.

<sup>&</sup>lt;sup>346</sup> OCHA oPt, *Protection of Civilians Report* | 22 February – 7 March 2022, 11 March 2022, consulted on 12 March 2022.

<sup>&</sup>lt;sup>347</sup> Confidential source, 2 February 2022.

 <sup>348</sup> Confidential source, 2 February 2022.
 349 Al Jazeera, Jerusalem: Israel forces Palestinians to self-demolish own homes, 31 January 2022, consulted on 1 February 2022.

BBC Monitoring, Palestinian girl held in stabbing of settler in Jerusalem, 8 December 2021, consulted on 9
December 2021; BBC Monitoring, Two Israeli teens arrested over stoning attack in Sheikh Jarrah, 11 December 2021, consulted on 12 December 2021; OCHA, Protection of Civilians Report | 16 November – 6 December 2021, 14 December 2021, consulted on 15 December 2021; The Independent, AP photographer beaten by Israeli police in east Jerusalem, 17 December 2021, consulted on 18 December 2021; OCHA, Protection of Civilians Report | 7-20 December 2021, 27 December 2021, consulted on 3 January 2022; Al Jazeera, Sheikh Jarrah: Palestinian family faces forced displacement, 5 January 2022, consulted on 6 January 2022; The Guardian, Israeli police in standoff with Palestinians over Sheikh Jarrah eviction, 17 January 2022, consulted on 17 January 2022.

<sup>351</sup> Confidential source, 7 February 2022.

International Palestine (DCIP)<sup>352</sup> has stated that Palestinian children are vulnerable to recruitment by Palestinian armed groups and are being used as human shields. Between 2011 and 2020, DCIP recorded nine deaths and two injuries of minors who were helping Palestinian armed groups or were accidentally at the scene. Among other things, they were the victims of 'active hostilities' with Israel, accidentally fired explosives and 'friendly fire'.<sup>353</sup>

## 3.2 Position of specific groups

#### 3.2.1 Journalists, human rights advocates and political activists

The Palestinian Territories are ranked 132 out of 180 countries surveyed in the World Press Freedom Index of Reporters Without Borders (RSF), where countries high on the list were found to impose restrictions on freedom of expression. 354 Between May 2018 and early March 2022, two Palestinian journalists were said to have been killed by Israeli troops and dozens of journalists injured during protests in Gaza. Journalists are said to apply self-censorship to avoid problems with the Israeli authorities. According to RSF, PA security services were also guilty of targeted violence against journalists during protests that were critical of the PA. Several websites critical of the PA are said to have been blocked by the PA since 2017. In addition, the profiles or content of journalists on social media platforms such as Facebook and Twitter were said to have been removed under Israeli political pressure. Local NGOs accuse these platforms of applying a double standard as far as Israeli and Palestinian content is concerned. 355 For example, Facebook shut down the profiles and pages of Palestinian journalists who were under suspicion of inciting violence and terrorism. 356

According to Freedom House and RSF, reporters were at risk of monitoring, attacks and detention by Israeli forces. In 2020, the Palestinian Centre for Development and Media (MADA) registered 210 cases in which press freedom was being compromised by Israeli forces in the Palestinian Territories. This included physical attacks, preventing journalists from reporting on certain events, interrogation, arrest and detention.<sup>357</sup> In late May 2021, thirteen Palestinian journalists were allegedly detained by the Israeli authorities.<sup>358</sup>

Human rights advocates and political activists were restricted in their actions by the Israeli authorities and the PA.<sup>359</sup> In June 2021, following the death of Nizar Banat, a human rights activist who had criticised the PA, a large number of protests were crushed by the *Palestinian Civil Police* (PCP). Banat died of his injuries after being

<sup>352</sup> DCIP is the Palestinian branch of Defence for Children International, an NGO that champions children's rights worldwide. DCIP works for children who become involved with the Israeli military and civilian justice systems as well as the Palestinian juvenile justice system. On 29 July 2021, the DCIP office in Al-Bireh (West Bank) was raided by Israeli paramilitary forces. The organisation's computers and hard drives were confiscated. DCIP, Israeli forces raid DCIP office, confiscate computers and client files, 29 July 2021, consulted on 13 February 2022.

 <sup>353</sup> Defense for Children International Palestine, Child Recruitment, 2022, consulted on 12 January 2022
 354 Reporters Without Borders, Palestine: Harassed journalists, 2021, consulted on 14 December 2021.

<sup>355</sup> Reporters Without Borders, Palestine: Harassed journalists, 2021, consulted on 14 December 2021; Reporters Without Borders, At least 35 journalists attacked while covering West Bank protests, 5 July 2021, consulted on 14 December 2021; Reporters Without Borders, WhatsApp blocks accounts of at least seven Gaza Strip journalists, 7 July 2021, consulted on 14 December 2021; 7amleh, The Attacks on Palestinian Digital Rights: Progress report, may 6-19, 2021, 21 May 2021.

<sup>&</sup>lt;sup>356</sup> Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021.

Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021.
 Reporters Without Borders, Israel now holding 13 Palestinian journalists, 28 May 2021, consulted on 14 December 2021.

<sup>359</sup> Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021.

severely beaten by PA security forces. <sup>360</sup> Journalists, civil society activists and lawyers were unlawfully arrested and tortured by PA security forces at the time. According to think-tank Al Shabaka, the level of repression in the summer of 2021 was unparalleled. <sup>361</sup> A report by Human Rights Watch stated that from January to September 2021, the Palestinian Independent Commission for Human Rights (ICHR) received 87 complaints of wrongful arrest, fifteen complaints of detention without trial or charge, and 76 complaints of ill-treatment and torture by the PA. <sup>362</sup> The ICHR stated in its 2020 report that it had received hundreds of complaints from both Gaza and the West Bank accusing police and security forces of ill-treatment and torture (see also 3.3.4). <sup>363</sup>

Opponents who criticise specific Hamas members are at risk of arrest and detention. It was said that Hamas also tried to maintain control over other groups in Gaza that it believed might damage its position of power. According to one source, Palestinians cannot oppose Hamas's control.<sup>364</sup> During demonstrations in Gaza in 2019, security forces beat protesters and security forces fired shots to disperse the crowd.<sup>365</sup>

### 3.2.2 Collaborators

According to the same source, (alleged) collaborators/informants of the PA or Israel are arrested, detained and mistreated by Hamas.<sup>366</sup>

3.2.3 Groups receiving the special attention of the Israeli authorities

It is not possible to give an unequivocal answer to the question of which groups receive the special attention of the Israeli authorities.

## 3.2.4 Conscientious objectors and deserters

According to the Oslo Accords, besides the Israeli Defence Forces (IDF) and the Palestinian Police no other armed groups are permitted to operate in the West Bank and Gaza. 367 One source says that there is no conscription in the Palestinian Territories. Members of the PA Security Forces (PASF) voluntarily choose to join. All members, including the civilian police, are subject to Palestinian military criminal law and the military court. 368 The Palestinian Police are part of the Security Forces (PASF), which are armed and all come under the Ministry of Interior. 369 The training for the security services is held at the Istiqlal University in Jericho. 370

There is therefore also no conscription for women. Women who work for the government in departments related to the security services are said to perform only administrative tasks.<sup>371</sup> According to another source, there are female officers, but it

<sup>360</sup> The New York Times, Activist Dies in Palestinian Authority's Custody, Inciting Widespread Outrage, 24 June 2021, consulted on 25 June 2021.

<sup>&</sup>lt;sup>361</sup> Al Shabaka, The Palestinian Security Sector: Entrenching State Repression, 14 November 2021, consulted on 15 November 2021.

<sup>362</sup> Human Rights Watch, Israel en Palestine: Events of 2021, 2022, consulted on 8 April 2022. See also
363 ICHR. The status of human rights in Palestine Everytive summary, 2020, 2021, p. 27-28. **Front Hyperli** 

<sup>363</sup> ICHR, The status of human rights in Palestine Executive summary 2020, 2021, p. 27-28. Error! Hyperlink reference not valid.

 $<sup>^{\</sup>rm 364}$  Confidential source, 3 February 2022.

<sup>365</sup> BBC News, *Palestinians vent against Hamas in rare online event*, 29 January 2022, consulted on 30 January 2022.

<sup>&</sup>lt;sup>366</sup> Confidential source, 3 February 2022.

<sup>&</sup>lt;sup>367</sup> Artikel XIV, Israeli-Palestinian Interim Agreement on the West Bank and the Gaza Strip, 28 September 1995, p. 11-12. See also Error! Hyperlink reference not valid.

<sup>&</sup>lt;sup>368</sup> Confidential source, 7 February 2022.

<sup>369</sup> Confidential source, 11 April 2022.

<sup>&</sup>lt;sup>370</sup> Confidential source, 3 February 2022.

<sup>&</sup>lt;sup>371</sup> Confidential source, 24 January 2022.

is not clear how many.<sup>372</sup> In 2007, Hamas is said to have set up a female military unit. It is not clear which tasks this unit performs.<sup>373</sup>

## 3.2.5 Ethnic and religious groups

The population in the Palestinian Territories is mostly made up of Palestinians. A large number of Jewish settlers also live in the West Bank. In 2019, there were 432,000 Israeli settlers in the West Bank and 227,100 in East Jerusalem.<sup>374</sup> One source says that both Muslims and Christians from the Palestinian Territories see themselves as ethnically Arab.<sup>375</sup> According to the CIA World Factbook, Gaza is made up of Palestinian Arabs of which 98-99% are Muslim (mainly Sunni), 1% Christian and less than 1% have an unknown religion or denomination. The population of the West Bank (excluding East Jerusalem) is made up of Palestinian Arabs, 80-85% of whom are Muslim (mainly Sunni), 1-2.5% Christian (mainly Greek Orthodox, but also Armenian and Syrian Orthodox, Lutheran and Catholic), a Jewish population of 12-14% and a 'remainder' category.<sup>376</sup> There is also the Samaritan community, which is a group of almost eight hundred people who see themselves as followers of the pure Jewish faith. They live in Nablus (West Bank) and Holon (Israel).<sup>377</sup>

The Christian population is said to be decreasing in size due to restrictions on internal freedom of movement and on residence permits for (Christian) Palestinian spouses from other parts of the Palestinian Territories due to the Israeli Citizenship Law (see also 2.3).<sup>378</sup> It is not known whether Christians can obtain permits more often or more easily. According to one source, Christians see themselves as an oppressed group that is not well represented in Palestinian politics. They are said to be often inclined to leave the Palestinian Territories. Among Muslims, the general view is that Christians are a privileged group who have easier access to travel permits.<sup>379</sup>

According to a USDoS report, unrecognised religious groups such as Jehovah's Witnesses are banned from proselytising activities by the PA. Some personal status documents, such as marriage and birth certificates issued by unrecognised Christian churches, were not accepted by the PA. Individuals were advised by church communities to register either with a church community abroad or with a recognised church of a different denomination. According to one source, church communities are very small and closed and do not readily accept converts. It is not clear whether recognised non-Muslim groups are permitted by the PA to engage in proselytising activities.

Christians born in Gaza are tolerated and are not forced to abide by Islamic law. Cases of discrimination or harassment against the Christian minority in Gaza were

<sup>&</sup>lt;sup>372</sup> Confidential source, 11 April 2022.

<sup>&</sup>lt;sup>373</sup> Al Monitor, *Hamas appoints first women to political bureau*, 12 March 2021, consulted on 2 February 2022.

<sup>&</sup>lt;sup>374</sup> CIA The World Factbook, West Bank, Gaza Strip, 16 November 2021, consulted on 29 November 2021.

<sup>&</sup>lt;sup>375</sup> Confidential source, 3 February 2022.

<sup>&</sup>lt;sup>376</sup> CIA The World Factbook, West Bank, Gaza Strip, 16 November 2021, consulted on 29 November 2021; Confidential source, 10 February 2022.

<sup>377</sup> EO, *Van Nablus naar Ninevé: Aflevering 1 – De Samaritanen (Westelijke Jordaanoever*), 2 July 2018, consulted on 29 November 2021.

<sup>&</sup>lt;sup>378</sup> USDoS, Israel (includes West Bank and Gaza) 2020 International Religious Freedom Report, 2021, p. 85-86.

<sup>&</sup>lt;sup>379</sup> Confidential source, 10 February 2022.

<sup>&</sup>lt;sup>380</sup> USDoS, Israel (includes West Bank and Gaza) 2020 International Religious Freedom Report, 2021, p. 75-76.

<sup>&</sup>lt;sup>381</sup> Confidential source, 10 February 2022.

not investigated by the local authorities. Christmas is regarded as a public holiday in the West Bank but not in Gaza. 382

Palestinian Christians and Muslims who want to marry a Jewish partner run the risk of opposition from society and family. Women and girls from Christian and Muslim communities who marry outside their faith risk being ostracised by their families.<sup>383</sup>

Conversion, non-practising believers, apostates and overt atheism Muslims who convert, people who have turned away from the faith and avowed atheists are unable to openly express their beliefs and are at risk of exclusion, discrimination and violence by family members, society and the authorities.<sup>384</sup> It is not clear whether this also applies to converts of other faiths.

Under Islamic law, Muslims are not permitted to convert to another faith. According to some sources, it is possible to convert to Islam but it would be problematic for a Muslim to convert to another faith. The sources are not aware of any Muslims who have converted to another faith.<sup>385</sup> The size of this group is not clear because the topic is not openly discussed.

Non-practising Muslims do not generally experience problems. According to one source, they only have problems if they talk about this openly or are openly critical of religion. Non-practising Muslims can face problems around Ramadan if they do not participate in public religious rituals. In the larger cities, it is said to be generally easier to be non-practising. 386

According to one source, apostates are considered to be people who criticise religion or speak about it in a defamatory way.<sup>387</sup> Apostasy or criticising the existence of religion are not openly discussed in Gaza.<sup>388</sup>

Atheism is said to be tolerated in the West Bank, but not when it is displayed explicitly. This includes the expression of atheism online. According to one source, in Gaza people can manifest themselves as atheists in certain ways on social media. The source refers to the existence of Marxist political parties that are indirectly atheistic. People who openly present themselves as atheists are said to be at risk of intimidation by their surroundings. In the source refers to the existence of Marxist political parties that are indirectly atheistic.

Blasphemy is punishable in the West Bank and people who violate this law can be prosecuted by the PA. Under Article 278 of the 1964 Jordanian Criminal Code, blasphemy is punishable by life imprisonment and under Article 330 of the 1979 PLO Criminal Code it is punishable by a fine and a prison sentence of up to three months. <sup>391</sup> In Gaza, Article 149 of the British penal code stipulates that persons who insult religion can be sentenced to a maximum of one year in prison. <sup>392</sup> Two sources said that they were not aware of any people being criminally prosecuted in the Palestinian Territories because of their religious beliefs during the reporting

USDOS, Israel (includes West Bank and Gaza) 2020 International Religious Freedom Report, 2021, p. 88.
 USDOS, Israel (includes West Bank and Gaza) 2020 International Religious Freedom Report, 2021, p. 92.
 Confidential source, 3 February 2022; Confidential source, 10 February 2022.
 Confidential source, 3 February 2022; Confidential source, 10 February 2022.

<sup>&</sup>lt;sup>386</sup> Confidential source, 10 February 2022.

<sup>387</sup> Confidential source, 10 February 2022.

<sup>388</sup> Confidential source, 3 February 2022.

<sup>&</sup>lt;sup>389</sup> Confidential source, 10 February 2022.

<sup>&</sup>lt;sup>390</sup> Confidential source, 3 February 2022.

<sup>&</sup>lt;sup>391</sup> Human Rights Watch, West Bank: Free Suspected Blogger, 5 December 2010, consulted on 6 December 2021; Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021; EASO, COI Query Palestine, 10 January 2020, p. 4.

<sup>&</sup>lt;sup>392</sup> EASO, COI Query Palestine, 10 January 2020, p. 4.

period.<sup>393</sup> One such case occurred before the reporting period. This example is illustrative but can by no means be regarded as exhaustive. In October 2010, Waleed Hasayin (pen name Al-Husseini) was arrested by the PA in the West Bank for blasphemy against Islam on his blog and on Facebook. He described himself as an atheist.<sup>394</sup>

One source said that people who are openly critical of religion are more likely to be viewed as political critics. Persons who express themselves critically about religion could also be prosecuted under the Cybercrime Act because posting online material can contravene digital security (see also 3.1.1). People who get into trouble with family members or other citizens because of their beliefs cannot count on the protection of the authorities, according to the source. Having strong contacts with the PA could make a difference, but even then it would be difficult. The source says that during the reporting period, the organisers of a planned meeting on freedom of religion or life principles in Hebron were harassed with online smear campaigns. The response of the authorities was that it was their own fault.<sup>395</sup>

### 3.2.6 LGBTI people

In the West Bank (including East Jerusalem), homosexual acts are not legally punishable under the Jordanian Criminal Code of 1951. In Gaza, homosexual acts are punishable under Article 152(2) of the 1936 Criminal Code that was in force during the British Mandate. According to this article, homosexual acts are punishable by a prison sentence of up to ten years. <sup>396</sup> According to the new draft version of the Palestinian Unified Penal Code of Palestine, homosexual acts constitute a criminal offence. There are no specific laws protecting LGBTI people from hate crime, gender-based violence or discrimination. <sup>397</sup> One source says that he only knows Palestinians who are openly LGBTI people abroad. The subject is said to be taboo in the Palestinian community and only takes place in secret. This makes it difficult to obtain a complete overview of the LGBTI community in the Palestinian Territories. <sup>398</sup>

## Discrimination and criminal prosecution

According to several sources, the freedom of LGBTI people is curtailed. They are at risk of (honour and/or gender-based) violence and discrimination, including from family members and the PA. According to one source, this is worse in Gaza than in the West Bank.<sup>399</sup> A news report in the Israeli newspaper *Haaretz* states that it is generally easier for the LGBTI community in Ramallah than in the rest of the Palestinian Territories because LGBTI people can visit certain areas and bars there.<sup>400</sup> It is not clear to which extent this can be done openly. The source is not aware of actual criminal charges against LGBTI people in the West Bank but says that persons who openly identify as LGBTI face the risk of violence, discrimination and exclusion by family members and the community. They would not or could not receive protection from the authorities. People who come out as LGBTI are said to

<sup>&</sup>lt;sup>393</sup> Confidential source, 3 February 2022; Confidential source, 10 February 2022.

<sup>&</sup>lt;sup>394</sup> Human Rights Watch, *West Bank: Free Suspected Blogger*, 5 December 2010, consulted on 6 December 2021; Freedom House, *West Bank: Freedom in the World 2021 Country Report*, consulted on 2 November 2021.

 <sup>395</sup> Confidential source, 10 February 2022.
 396 Refugee Legal Aid Information, Occupied Palestinian Territory LGBTI Resources, consulted on 27 July 2021.

<sup>&</sup>lt;sup>397</sup> UNDP, *State of Palestine: Gender Justice & the Law*, 2018, p. 23. <sup>398</sup> Confidential source, 10 February 2022.

<sup>399</sup> UNIDP, State of Palestine: Gender Justice & the Law, 2018, p. 23; Amnesty International, Palestine (State of) 2020, consulted on 27 July 2021; Confidential source, 10 February 2022.

<sup>400</sup> Haaretz, Pride and Prejudice: The Hellish Life of Gaza's LGBTQ Community, 25 June 2019, consulted on 17 December 2021.

voluntarily (or involuntarily) leave the Palestinian Territories. 401 Another source says that there are cases of LGBTI people being prosecuted under legislation that addresses behaviour in public places. LGBTI people are also threatened with disclosure of their identity. This is done to prevent them from becoming activists and taking part in demonstrations, for example. It is not clear who has made these threats.<sup>402</sup> In Gaza, LGBTI people are said to be at risk of intimidation, torture or worse offences by Hamas or their own relatives. A victim belonging to the LGBTI group says in a news report from Haaretz that Hamas security forces visited him because of rumours about his sexual orientation. He was interrogated about this and tortured while in detention. He was released after he had signed a document listing the names of people with whom he had allegedly had sexual relations. The victim says that the specific story and the names in the document did not reflect the truth and believes that it may have been a tactic of Hamas to persecute Fatah members in Gaza in this way. 403

### Forced marriage

According to a news report, there are cases of Palestinian lesbian women being forced by their families to get married, mainly in rural areas. LGBTI people are being pressured by their families to enter into heterosexual marriages. The same applies to homosexual men, who are also put under extreme pressure by their families and communities to marry women. 404 One source says that (forced) marriages are taking place everywhere in the Palestinian community, and this also involves the LGBTI group.405

## Civil society organisations

There are two civil society organisations in Israel that defend the rights of Palestinian LGBTI people: the Al Qaws and Aswat organisations, which are aiming to increase social awareness of these topics. They mostly operate from Israel. According to a UNDP report, there are also organisations in Ramallah, 406 but another confidential source says that this only involves organisations that operate from abroad.407 In August 2019, Al Qaws staged a Palestinian LGBTI protest in Haifa.408 The Palestinian Authorities prevented Al Qaws from organising events in the West Bank in 2019 and threatened to arrest those in attendance. 409

Palestinian LGBTI people sometimes try to apply for asylum in Israel. However, many of them remain undocumented in Israel for fear of being deported to the Palestinian Territories. 410 In practice, they only have access to a temporary residence permit of three to four months that does not give access to healthcare, social services or employment. In order to renew a residence permit, an application for relocation/resettlement to a third country must be submitted. 411 Some

```
<sup>401</sup> Confidential source, 10 February 2022.
```

 <sup>402</sup> Confidential source, 25 March 2022.
 403 Haaretz, Pride and Prejudice: The Hellish Life of Gaza's LGBTQ Community, 25 June 2019, consulted on 17 December 2021.

<sup>404</sup> Haaretz, Pride and Prejudice: The Hellish Life of Gaza's LGBTQ Community, 25 June 2019, consulted on 17 December 2021.

<sup>&</sup>lt;sup>405</sup> Confidential source, 10 February 2022.

<sup>&</sup>lt;sup>406</sup> UNDP, State of Palestine: Gender Justice & the Law, 2018, p. 23.

<sup>407</sup> Confidential source, 10 February 2022.

<sup>&</sup>lt;sup>408</sup> The New York Times, *Palestinian Authority Bans Activities by Gay Rights Group*, 19 August 2019, consulted on 27 July 2021.

<sup>&</sup>lt;sup>409</sup> The New York Times*, Palestinian Authority Bans Activities by Gay Rights Group*, 19 August 2019, consulted on 27

<sup>410</sup> Refugee Legal Aid Information, Occupied Palestinian Territory LGBTI Resources, consulted on 27 July 2021; Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 27 July 2021.

<sup>411 +972</sup> Magazine, 'I'd rather die in the West Bank': LGBTQ Palestinians find no safety in Israel, 17 September 2021, consulted on 8 April 2022.

Palestinian LGBTI people are being given the opportunity to relocate to/resettle in a third country through UN organisations. Aguda, an Israeli LGBTI organisation that is helping Palestinian LGBTI people to do this, says that this process is very difficult and can take up to two years. This mainly concerns Palestinians from the West Bank. In the years before 2019, fewer than twenty people would have been eligible for relocation in this way. 412

## Transgender people

There is no legal recognition of transgender people in the Palestinian Territories. Gender surgery (sex reassignment) is uncommon, according to a UNDP report, but some of these operation have been performed under certain conditions. It is not clear which conditions were involved.<sup>413</sup>

#### 3.2.7 Women

The Independent Commission for Human Rights (ICHR) has said that the participation of women in social life in the Palestinian Territories is below average. The employment rate in 2020 was 15.7% among women from the age of fifteen, compared to 65.2% among men from the age of fifteen. It is true that 42.7% of the people working in the public sector in 2017 were women and 57.3% men, but only 11.3% of the senior positions in the sector were occupied by women. In addition, there was high unemployment among younger women (72.7% of women aged 20-24 compared to 31.2% of men aged 20-24 in 2020). <sup>414</sup> There is no legislation for equal pay for men and women. Another problem is said to be the exploitation of women in family businesses, where they often receive no salary and/or benefits. <sup>415</sup> According to PA legislation, women are not allowed to perform certain hazardous occupations, such as working at night or in the mining or energy sectors. <sup>416</sup> However, access to education for girls and women is said to be widespread. <sup>417</sup>

Although women have played a major role in the history of the Palestinian resistance and in civil society, numerous sources say that this has not led to the desired emancipation of women. And although Palestinian law states that men and women are equal, in practice women do not occupy the same position as men in the public and political spheres. Although the rise of the PA in which the leadership was dominated by male politicians. Although the Palestinian Ministry of Women's Affairs (MoWA) was set up in 2003 and all Palestinian government departments were given a gender unit to promote female participation and gender equality between 2012

<sup>412</sup> Haaretz, Pride and Prejudice: The Hellish Fife of Gaza's LGBTQ Community, 25 June 2019, consulted on 17 December 2021.

<sup>413</sup> UNDP, State of Palestine: Gender Justice & the Law, 2018, p. 23.

<sup>&</sup>lt;sup>414</sup> ICHR, The Report of the Independent Commission for Human Rights (ICHR) On The Initial Report of the State of Palestine Presented to The United Nations Committee on The Elimination of All Forms of Discrimination against Women, July 2018, p. 6-8; Palestinian Central Bureau of Statistics, PCBS: Labour Force Survey, Q4 – 2020, 2020, p. 14, 17.

<sup>415</sup> ICHR, The Report of the Independent Commission for Human Rights (ICHR) On The Initial Report of the State of Palestine Presented to The United Nations Committee on The Elimination of All Forms of Discrimination against Women, July 2018, p. 6-8.

<sup>416</sup> USDoS, West bank and Gaza 2020 Human Rights Report, 2021, p. 73.

<sup>&</sup>lt;sup>417</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>418</sup> Al Shabaka, The political marginalization of Palestinian women in the West Bank, consulted on 16 November 2021; Al Muntada, Parallel Follow-up Report Prepared by the Women's Center for Legal Aid and Counselling (WCLAC) and the Palestinian Non-Governmental Organizations Forum to Combat Violence Against Women, September 2020, p. 1-2; Confidential source, 26 January 2022.

 $<sup>^{419}</sup>$  Period of Palestinian resistance and escalating conflict with Israel from 1987 to 1993.

and 2014, in practice this is said to be having a minimal impact on the position of women. $^{420}$ 

The Palestinian Legislative Council has had a women's quota of 20% since 2006. According to one source, this was increased to 26% by presidential decree. In the municipal elections, there is a women's quota of 30%. 421 It is not clear whether women will actually be assigned these seats in practice. Up to December 2020, the fifteen-member PLO Executive Council had one female member (Hanan Ashrawi). There are currently only male members. 422 Of the sixteen districts in the West Bank and Gaza, one district has a female governor (Ramallah). Three positions are held by women in the current cabinet of 22 ministers: the Ministries of Health, Tourism and Women's Affairs (MoWA). In addition, women are said to hold only 11% of ambassadorial positions in the diplomatic service. In Gaza, women are said to occupy 17% of the 320 positions in the Hamas parliament. Moreover, the first female member of the Hamas political bureau, which has a total of fifteen members, was only appointed in March 2021.<sup>423</sup> The PA signed the UN resolution Convention on the Elimination of All Forms of Discrimination against Women (CEDAW) in 2014, but it has not been transposed into domestic law. In November 2017, the Supreme Constitutional Court ruled that courts cannot apply international treaties that conflict with Palestinian law. Al Shabaka says that this will allow the authorities to uphold patriarchal legislation.424

#### Political activities

The Al Shabaka think-tank says that the PA and Israeli authorities use gender-based violence to discourage women from participating in political activity. According to Al Shabaka, the Israeli military take pictures of female prisoners, unnecessarily order them to undress to be searched, and deny them access to sanitary towels and toilets during menstruation. There have also been cases of physical and sexual harassment of women by PA security forces during demonstrations and protests.<sup>425</sup> Women were attacked by PA police and undercover PA security forces during the peaceful May 2021 protests, according to one source. Amongst other things, swear words were used against the women and their phones were taken and searched for messages and photos. Arrests were made and the women were not given access to a lawyer. In addition, photos of the women were publicly distributed to show that they were not adhering to the cultural norms. 426 The security forces were said to visit the fathers of women and girls to discuss their 'activism', such as taking part in demonstrations, with the result that women and girls were prevented from participating in political activities by their families. The security forces are also said to influence the employers of women who meddle in 'activism' to force their resignation. In addition, the security forces also spread rumours about female activists on social media in order to make things more difficult for them in their

<sup>&</sup>lt;sup>420</sup> Al Shabaka, The political marginalization of Palestinian women in the West Bank, consulted on 16 November 2021; Al Muntada, Parallel Follow-up Report Prepared by the Women's Center for Legal Aid and Counselling (WCLAC) and the Palestinian Non-Governmental Organizations Forum to Combat Violence Against Women, September 2020. p. 1-2.

<sup>421</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>422</sup> ECFR, Mapping Palestinian Politics: Institutions: PA: PLC: Hanan Ashrawi (Third Way), consulted on 12 March 2022. See also **Error! Hyperlink reference not valid.** 

Al Monitor, Hamas appoints first women to political bureau, 12 March 2021, consulted on 2 February 2022.
 Al Shabaka, The political marginalization of Palestinian women in the West Bank, consulted on 16 November 2021; Al Muntada, Parallel Follow-up Report Prepared by the Women's Center for Legal Aid and Counselling (WCLAC) and the Palestinian Non-Governmental Organizations Forum to Combat Violence Against Women, September 2020, p. 1-2; Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>425</sup> Al Shabaka, *The political marginalization of Palestinian women in the West Bank*, consulted on 16 November 2021.

<sup>426</sup> Confidential source, 26 January 2022.

surroundings. $^{427}$  According to one source, the Palestinian Prime Minister has publicly apologised for the deployment of undercover PA security forces during the demonstrations in May 2021. $^{428}$ 

## Codes of conduct and dress codes: the West Bank

According to one source, Palestinian women live in a patriarchal system that mainly keeps women inside the family circle and assigns them a low (social) position and a reproductive role. The source says that it is not possible to generalise about codes of conduct and dress codes for women or the differences between married and unmarried, single or divorced women and widows. This differs in each individual case. There is said to be more freedom in the cities compared to rural areas. Despite growing conservative tendencies in society, women are said to have some freedom to choose whether or not to wear a headscarf. The decision to wear a headscarf can also differ among family members in the same family. Most women have access to education, work and public life, such as going to bars and restaurants. The employment rate for women is low, but this only affects women who work in the formal sector. Access to work or independent living is said to be accepted due to the precarious economic situation and the restrictions on freedom of movement due to the many checkpoints. According to the same source, this acceptance generally has more to do with the precarious situation in which Palestinians live than with general social ideology or belief about women's self-determination.<sup>429</sup>

#### Codes of conduct and dress: Gaza

Society in Gaza is more conservative than in the West Bank. According to one source, women in Gaza live in a patriarchal society on a daily basis and are not considered to be equal to men. It is not only religious and conservative parties that are limiting the role of women in political and social life – moderate parties are also not doing a great deal in practice to promote the participation of women. According to a United States Department of State (USDoS) human rights report, women's freedom of action in Gaza is restricted by Hamas's interpretation of Islamic norms. The security services were said to sometimes check whether couples travelling together were married. Women often have to travel in groups so that they can visit certain public spaces, such as the beach. Employers were said to discriminate against female employees. In factories, there were cases where pregnant or newly married women were not hired. Housekeeping/cleaning work is not regulated and often takes place in the informal sector. This is putting women at risk of forced labour.

According to one source, driving is normal for women and women can work in different sectors. However, the employment rate for women is low.<sup>432</sup> In 2016, women in Gaza were obliged to take a male relative (*mahram*) with them when taking driving lessons.<sup>433</sup> Women's participation in sports is limited. For example, in 2020 not all sports centres consented to a women's softball team and a bicycle project for women that was organised in a sports centre was criticised.<sup>434</sup> Women

<sup>427</sup> Al Shabaka, The political marginalization of Palestinian women in the West Bank, consulted on 16 November 2021.

 $<sup>^{\</sup>rm 428}$  Confidential source, 26 January 2022.

<sup>429</sup> Confidential source, 26 January 2022.

<sup>430</sup> Confidential source, 24 January 2022.

<sup>431</sup> USDoS, West bank and Gaza 2020 Human Rights Report, 2021, p. 61, 71, 73.

<sup>432</sup> Confidential source, 24 January 2022.

<sup>433</sup> NPR, Hamas: Gaza women learning to drive must have a chaperone, 1 June 2016, consulted on 23 December

<sup>434</sup> Al Monitor, Gaza's female softball players in league of their own, 4 March 2020, consulted on 23 December 2021; Al Monitor, Bicycling program for women finds both supporters, critics in Gaza, 10 May 2021, consulted on 23 December 2021.

are also not allowed to attend football matches. In 2018, an attempt to admit women to a football match in Nuseirat was blocked. To hold a music performance, permits must be obtained from four different agencies of the Hamas authorities. It is said to be difficult to get a permit if women are permitted to attend the concert. Female musicians are not allowed to perform, even in restaurants and bars. Since December 2019, female and male musicians at the Conservatory of Music in Gaza are no longer allowed to perform together. The USDoS has reported on cases where teachers at Hamas schools in Gaza have sent girls home for not wearing conservative clothing. Also in 2018, the Be Different' campaigns of the Ministry of Endowments and Religious Affairs were deployed in schools and universities to promote the wearing of headscarves among girls and young women. Girls who did not wear a headscarf were required to attend two meetings of the promotional campaign.

## Legal capacity

In the Palestinian Territories, before entering into her first marriage a Muslim woman requires the permission of a guardian - a male relative/guardian (wali). This must be a male relative, in practice often from the father's side. This permission is not required for a second marriage. A quardian also has the right, under Article 165 of the Personal Status Laws (PSL), to determine where a woman under the age of forty, who is a virgin and who is under his guardianship, may live. 440 Older women can in principle marry without permission, but according to one source this is not easy due to societal norms. It is not clear from which age this applies.<sup>441</sup> According to the PSL for Muslims, in a marriage a woman must adhere to a number of specific laws. In a marriage, the husband is the head of the family and a woman must obey him. Under this law, a woman requires her husband's approval to leave the house, must move in with him (including moving abroad) and may not take a job without her husband's consent. If women do not comply, they run the risk of losing their right to financial maintenance. There are some minor differences in how the PSLs of the West Bank and Gaza are written, but they generally match, according to the Musawah organisation. The above-mentioned prenuptial agreements do not apply to Christians as they have their own legislation for the various denominations. 442 It is not clear whether a Muslim woman can evade any of these specific laws by including them in the prenuptial agreement. It is also not clear to which extent men actually exercise these rights over women in practice.

## Divorce

Men can usually easily divorce their wives, but it is legally more difficult for women to file for divorce and it often requires the husband's consent. Palestinian society is said to have a negative view of women who file for divorce or are divorced. There are different guidelines for financial compensation and alimony for divorced women and their children. There are no rules regarding the distribution of funds that have not been included in the prenuptial agreement.<sup>443</sup> According to one source, women

<sup>&</sup>lt;sup>435</sup> Arab News, Gaza police block women from attending football match, 28 January 2018, consulted on 23 December 2021.

<sup>436 +972</sup> Magazine, '*To sing is not a right in Gaza'*, 6 June 2019, consulted on 23 December 2021.

<sup>437</sup> The Guardian, Musicians decry Hamas ban on co-ed school concerts in Gaza, 22 December 2019, consulted on 23 December 2021.

<sup>438</sup> USDoS, West bank and Gaza 2020 Human Rights Report, 2021, p. 61, 71, 73.

<sup>439</sup> Al Monitor, *Hamas visits Gaza schools to promote hijab*, 31 October 2018, consulted on 23 December 2021.

<sup>440</sup> Musawah, Thematic report on Muslim family law and Muslim women's rights in Palestine, July 2018, p. 10-12.

<sup>441</sup> Confidential source, 24 January 2022.

Musawah, Thematic report on Muslim family law and Muslim women's rights in Palestine, July 2018, p. 8.
 Musawah, Thematic report on Muslim family law and Muslim women's rights in Palestine, July 2018, p. 23-24.

can go to court to file for divorce, but without legal representation the case could take more than a year to come to court.444

## Exit permits

Several sources says that women do not need an exit permit to leave the Palestinian Territories. 445 According to the USDoS, unmarried women in Gaza sometimes have trouble getting out of Gaza. Hamas controls all travel to and from Gaza (see also 1.4). 446 One source says that a man can impose a travel ban on his wife so that she cannot travel abroad, although this can be legally challenged. 447 Women are legally competent but girls under the age of 18 sometimes need permission from their father or husband to make use of government services. It is not clear when this permission is required. Divorced women who wish to leave the Palestinian Territories with their children require the permission of the children's father. 448

### Reporting sexual and gender-based violence

In the Palestinian legal system, it is possible to report incidents of (sexual) violence. These can be dealt with through the criminal justice system even if the perpetrator is the spouse or a government official. According to one source, due to different legislation and modus operandi there are regional differences in what is punishable by law and how access to protection is organised. Rape is a crime but there is no legislation that addresses marital rape. There is no existing legislation that addresses sexual harassment. 449 The Public Prosecution Service can only file proceedings if the victim presses charges. Sexual harassment must be reported no later than three months after it occurred and women must demonstrate that the violence resulted in physical damage that lasted at least ten days. It is not clear whether this requires a separate doctor's report. Women often do not report violence for fear of repercussions and/or damage to their reputation, because they cannot easily leave their home or because the perpetrator is a family member. 450 Complaints are submitted to the Family Protection Units (FPU) of the Palestinian Police. The FPU in the Palestinian Police and the Public Prosecution Service can conduct further investigations and take legal action. This National Referral System for female victims of gender-based violence was adopted in the West Bank in 2013 but does not apply in Gaza.<sup>451</sup> There are also shelters where people can go (see also the section on shelters and relief). Victims in Gaza can only report a crime at the police station, but a source says it is better if the woman is accompanied by a person from a women's organisation because women run the risk of not being believed at the police station.<sup>452</sup>

FPU can provide protection and support for female victims of violence in an emergency. 453 The FPUs can be contacted through a police station or a telephone number. The WoMA works with the Ministry of Social Development, the Attorney General's Office, the Palestinian Civil Police, the Ministry of Health, Shari'a Supreme

<sup>444</sup> Confidential source, 24 January 2022.

Confidential source, 24 January 2022; Confidential source, 26 January 2022.

<sup>446</sup> USDoS, West bank and Gaza 2020 Human Rights Report, 2021, p. 43.

<sup>447</sup> Confidential source, 26 January 2022 448 Confidential source, 24 January 2022

<sup>&</sup>lt;sup>449</sup> USDoS, West bank and Gaza 2020 Human Rights Report, 2021, p. 58-59; Confidential source, 26 January 2022. 450 UNICHA, State of Palestine: Gender Justice & the Law Law and gender, 2018, p. 17; WCLAC, Access to justice for women victims of violence in Palestine, 2021, p. 37-38; WCLAC and Al-Muntada, Femicide in the Palestinian Society, April 2021, 5.

<sup>&</sup>lt;sup>451</sup> Palestinian Police, Family Protection Unit and Juvenile Unit, consulted on 16 November 2021. See also **Error!** Hyperlink reference not valid.; Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>452</sup> Confidential source, 24 January 2022.

<sup>&</sup>lt;sup>453</sup> UN Development Programme, *Everyday hero: Access to justice for Palestinian women*, 4 December 2018, consulted on 16 November 2021.

Council and the Civil Society Forum to combat gender-based violence.<sup>454</sup> Women living in East Jerusalem fall under the jurisdiction of the Israeli Police and different legislation under Jordanian, Israeli or Sharia courts (under Jordan<sup>455</sup>). According to one source, the law and court used varies on a case-by-case basis. The source says that women often fail to report to the Israeli police for fear of being seen as collaborators by their families. In addition, women with expired residence permits are said not to report sexual and gender-based violence for fear of being sent back to the West Bank after a divorce and losing access to and/or custody of their children.<sup>456</sup>

The ICHR says that women are at risk of violence by police personnel both in and out of detention. In 2017 the ICHR registered 45 such cases, of which thirteen were in the West Bank and thirty-two in Gaza. $^{457}$ 

#### Shelters and assistance

There are three shelters in the West Bank: *Mehwar Centre* in Bethlehem, the Family Defence Society shelter in Nablus and the Women's Centre for Legal Aid and Counselling (WCLAC) emergency shelter in Jericho. There are two shelters in Gaza, one managed by the Hamas authorities (*Beit al Aman*) and one by women's organisations (*Hayat Centre*). According to one source, the management of the former shelter in Gaza is not functioning as it should. In East Jerusalem, women have access to Israeli shelters. The Directorate of Social Affairs and Child Safety and Protection Network, both part of the Palestinian Ministry of Social Affairs, also provide access to protection, shelters and emergency aid for women who are the victims of violence. UNRWA is said to refer victims in refugee camps to professional help by way of Public Service Centres.

In 2020, WCLAC assisted seven hundred women with its free telephone helpline. This was an increase of 160% over 2019. By November 2021, WCLAC had helped three hundred new cases through this helpline. Here is also a One Stop Centre in Ramallah, which, as part of a partnership between UN Women, UNDP and UNICEF under the Sawasya development programme, provides medical and legal aid, temporary shelter and police protection for female and underage victims of violence. Several organisations offer a telephone helpline and psychosomatic, legal and medical assistance in addition to advocacy and awareness campaigns (such as WCLAC, SAWA, AISHA Association for Women and Child protection and the Women's Affair Centre in Gaza, Al Muntada, TAKAMOL, Palestinian Family Planning and Protection Association). According to one source, all the services are provided

<sup>454</sup> UNDP, State of Palestine: Gender Justice & the Law, 2018, p. 14.

 $<sup>^{\</sup>rm 455}$  Confidential source, 11 April 2022.

<sup>456</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>457</sup> ICHR, The Report of the Independent Commission for Human Rights (ICHR) On The Initial Report of the State of Palestine Presented to The United Nations Committee on The Elimination of All Forms of Discrimination against Women, July 2018, p. 6.

<sup>&</sup>lt;sup>458</sup> UNDP, State of Palestine: Gender Justice & the Law, 2018.

<sup>459</sup> Confidential source, 24 January 2022.

<sup>&</sup>lt;sup>460</sup> MIFTAH, Booklet on Institutions Providing Emergency Protection Services to Battered Women: West Bank and Gaza Strip, December 2015, p. 7, 9, 13-15, 21.

<sup>461</sup> Al Jazeera, Palestine: Femicide highlights need for domestic violence law, 29 November 2021, consulted on 30 November 2021.

<sup>&</sup>lt;sup>462</sup> UN Development Programme, *Everyday hero: Access to justice for Palestinian women*, 4 December 2018, consulted on 16 November 2021.

<sup>463</sup> UNDP, State of Palestine: Gender Justice & the Law, 2018; International Planned Parenthood Federation, Palestinian Family Planning and Protection Association (PFPPA), consulted on 12 March 2022. See also Error! Hyperlink reference not valid.

through the government but due to capacity shortages these services are provided by local or national NGOs.  $^{464}$ 

## Court cases and alternative protection

The WCLAC says that women face discrimination in court and are themselves at risk of being prosecuted for adultery in cases of sexual assault. Of the 22 court cases that the WCLAC monitored in 2020 in which violence against women occurred, all the cases involved a relaxation/reduction of sentences for the perpetrator. In all abuse cases (14 cases out of 22 court cases in 2020) there was reconciliation with the perpetrator. The WCLAC says that victims are often blamed for violent incidents and/or are mistreated by their family and/or environment when they file proceedings. Women who experience domestic violence are often unable to change their situation because of financial dependence on their husband if he is the perpetrator. Social barriers, such as keeping the family together and social stigmas when broaching the subject of abuse and violence, are also a problem.<sup>465</sup>

According to one source, women also go back to their husbands for fear of losing custody of their children over the age of eight and nine or because they cannot find a home after spending time in shelters. In addition, staying with their family would not be attractive to women. According to this source, women then run the risk of not seeing their children, of having their freedom of movement curtailed and of having their salary withheld by family members. In addition, women living with relatives are said to be more at risk of sexual and gender-based violence.<sup>466</sup>

### 3.2.8 (Unaccompanied) minors

At the age of eighteen, a person is of majority age and considered to be competent in the Palestinian Territories. Compulsory education in the Palestinian Territories applies from the ages of six to fifteen. There are special provisions for Palestinian minors that are arrested by Israeli forces and prosecuted in the Israeli military court. Minors between the ages of twelve to fourteen and between the ages of fourteen and fifteen are viewed differently in terms of legal liability. From the age of sixteen, Palestinian minors are regarded as adults in the Israeli military legal system (see also 3.3.2 and 3.3.4). According to one source, the age of majority was raised to eighteen in 2011. Despite this, military law does not properly protect minors. However, another source says that, in practice, minors aged sixteen and seventeen are still regarded as adults in this system.

### Marriage age for minors

Child marriages are known to occur in the Palestinian Territories. According to one source, 20% of married women are married before the age of eighteen (as a girl).<sup>471</sup> Since November 2019, the minimum marriage age in the West Bank is eighteen, according to presidential decree. However, this age can be lowered by the Sharia

<sup>464</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>465</sup> WCLAC, Access to justice for women victims of violence in Palestine, 2021, p. 31, 38.

<sup>466</sup> Confidential source, 24 January 2022.

<sup>467</sup> UNICEF, State of Palestine Country report on out-of-school children, July 2018, p. 10; Confidential source, 10 February 2022.

<sup>468</sup> DCIP, No way to treat a child: Palestinian children in the Israeli military detention system, April 2016, p. 51; Confidential source, 27 January 2022.

<sup>469</sup> Confidential source, 14 April 2022.

 $<sup>^{\</sup>rm 470}$  Confidential source, 27 January 2022.

<sup>&</sup>lt;sup>471</sup> Confidential source, 24 January 2022.

court in certain situations, such as in the case of pregnancy. 472 In the past, the minimum marriage age in the West Bank was fifteen for girls and sixteen for boys. In Gaza, the minimum marriage age is seventeen for girls and eighteen for boys. Various laws permit the minimum marriage age to be lowered, but it is not clear how these laws relate to each other. In Gaza, a judge may authorise a marriage for children under the minimum age if these minors have reached puberty and the girl's guardian (wali), often the father or grandfather, agrees to the intended marriage. Under the Egyptian Family Rights Law of 1954, the absolute minimum marriage age is nine years old for girls and twelve years old for boys. According to the Chief Islamic Justice (Qadi-Al-Quda) of 1996, this is fifteen years old for girls and sixteen years old for boys. However, aiding a girl under the age of fifteen to get married is punishable under Article 279(2) of the 1960 Criminal Code in the West Bank and Article 182 of the 1936 Criminal Code in Gaza. 473 Forced marriages are prohibited in the West Bank. The MoWA defines forced marriage as compelling a woman to marry through the use of temptation, payment or the power that her guardians have over her. There is no such law in force in Gaza. In both regions, however, both partners must grant their permission to legally conclude a marriage. 474

## Supervision and care

One source states that the supervision of unaccompanied minors and the care of these minors is regulated by the Palestinian authorities but that in practice there may be exceptions and individual cases may deviate from the norm. The following information focuses on the situation in the West Bank and Gaza but not East Jerusalem. It is not known how this system works in East Jerusalem.<sup>475</sup>

## Guardianship

In the Palestinian Territories, there are two categories of minors without parents: a minor who has lost both parents or a minor who has lost one parent. Orphans of Muslim descent who have lost both parents come under the guardianship of the paternal grandfather. According to one source, Christian courts have different personal status laws that decide on custody for each denomination. The Personal Status Laws in the Palestinian Territories differentiate guardianship between guardianship (wilaya) and custody (hadana). Wilaya is usually granted to the father and includes the right to make decisions for the child. Hadana involves the daily care of the child and is usually granted to the mother until the child reaches a certain age. Among Muslims, the father is the guardian of his minor children. Children who lose their mother therefore remain legally under the guardianship of their father. Palestinian minors without a father are seen as a special kind of orphan.

Divorced women have the right to (hadana) custody of their minor children. In the West Bank, this is up to the age of twelve for daughters and the age of ten for sons

<sup>&</sup>lt;sup>472</sup> The New Arab, Palestinian Authority outlaws child marriage in occupied West Bank, 7 November 2019, consulted on 12 March 2022; UNFPA, Seeking protection: Survivors of sexual violence and their access to services in Palestine, February 2020, p. 21.

<sup>473</sup> Musawah, *Thematic report on Muslim family law and Muslim women's rights in Palestine*, July 2018, p. 14-15.

<sup>474</sup> Musawah, Thematic report on Muslim family law and Muslim women's rights in Palestine, July 2018, p. 11, 14-15; The Palestinian National Authority Ministry of Women's Affairs National Strategy to Combat Violence Against Women 2011-2019, 2011, p. 66.

<sup>&</sup>lt;sup>475</sup> Confidential source, 10 February 2022.

<sup>&</sup>lt;sup>476</sup> Confidential source, 10 February 2022.

<sup>&</sup>lt;sup>477</sup> NRC, The Shari'a Courts and Personal Status Laws in the Gaza Strip, January 2011, p. 22; Musawah, Palestine Overview of Muslim family law and practices, 21 May 2017, p. 25-27.

<sup>478</sup> UNDP, State of Palestine: Gender Justice & the Law, 2018, p. 20; Confidential source, 1 May 2022.

 $<sup>^{\</sup>rm 479}$  Confidential source, 10 February 2022.

<sup>&</sup>lt;sup>480</sup> UNRWA, Gaza situation report 87, 9 April 2015, consulted on 29 November 2021.

and in Gaza up to the age of eleven for daughters and nine for sons. After puberty, boys can choose which parent they want to live with; girls are then automatically placed under the guardianship of the father. In both the West Bank and Gaza, a divorcee who remarries loses her right to (hadana) custody of her children. In 2018, the PA stated that women who have (hadana) custody of their children may open a bank account for them, choose/change the children's school and apply for a passport for them.481

Widows are given (hadana) custody of their children with no specific legal minimum age. In the Gaza Strip, this is said to apply to girls until they marry and to boys until the age of eighteen. According to some sources, widows lose custody of their children when they remarry. These children then come under the full guardianship of the paternal grandfather. 482

Children who are not placed under the guardianship of close relatives automatically come under the guardianship of the PA (Ministry of Social Development) until they reach the age of majority. If distant relatives can take care of these children, the relatives receive (hadana) custody of the children, but the authorities remain legal guardians (guardianship). 483 According to one source, in such cases a Sharia court can also appoint a family member as legal guardian.<sup>484</sup>

A government agency (Committee for Orphans) manages all the finances (for example from an inheritance) of minor orphans. According to one source, the preferred course of action is to place the children with relatives, but this often does not take the best interests of the child into account. Due to the social structure of Palestinian society and the governmental structures, it is generally assumed that a child will be cared for by the family or the authorities, according to one source. However, it can happen that a child ends up on the street because of problems between families. Restrictions in relation to registration and freedom of movement between Gaza, the West Bank and East Jerusalem are said to also play a role. As a result, children who come from another region run the risk of becoming stateless because they cannot register in their new place of residence. Minors who are stateless can be granted humanitarian registration from the Israeli authorities, who temporarily suspend their statelessness. When they reach the age of eighteen, they lose this status and become stateless again.<sup>485</sup>

## Illegitimate children

Illegitimate children who are abandoned come under the care (hadana) and guardianship of the Palestinian authorities. The authorities give the child an identity and the child's religion is decided through the courts. Sometimes this is indicated, for example, by the location where the child was abandoned – for example a church. If there is no indication of the child's religion, the child is considered to be a Muslim and a name is chosen through the Sharia court. According to one source, most of these children are newborns and are adopted by Palestinians. If they are not adopted, they remain under the guardianship of the authorities. 486

### Illegitimate children

<sup>&</sup>lt;sup>481</sup> UNDP, State of Palestine: Gender Justice & the Law, 2018, p. 20; Musawah, Thematic report on Muslim family law and Muslim women's rights in Palestine, July 2018, p. 30.

482 Confidential source, 10 February 2022; Confidential source, 1 May 2022.

483 Confidential source, 10 February 2022; Confidential source, 1 May 2022.

<sup>484</sup> Confidential source, 1 May 2022.

<sup>485</sup> Confidential source, 10 February 2022.

<sup>&</sup>lt;sup>486</sup> Confidential source, 10 February 2022.

Illegitimate children who are abandoned come under the care and guardianship of the Palestinian authorities. The authorities give the child an identity and the child's religion is decided through the courts. Sometimes this is indicated, for example, by the location where the child was abandoned – for example a church. If there is no indication of the child's religion, the child is considered to be a Muslim and a name is chosen through the Sharia court. According to one source, most of these children are newborns and are adopted by Palestinians. If they are not adopted, they remain under the guardianship of the authorities. 487

#### Care of minors

There are institutions for minors in the Palestinian Territories. Orphans who cannot be placed with relatives and children born out of wedlock can be cared for in these institutions. According to one source, there are around thirteen institutions in the Palestinian Territories for different ages of children. There is said to be one institution for children up to the age of six. For older children, besides institutions that provide only accommodation there are also institutions that provide education in addition to accommodation. The source did not know what the conditions were like in these institutions.<sup>488</sup>

### **Organisations**

According to one source, the care of abandoned minors/orphans is a domestic matter that is managed by the Palestinian authorities. A number of domestic religious organisations and churches are said to be active in this system and to run institutions. For example, the Palestinian NGO Al-Amal Orphan Society runs an orphanage in Gaza. In Gaza, the Wajd program (Qatar Fund for Development) provides assistance in the form of psychosocial support, education, health care and training to more than 2,165 orphans and fatherless children and their mothers. In addition, other organisations and NGOs, such as Islamic Relief, Interpal, Orphans in Need, Skt Welfare, Muslim Global Relief, Zakat Foundation, UPA and Palestine Children Relief Fund (PCRF), provide relief (including clothing, food and medicine) to orphans in the Palestinian Territories.

International organisations are said to be active only in providing assistance to children who need specific protection due to violent incidents. <sup>492</sup> The organisation SOS Children's Villages has been active in the Palestinian Territories since 1968. The organisation's head office is located in Bethlehem. Its activities consist of providing care (healthcare and education) to children, supporting vulnerable families in order to keep them together, placing children with temporary foster parents (SOS families) and providing temporary care for children in an Interim Care Centre where children can stay until they return to their families. There are two SOS children's villages: one in Rafah (Gaza Strip) and one in Bethlehem (West Bank). SOS children's villages reached almost 4,000 children and young people and 705 families in 2020. <sup>493</sup>

<sup>&</sup>lt;sup>487</sup> Confidential source, 10 February 2022.

<sup>&</sup>lt;sup>488</sup> Confidential source, 10 February 2022. <sup>489</sup> Confidential source, 10 February 2022.

<sup>&</sup>lt;sup>490</sup> The Independent, *Gaza's children - resilient or traumatized*, 15 January 2022, consulted on 17 January 2022;

<sup>491</sup> Islamic Relief, International charity launches campaign to change lives of Palestinian orphans, 5 February 2018, consulted on 29 November 2021; Palestine Children's Relief Fund (PCRF), Annual report 2020, 2020, p. 26.

<sup>&</sup>lt;sup>492</sup> Confidential source, 10 February 2022.

<sup>&</sup>lt;sup>493</sup> SOS Children's Villages Palestine, *Annual Report 2020*, 2020, p. 16-17.

## 3.3 Oversight and legal protection

### 3.3.1 Judicial process

### Reporting a crime

Palestinians can report a crime at a police station and by telephone. The public prosecutor determines whether it is necessary to subsequently record the report in writing. Certain crimes, such as gender-based violence, can only be reported by the victim. 494 According to one source, crimes that are reported are generally taken seriously and processed. There is a chance that reports of some crimes, such as gender-based violence, will not be taken seriously. In some cases, women are sent back to their families when they report a crime at a police station. Reports of crimes are generally processed promptly at the start of the process. This source says that the problem lies more with the length of the court case; due to limited capacity, rulings are said to take a long time. The source adds that rulings of court cases dating from 2020 had still not been made by early 2022, partly due to the backlog caused by the coronavirus pandemic. 495

In court cases against political opponents or critics, extensions and the lack of witnesses are said to work to the advantage of the public prosecutor. For example, they can investigate new evidence or assess the trial based on the lack of witnesses. According to some sources, people cannot seek protection from other (higher) authorities if the police are in default unless they have influential contacts higher up. Political opponents or critics of the PA or Hamas are said not to have access to such contacts. NGOs or international organisations cannot help people to obtain protection but they can provide certain services such as legal aid and representation. One source says that legal aid does not begin immediately after arrest but usually only after a court case has started.

### Protection

According to a public opinion poll held in December 2021, 51% of Palestinians in the West Bank and 79% of Palestinians in Gaza perceive the security situation as safe. 499 The protection of civilians by the authorities, according to sources, is said to depend on the incident and on the individual it concerns. Political opponents, critics of the authorities and LGBTI people cannot count on protection, and women can also face discrimination when trying to obtain protection. 500 According to one source, having certain contacts helps to obtain protection but this occurs on a case-by-case basis. There is said to be no protection for people who have problems with leaders in the authorities. 501 According to one source, when Palestinians have problems with the (de facto) authorities, they cannot resettle from Gaza to the West Bank or vice versa without the permission of the Israeli authorities. In practice, settlement from Gaza to the West Bank or vice versa is not possible, according to the source. Palestinians who are seen as a threat to national security are usually detained by the Palestinian authorities. The source was not aware of any cases of nationals being expelled from the Palestinian Territories. 502 According to two sources, the Palestinian

<sup>494</sup> Confidential source, 23 February 2022.
495 Confidential source, 3 February 2022.
496 Confidential source, 3 February 2022.
497 Confidential source, 3 February 2022; Confidential source, 3 February 2022.
498 Confidential source, 3 February 2022.
499 Palestinian Center for Policy and Survey Research, *Public Opinion Poll No (82)*, 14 December 2021, p. 4, 10.
500 Confidential source, 3 February 2022; Confidential source, 3 February 2022.
501 Confidential source, 3 February 2022.
502 Confidential source, 3 February 2022.
503 Confidential source, 3 February 2022.

authorities cannot protect their nationals from detention by the Israeli authorities or mediate for their release. 503

According to one source, in the West Bank, the PA is unable to provide protection if there are problems involving armed groups, family feuds or violence in refugee camps. In Area B (see also 1.3.3) of the West Bank, people are said to receive no government welfare, neither from the PA nor from Israel, which means they risk not having access to an ambulance, police or the fire service, for example. 504

#### 3.3.2 Arrests, custody and detentions

Palestinians face illegal detention by both the Palestinian and Israeli security services. 505 In 2020, the Palestinian organisation Al-Haq registered hundreds of violations by the PA and the 'de facto' [Hamas] authorities in Gaza, including unlawful detention, unfair trials, mistreatment, physical assault, torture, unlawful confiscation of property, violations of the right to freedom of expression and inhumane prison conditions. 506 According to B'Tselem, at the end of September 2020 4,184 Palestinian 'security' detainees and convicted prisoners from the Palestinian Territories were being held in Israeli prisons for security violations. The Israeli Prison Service (IPS) sees this group as criminal offenders. In addition, there were 545 Palestinians detained in Israeli prisons because they were found to be in Israel illegally. 507 Organisation Addameer 508 referred to a total of 4,650 political prisoners at the end of 2021 and five hundred people in administrative detention, which means that people are being held without charge and/or the intervention of the police, court or other government body. 509 In principle, this detention period can be imposed for a maximum period of six months and can be extended. It is not clear how often it can be extended. 510

Administrative detainees fall under the jurisdiction of the IPS and are held in the following prisons along with regular detainees: Ofer Prison south of Ramallah, Ketziot Prison (also called Ansar or Negev) in the Negev desert and Megiddo Prison on the road between Jenin and Haifa. 511 Individuals can spend years in administrative detention. This happened to Mohammed El Halabi, an employee of American relief organisation World Vision in Gaza. He is suspected of passing relief funds to Hamas and has been in administrative detention since 2016.512 This example is intended to illustrate these types of cases and is by no means an isolated incident.

Access to a lawyer in the West Bank

<sup>&</sup>lt;sup>503</sup> Confidential source, 27 January 2022; Confidential source, 3 February 2022.

<sup>504</sup> Confidential source, 7 February 2022.
505 Human Rights Watch, *Abusive System: Failures of Criminal Justice in Gaza*, September 2012, p. 17; Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021; The New York Times, Critic's Death Puts Focus on Palestinian Authority's Authoritarianism, 7 July 2021, consulted on 27 July

<sup>&</sup>lt;sup>506</sup> Al-Haq, *Al-Haq Field Report on Human Rights Violations in 2020*, 2020, p. 20-21.

<sup>&</sup>lt;sup>507</sup> B'Tselem, *Statistics on Palestinians in the custody of the Israeli security forces*, 24 November 2021, consulted on 4 March 2022.

<sup>&</sup>lt;sup>508</sup> Addameer is one of the six organisations suspected of having terrorist links by the Israeli authorities in October 2021. On 19 September 2019, this organisation's office in Ramallah was raided and the authorities confiscated laptops, hard drives and other equipment. Addameer, Annual Report 2019, 2019, p. 9.

<sup>&</sup>lt;sup>509</sup> Addameer, *Prisoners Statistics*, 7 November 2021, consulted on 10 December 2021.

<sup>&</sup>lt;sup>510</sup> Addameer, *Adminstrative detainees*, December 2019, consulted on 10 December 2021.

<sup>&</sup>lt;sup>511</sup> Addameer, *Adminstrative detainees*, December 2019, consulted on 10 December 2021.

<sup>512</sup> The Guardian, Israel under pressure to conclude flawed case against aid worker, 28 February 2022, consulted on 12 March 2022.

Suspects have legal access to a lawyer under Article 12 of the PLO Penal Procedure Law but this access is not always granted. After arrest, under the 2001 Palestinian Penal Code that is in force in the West Bank, Palestinian police may detain a person for 24 hours without allowing access to a lawyer. After this, the person may be held in custody for another 48 hours before being brought to court. Then the court can extend the pre-trial detention period by fifteen days. A higher court may also extend a detention order by an additional 45 days up to a maximum of six months. In military courts, under Palestinian Revolutionary Penal Procedure Law No. 79 of 1979, a person may be detained without charge for a period of fifteen days, which may be extended for a total of 45 days. This applies when a suspect is believed to represent a threat to society or there is a risk that the suspect will destroy evidence or flee the Palestinian Territories. Detainees/suspects can be held without charge for up to six months under Article 88 of the Revolutionary Penal Procedure Law for crimes against the security of the Palestinian revolution. 513 According to one source, political opponents and critics of the authorities do not have immediate access to a lawyer (see also 3.3.1).<sup>514</sup>

#### Access to a lawyer in Gaza

In Gaza, the judiciary is said to assign a lawyer to people who cannot afford one themselves. The bar association is said to also help in certain cases. Because the number of suspects and detainees in Gaza is not publicly known, it is not clear whether there are enough lawyers in Gaza to cover all the criminal cases. According to one source, judges and court employees must be supporters of Hamas if they want to work in the judiciary. Lawyers who are critical of Hamas or who support Fatah are at risk of intimidation, prosecution and torture. Lawyers sometimes have to deal with cases where their clients cannot be reached or their detention location is not clear – for example because their clients are being detained by the security services. The military court in Gaza is legally obliged to assign a lawyer to persons who are not legally represented. 515

## Access to a lawyer in Israel

Most Palestinians in Israeli prisons are being detained on suspicion of endangering security. They have access to a lawyer who is registered in Israel. It can take up to ninety days after arrest for a person to be given access to a lawyer (see also the following section). This period can also be extended.<sup>516</sup>

## Access to a lawyer for minors

Minors in the West Bank are at risk of detention in Israeli prisons. According to one source, this is almost non-existent in Gaza. There are hardly any arrests of Palestinians by Israeli troops in Gaza. When this does happen, detainees are usually released after a few hours (see also 1.4.1). Palestinian minors arrested by Israeli forces are usually suspected of throwing stones, an offence that carries a sentence of ten to twenty years in detention under Israeli military law. It is not clear what the different lengths of punishment are based upon. These minors are tried by the Israeli military court. Depending on their age, minors must appear before a military court within 24 to 72 hours of arrest. Their appearance in court is often the first

<sup>&</sup>lt;sup>513</sup> Human Rights Watch, *Abusive System: Failures of Criminal Justice in Gaza*, September 2012, p. 15.

<sup>514</sup> Confidential source, 3 February 2022. 515 Human Rights Watch, *Abusive System: Failures of Criminal Justice in Gaza*, September 2012, p. 8-11, 13.

 $<sup>^{\</sup>rm 516}$  Confidential source, 2 February 2022.

<sup>517</sup> Confidential source, 27 January 2022.

time minors see a lawyer or their family. <sup>518</sup> According to one source, children can be detained for several hours without charge and the period of detention can be extended to twice the legal length. This requires a judge to extend the pre-trial detention. In some cases, children are denied access to a lawyer and face charges without legal representation. <sup>519</sup> Minors are often questioned without the presence of a lawyer or parental guardian. <sup>520</sup> One source says that over the past two years there has been an increase in cases of children being interrogated before being granted access to a lawyer. <sup>521</sup> This happened even though, prior to an interrogation, the lawyer of a minor must be notified of this in advance. Under Israeli military law, however, access to a lawyer may be refused for up to ninety days if this is beneficial to the interrogation or the security situation. <sup>522</sup> The interpretation of the security situation is not clear.

Minors can be interrogated at a police station or in a prison.<sup>523</sup> Palestinian minors who have been interrogated say that they were not informed during detention that they were entitled to have access to a lawyer. The majority of respondents said that they had not spoken to a lawyer before the interrogation. In some cases, minors were asked to supply the phone number of a lawyer and failure to provide this number was construed as surrendering the right to a lawyer.<sup>524</sup>

According to one source, lawyers are notified by telephone by the Israeli police that a minor has been arrested and that an interrogation will take place. This source also says that the interrogation often begins before a lawyer can reach the place of detention. Minors usually only see a lawyer for the first time a few minutes before appearing in a military court. Two family members may be present at the trial. These relatives do have eye contact but no physical or direct verbal contact with the suspect. To avoid a longer legal process, court cases usually end with a settlement. Suspects are often asked to sign an admission of guilt in a document written only in Hebrew. Me 297 cases conducted by lawyers for Defence for Children International Palestine (DCIP) between 2012 and 2015, 295 ended in a settlement. According to one source, 99% of cases still ended in a settlement and this course of action was said to be the fastest way to have a child released from detention. It is not clear whether a settlement is possible in cases involving repeat offences.

<sup>&</sup>lt;sup>518</sup> Punishable by military order 1651. DCIP, No way to treat a child: Palestinian children in the Israeli military detention system, April 2016, p. 2; Hamoked, Childhood in Chains: The detention and interrogation of Palestinian Teenagers in the West Bank, April 2018, p. 1-2.

<sup>&</sup>lt;sup>519</sup> Confidential source, 27 January 2022.

<sup>520</sup> Hamoked, Childhood in Chains: The detention and interrogation of Palestinian Teenagers in the West Bank, April 2018, p. 1-2; Freedom House, West Bank: Freedom in the World 2021 Country Report, consulted on 2 November 2021.

<sup>&</sup>lt;sup>521</sup> Confidential source, 27 January 2022.

<sup>&</sup>lt;sup>522</sup> According to the Security Provisions (Amendment No. 10) (Judea and Samaria) (No. 1676), 5781-2011, 27 September

<sup>2011.</sup> Now section 136c in Security Provisions. See also Department of Justice, Note 13, section 19 and Security Provisions, sections 56-59. B'Tselem, Minors in Jeopardy, Violations of the Rights of Palestinian Minors by Israel's Military Courts, March 2018, p. 21.

<sup>523</sup> B'Tselem, Minors in Jeopardy, Violations of the Rights of Palestinian Minors by Israel's Military Courts, March 2018, p. 32.

<sup>&</sup>lt;sup>524</sup> B'Tselem, Minors in Jeopardy, Violations of the Rights of Palestinian Minors by Israel's Military Courts, March 2018, p. 8-9, 34; Hamoked, Childhood in Chains: The detention and interrogation of Palestinian Teenagers in the West Bank, April 2018, p. 4.

<sup>525</sup> Confidential source, 27 January 2022.

<sup>526</sup> B'Tselem, Minors in Jeopardy, Violations of the Rights of Palestinian Minors by Israel's Military Courts, March 2018, p. 8-9, 34; Hamoked, Childhood in Chains: The detention and interrogation of Palestinian Teenagers in the West Bank, April 2018, p. 5.

<sup>527</sup> B'Tselem, Minors in Jeopardy, Violations of the Rights of Palestinian Minors by Israel's Military Courts, March 2018, p. 8-9, 34.

<sup>&</sup>lt;sup>528</sup> Confidential source, 27 January 2022.

Minors are also at risk of arrest during night-time raids (even when they are staying at home or with relatives). 529 A source says that mothers in the West Bank often keep their children indoors for fear of arrest by Israeli troops. This is also said to be having an impact on the freedom of movement of mothers in the West Bank. 530

### Help from organisations

Families of minors are often not told where their children are being held. They must contact the Red Cross and the Palestinian Prisoners' Society to find out if and where a child is being held. Other organisations such as Defence for Children International Palestine (DCIP), Addameer<sup>531</sup> and the PA Commission of Detainees and Exdetainees Affairs can also notify families of arrested minors and provide legal assistance and representation. 532 According to one source, DCIP handles around a third of the cases for minors. 533 Organisation Addameer provides legal aid and representation for Palestinian detainees in both Israeli and Palestinian prisons. In 2019, Addameer helped 89 families with legal advice, provided legal representation for 67 inmates in Palestinian prisons and arranged representation in 683 court cases in the Israeli justice system. Addameer also provides training to Palestinians in the West Bank and Jerusalem in order to raise awareness about the rights of detainees and about the Israeli military justice system. In 2019, fifty inmates in Israeli prisons were said not to have had access to a lawyer. Organisation Addameer was not allowed to report on this between September and December 2019 due to censorship by Israeli courts. 534

### Prison conditions

Addameer says that mainly journalists, activists and students are at risk of detention by the PA. Offences and torture are said to take place in PA prisons. It is not clear which types of offences this involves. Addameer says that it is difficult to visit and monitor PA detention centres and prisons.<sup>535</sup> According to one source, the PA is known to be guilty of torture in the prisons in Jericho and Nablus.<sup>536</sup> There is very little information available about the situation in Palestinian prisons. As a result, in this country of origin information report it is not possible to paint a clear picture of the situation in these prisons.

Palestinian prisoners and Palestinian administrative detainees in Israeli prisons are restricted in terms of their right to education and medical assistance, communication with family and receiving visitors. Addameer said inmates are at risk of solitary confinement, may be kept in isolation and are denied access to the cafeteria where additional food and other products can be purchased. Besides medical neglect, most prisons are said not to meet international standards in relation to overcrowding, personal hygiene facilities and climate control. For example, there were also problems with insects and vermin in the prisons. Palestinian detainees in Israeli

<sup>&</sup>lt;sup>529</sup> Hamoked, *Childhood in Chains: The detention and interrogation of Palestinian Teenagers in the West Bank*, April 2018, p. 2-3; Hamoked, Under Cover of Darkness: Night arrests of Palestinian minors, by Israeli Security Forces in the West Bank, October 2020; Confidential source, 26 January 2022.

 $<sup>^{\</sup>rm 530}$  Confidential source, 26 January 2022.

<sup>531</sup> Addameer, de Prisoner Support and Human Rights Association, is a Palestinian NGO that helps Palestinian political prisoners in Israeli and Palestine prisons.

BTselem, Minors in Jeopardy, Violations of the Rights of Palestinian Minors by Israel's Military Courts, March 2018, p. 8-9, 34; Hamoked, Childhood in Chains: The detention and interrogation of Palestinian Teenagers in the West Bank, April 2018, p. 7; Confidential source, 27 January 2022.

<sup>533</sup> Confidential source, 27 January 2022.

<sup>&</sup>lt;sup>534</sup> Addameer, *Annual Report 2019*, 2019, p. 6, 9, 11, 13.

<sup>&</sup>lt;sup>535</sup> Addameer, *Annual Report 2019*, 2019, p. 17.

<sup>&</sup>lt;sup>536</sup> Confidential source, 3 February 2022.

prisons are only permitted to study at the Israeli open university and are not allowed access to Arab universities for security reasons. 537

According to one source, inmates in Israeli prisons are not keeping medical care and court appointments because they are afraid of the inhumane transport conditions. The travel time for this type of transport is supposed to be several hours but in practice it takes one to two days for detainees. Detainees are said to be held in a cage with no air circulation or access to a toilet and the lights are kept on all night. 538

According to the Palestinian Prisoners Society (PPS), 72 Palestinian prisoners are said to have died of medical neglect in Israeli prisons since 1967. One source says that during the reporting period, women in administrative detention in Israeli prisons were at high risk of contracting the coronavirus due to poor safety and health regulations. 540

## Hunger strikes

Hunger strikes are used by convicted Palestinian prisoners and detainees to demand their release from administrative detention or to draw attention to the poor living conditions in Israeli prisons. In April 2019, hundreds of inmates went on hunger strike to protest against violations of their basic rights, including the use of public telephones and conditions for family visits. A further group of around fifty administrative detainees went on hunger strike against their prolonged pre-trial detention without charge.<sup>541</sup> In 2021, a number of inmates in Israeli prisons went on hunger strike to demand their release after a lengthy period in administrative detention.<sup>542</sup> On 22 November 2021, it was announced that two people from this group, Kayed al-Fasfous and Ayyad al-Hreimi, would be released. Fasfous was released on 5 December 2021, and an agreement was made in early January 2022 to release another detainee, Hisham abu Hawash. In solidarity with Abu Hawash, who was on hunger strike, around five hundred Palestinian prisoners in administrative detention had initiated a boycott by failing to appear before the Israeli military court. Other detainees who remained on hunger strike included Loai al-Ashkar and Nidal Balout. The prolonged hunger strike has had lasting effects on the medical condition of former detainee Maher al-Akhras, who was released after his hunger strike. Over the past two years, at least eleven people are said to have gone on hunger strike to demand their release. 543 On 24 February, Miqdad Qawasmi was released after a 113-day hunger strike. 544 It is not clear how many people have actually been released. In March 2022, thousands of Palestinian prisoners went on

<sup>537</sup> Addameer, Key Issues: Isolation, Restrictions on family visits, July 2017, consulted on 10 December 2021; Addameer, Key Issues: Medical Negligence, July 2016, consulted on 10 December 2021; Addameer, Adminstrative detainees, December 2019, consulted on 10 December 2021.

Source, 2 February 2022.

<sup>539</sup> AL Jazeera, Palestinian prisoner from Gaza dies in Israeli custody, 18 November 2021, consulted on 19 November 2019.

<sup>540</sup> Confidential source, 26 January 2022.

<sup>541</sup> Addameer, *Annual Report 2019*, 2019, p. 6; Confidential source, 2 February 2022.

<sup>542</sup> BBC Monitoring, Palestinians rally behind prisoners on hunger strike in Israel, 5 November 2021, consulted on 6 November 2021.

<sup>543</sup> BBC Monitoring, Palestinian prisoners on hunger strike reach release deal, 23 November 2021, consulted on 24 November 2021; The Independent, Israel return ex-prisoner weeks after ending hunger strike, 5 December 2021, consulted on 6 December 2021; The Independent, Palestinians freed after hunger strikes have lifelong damage, 10 December 2021, consulted on 10 December 2021; The New York Times, Palestinian Prisoner's Hunger Strike Leads to Deal for Release, 4 January 2022, consulted on 5 January 2022; Al Jazeera, Palestinian prisoners launch boycott of Israeli military courts, 4 January 2022, consulted on 5 January 2022.

<sup>544</sup> Al Jazeera, 'Love for a free life': Why Palestinian prisoners hunger strike, 10 March 2022, consulted on 11 March 2022.

hunger strike in Israeli prisons in protest against worsening living conditions since the escape of six Palestinian prisoners from an Israeli prison in 2021.<sup>545</sup>

## Minors

According to DCIP, between 500 to 700 Palestinian children between the ages of twelve and seventeen are held in Israeli military detention centres every year. It is not clear how long children usually remain in detention.<sup>546</sup> In 2019 and 2020, there were no more girls in military detention.<sup>547</sup> According to one source, 41 children have been held in administrative detention since 2015.548 In 2019 and 2020, approximately one to four children were held in Israeli administrative detention each month<sup>549</sup> In 2021, this was six children and in 2022 four children. Administrative detention for minors can be imposed for up to six months but can be extended indefinitely by a military prosecutor with the permission of a judge. 550 Between January 2016 and December 2019, 1,957 Palestinian children were detained by the Palestinian authorities. DCIP reports that in 72% of the 61 cases they investigated in the West Bank between January and July 2020, children had experienced illtreatment. This varied from physical and psychological violence to poor living conditions. According to the 2016 Juvenile Protection Law, the Palestinian juvenile police force has jurisdiction over Palestinian minors. However, minors have also been arrested by other authorities, such as the Preventative Security Forces. DCIP says that the rights of children going through the juvenile police process are almost never violated. 551

Abuse and torture are widespread in Israeli military detention centres. In 2013, the Committee on the Rights of the Child stated that Palestinian children arrested by Israeli forces were being subjected to degrading treatment and torture. The organisation Military Court Watch has said that not enough improvements have been made in the Israeli military system for minors since 2013. The organisation Hamoked says that during interrogation, minors were exposed to physical and mental torture such as isolation, sleep and food deprivation, threats and physical violence. According to one source, minors are more at risk of ill-treatment and torture at the start of the process, especially during arrest and interrogation. Interrogation includes physical and psychological pressure, threats and solitary confinement. Children are not allowed to have visitors during the first three months of detention. During the entire detention period, children are not allowed to have telephone conversations with family members.

In 2015, DCIP said that the minors' section at Givon Prison operated by the Israel Prison Service was not adhering to the minimum standards. Cells were overcrowded,

<sup>&</sup>lt;sup>545</sup> Middle East Eye, Palestinian prisoners to go on hunger strike against Israel's punitive measures, 1 March 2022, consulted on 12 April 2022.

<sup>546</sup> DCIP, Military detention: Number of Palestinian Children (12-17) in Israeli Military Detention, 2 October 2020, consulted on 25 November 2021. See also Error! Hyperlink reference not valid.

<sup>547</sup> DCIP, Military detention: Number of Palestinian Girls in Israeli Military Detention, 2 October 2020, consulted on 25 November 2021. See also Error! Hyperlink reference not valid.

<sup>548</sup> Confidential source, 27 January 2022.

<sup>549</sup> DCIP, Military detention: Number of Palestinian Children (12-17) in Israeli Administrative Detention, 2 October 2020, consulted on 25 November 2021. See also Error! Hyperlink reference not valid.

<sup>550</sup> According to the same source, the military officer in question is required by law to announce a ruling to impose administrative detention in writing and sign that ruling with his/her name. According to the source, these rulings have only been signed with initials in recent times. Confidential source, 27 January 2022.

<sup>551</sup> DCIP, Juvenile Justice, consulted on 26 April 2022. See also Error! Hyperlink reference not valid.

<sup>552</sup> DCIP, No way to treat a child: Palestinian children in the Israeli military detention system, April 2016, p. 2; Tone Anderson, IKON, When the boys return, 2012.

<sup>553</sup> Military Court Watch, *The UNICEF Report – 9 years on*, 31 March 2022, consulted on 15 April 2022.

<sup>&</sup>lt;sup>554</sup> Hamoked, *Childhood in Chains: The detention and interrogation of Palestinian Teenagers in the West Bank*, April 2018, p. 10.

<sup>555</sup> Confidential source, 27 January 2022.

the building had no proper heating or showering facilities and children complained about insufficient food and its poor quality. <sup>556</sup> In a 2018 study by B'Tselem, in one case a minor said that he was kept in a two-by-three-metre cell during Israeli military detention. This cell could hold a maximum of fifteen boys, sleeping on bunk beds. There was one toilet, each bed had a mattress and a blanket, there was a television and there were books and games in the room. Detainees were given two meals a day. They could also buy packaged food in a canteen. The doors of the cell opened every half hour so that they could go to an outdoor area to take part in sports, for example. <sup>557</sup> According to one source, education in prisons consists only of language and maths lessons and all minor detainees receive the same education, regardless of their age and their class at school. In addition, the curriculum is not consistent with Palestinian education. For this reason, many children leave school when they are released after a period of detention. <sup>558</sup>

## 3.3.3 The death penalty

The death penalty can be imposed in the West Bank for seventeen different offences under the Jordanian Criminal Code of 1951. In Gaza, the death penalty can be imposed for fifteen different offences under the Criminal Code under Egyptian law. The death penalty can also be imposed according to the 1979 PLO Revolutionary Penal Code, which includes forty-two different offences recognised as criminal offences in Palestinian military courts. <sup>559</sup> Another source says that it involves 45 different offences in the Revolutionary Penal Code. In addition, according to the 1936 Criminal Code of the British Mandate, the death penalty can also be imposed for fifteen offences. Most of the offences under the various legislations relate to political crimes. <sup>560</sup> Until 2003, the PLO Revolutionary Penal Code was also used by special courts in the PA. In June 2005, Palestinian President Mahmoud Abbas announced that all persons who had been sentenced to death by this court were entitled to a new and fair trial. Eleven detainees were granted a new trial under this arrangement. <sup>561</sup>

In the West Bank, courts handed down 72 death sentences from 1995 to September 2021. Executions were actually carried out in fifteen cases, the last of which took place in 2005. The death penalty has been imposed 130 times in Gaza since 2008. Twenty-five executions were actually carried out, including some during the reporting period of this country of origin report. Some prisoners were sentenced in absentia. In most cases, death sentences are imposed by military courts. <sup>562</sup> Although the Palestinian Territories have signed the International Covenant on Civil and Political Rights, it is not possible for an individual to ask for a pardon on the

DCIP, No way to treat a child: Palestinian children in the Israeli military detention system, April 2016, p. 2.
 B'Tselem, Minors in Jeopardy, Violations of the Rights of Palestinian Minors by Israel's Military Courts, March

<sup>2018,</sup> p. 34. S58 Confidential source, 27 January 2022.

<sup>559</sup> B'Tselem, Death Penalty in the Palestinian Authority and Under Hamas Control, 16 October 2013, consulted on 27 October 2021.

Falsetinian Centre for Human Rights, On World Day against the death penalty, death penalty is not a solution, only an illusion of deterrence to avoid liability, 10 October 2020, consulted on 3 March 2022.

<sup>&</sup>lt;sup>561</sup> B'Tselem, Death Penalty in the Palestinian Authority and Under Hamas Control, 16 October 2013, consulted on 27 October 2021.

<sup>562</sup> B'Tselem, Death Penalty in the Palestinian Authority and Under Hamas Control, 16 October 2013, consulted on 27 October 2021; B'Tselem, Statisics on the death penalty in the Palestinian Authority and under Hamas control in Gaza, 13 September 2021, consulted on 27 October 2021; UN, EU and UN Human Rights Office in Palestine mark the World Day Against the Death Penalty – Press Release, 10 October 2021, consulted on 12 March 2022.

basis of this treaty. Pardons could be granted, however, by the Palestinian president. <sup>563</sup>

### 3.3.4 Ill-treatment and torture

According to some sources, Palestinian security forces in the West Bank and Gaza use violence and torture against detainees. Violence against civilians was also used during enforcement activities, killing civilians or injuring them so seriously that they required hospital treatment (see also PA critic Nizar Banat 3.2.1). Government officials or persons belonging to groups considered to be the (de facto) authorities are generally not prosecuted or punished if they are the perpetrators or suspects of acts such as torture, serious ill-treatment and rape. The PA has indicated that it will launch an investigation into these types of crimes but Amnesty says that these crimes are not being effectively investigated. 564 According to one source, it is difficult to provide evidence of these violent offences. It is claimed that the PA asks individuals who complain about such offences to submit a medical report listing their injuries. However, this report must be drawn up immediately after the offence has been committed. In practice, it would not be possible to have such a report drawn up immediately. If it comes to a trial, suspects are often not sentenced or are simply transferred to another facility. 565 In the recent trial of the fourteen security officers suspected of the death of Palestinian activist Nizar Banat, who was a critic of the PA, sums of money/damages were offered as compensation. Banat's family refused this money, fearing it would lead to lesser sentences for the perpetrators. 566 The verdict has not yet been pronounced. During the reporting period, the trial was adjourned until 18 May 2022. 567

According to HRW, the PA received 346 complaints of wrongful arrest and assault between January 2018 and March 2019. Most of the cases involve violence perpetrated by the police. Of the cases that were investigated, violations were detected in 48 cases. The Hamas authorities said that they had received 47 complaints of wrongful arrest and torture during the same period. Of the eight cases investigated, seven were subject to administrative sanctions, including transfer and detention without trial for a maximum period of one month. One case was referred to the military court but a verdict has not yet been pronounced. <sup>568</sup>

The Al Haq organisation stated in 2020 that although law enforcement agencies claim that the perpetrators of torture and ill-treatment are being punished, there are no clear/reliable figures available on what is happening with these cases. Victims are also not able to obtain compensation for torture and ill-treatment under Palestinian

<sup>&</sup>lt;sup>563</sup> UN, OHCHR Condemns in the strongest terms the execution of three men in Gaza – UNOG Press Briefing Note (Excerpts), 7 April 2017, consulted on 3 March 2022; Palestinian Centre for Human Rights, On World Day against the death penalty, death penalty is not a solution, only an illusion of deterrence to avoid liability, 10 October 2020. consulted on 3 March 2022.

Human Rights Watch, Abusive System: Failures of Criminal Justice in Gaza, September 2012, p. 18-19; Human Rights Watch, Two Authorities, One Way, Zero Dissent: Arbitrary Arrest and Torture under the Palestinian Authority and Hamas, 23 October 2018, consulted on 20 December 2021; Amnesty International, Report: State of Palestine, 2020, consulted on 27 October 2021; Al Haq, Prevention of Torture: Monitoring, Documentation and means of Confrontation, 2020.

<sup>&</sup>lt;sup>565</sup> Confidential source, 3 February 2022.

<sup>566</sup> Al Jazeera, Nizar Banat family to seek international justice 'if necessary', 24 October 2021, consulted on 27 October 2021.

<sup>567</sup> Lawyers for Justice, Trial postponed for those accused of killing activist Nizar Banat, 13 February 2022, consulted on 13 February 2022; Lawyers for Justice, Trial postponed for those accused of killing activist Nizar Banat, 11 April 2022, consulted on 12 April 2022.

<sup>568</sup> Human Rights Watch, Palestine: No letup in arbitrary arrests, torture, 29 May 2019, consulted on 20 December 2021.

law according to the PA-ratified Convention against Torture (CAT) and the Optional Protocol to the CAT (OPCAT). 569

There have also been reports of incidents of physical and psychological torture in Israeli prisons. <sup>570</sup> According to one source, most cases of serious ill-treatment and torture are said to take place during questioning/interrogation after arrest. Tactics used during interrogations include sleep deprivation, being forced to maintain stressful positions for a long period, sexual harassment and threats, such as threatening to revoke the work or travel permits of family members. <sup>571</sup> In 2019, Addameer conducted 129 prison visits in seventeen Israeli prisons and detention centres. During those visits, it documented 73 cases of torture and ill-treatment and 23 cases of medical neglect of prisoners. In 2019, Israeli forces are said to have raided Al-Naqqab, Rimon and Ofer prisons several times, firing gas and sonic bombs and bullets. Hundreds of prisoners are said to have been beaten and inspected naked. <sup>572</sup>

## 3.3.5 Disappearances and kidnappings

According to Amnesty, there are no widespread disappearances in the Palestinian Territories. However, the PA and Hamas are said to have been involved in numerous disappearances over the past twenty years. There are still hundreds of missing persons who disappeared during the wars between Israel and the Palestinians. One source says that during the reporting period political opponents and critics of the PA went missing for several days before their arrest was announced.

## 3.3.6 Extrajudicial executions and killings

According to the B'Tselem database, from 2000 to the end of 2021, more than 10,000 Palestinians – almost 2,200 of them minors – died as a result of attacks by Israeli security forces and Israeli civilians. Almost 1,300 Israelis, 139 of them minors, have died in Palestinian attacks since 2000.<sup>576</sup> The B'Tselem organisation has indicated that, based on its own investigation into the sixteen Palestinians who died in 2020 in the West Bank as a result of violence by Israeli troops, in eleven cases there was no clear threat that justified the violence.<sup>577</sup>

The following examples (which are by no means exhaustive) serve as illustrations of cases in which the Palestinian authorities, media or other organisations have referred to extrajudicial executions and killings.

Al Jazeera has reported extrajudicial killings of Palestinians by Israeli soldiers. For example, the death of Ahmed Erekat at a checkpoint between Jerusalem and Bethlehem in June 2020 was said to be unlawful because, although he posed no threat, Israeli soldiers fired on him and then refused to grant him access to first

<sup>&</sup>lt;sup>569</sup> Al Haq, Prevention of Torture: Monitoring, Documentation and means of Confrontation, 2020, p. 59-61; Al Haq, National Preventive Mechanism against Torture and Ill-Treatment, 16 August 2018, consulted on 12 March 2022.

<sup>&</sup>lt;sup>570</sup> PCATI, *Documenting torture while providing legal aid: a handbook for lawyers*, June 2016, p. 9; Confidential source, 2 February 2022.

<sup>&</sup>lt;sup>571</sup> Confidential source, 2 February 2022.

<sup>&</sup>lt;sup>572</sup> Addameer, *Annual Report 2019*, 2019, p. 6.

<sup>&</sup>lt;sup>573</sup> Amnesty International, *Neither dead nor alive*, August 2018, consulted on 23 November 2021.

<sup>&</sup>lt;sup>574</sup> Al Jazeera, Gone but not forgotten: The case of missing Palestinians, 12 June 2018, consulted on 23 November 2021.

<sup>&</sup>lt;sup>575</sup> Confidential source, 3 February 2022.

 <sup>576</sup> B'Tselem, Fatalities: All Data, consulted 24 November 2021. See also Error! Hyperlink reference not valid.
 577 Al Jazeera, Killing of Palestinian man by Israel an 'extrajudicial execution', 24 February 2021, consulted on 23 November 2021.

aid.<sup>578</sup> The UN OHCHR in the Palestinian Territories spoke out against the extrajudicial killing of Palestinian Mohammad Salima/Sleemah, who stabbed an Orthodox Jewish man in East Jerusalem in early December 2021 and was then shot and killed by Israeli police while he was already lying on the ground.<sup>579</sup> On 14 and 15 February 2022, a Palestinian minor in Jenin and a nineteen-year-old Palestinian man in Ramallah were shot dead by Israeli forces in the West Bank. This occurred when violence broke out in Jenin between Palestinians and Israeli troops when the troops started demolishing a building. In the Ramallah case, the nineteen-year-old Palestinian man was shot by Israeli troops as Palestinian youths were throwing stones at these troops.<sup>580</sup> On 1 March 2022, a Palestinian man was shot dead by Israeli forces while being pursued near Bethlehem. It was not clear why the man was regarded as a suspect. The PA called it an execution.<sup>581</sup>

## 3.3.7 'Blood feuds and honour-related violence' 582

One source says that violence and feuds between families, often with ties to organised crime, are prevalent in the Palestinian Territories. Another source says that the PA has no control over these armed (family) groups. <sup>583</sup> According to another source, there were no known criminal cases of feuds in the Palestinian Territories. <sup>584</sup> It is not known whether Hamas has control over armed (family) groups when it involves violence and feuds between families. One source said that Hamas tries to maintain control over other groups in Gaza if it feels they could otherwise damage Hamas's position of power (see also 3.2.1). <sup>585</sup>

International organisations often speak of (sexual and) gender-based violence – (S)GBV<sup>586</sup>– of which femicide (the killing of women and girls) is an extreme form, rather than of honour-based violence. The WCLAC has said that cases of gender violence are difficult to investigate because femicide is not registered under this heading or because the cause of death is never investigated. The latter is said to happen especially in Gaza. The following information relates to various cases of SGBV.

(Sexual and) gender-based violence

<sup>&</sup>lt;sup>578</sup> Al Jazeera, Killing of Palestinian man by Israel an 'extrajudicial execution', 24 February 2021, consulted on 23 November 2021.

<sup>&</sup>lt;sup>579</sup> Facebook, UN Human Rights - Palestine, 4 December 2021, consulted on 6 December 2021; Al Jazeera, Palestinians, rights group condemn Jerusalem 'execution', 5 December 2021, consulted on 6 December 2021.

<sup>580</sup> OCHA OPt, Protection of Civilians Report 8-21 February 2022, 28 February 2022, consulted on 1 March 2022.

<sup>&</sup>lt;sup>581</sup> BBC Monitoring, *Briefing: Israeli army kills Palestinian in West Bank chase*, 2 March 2022, consulted on 3 March 2022.

<sup>582</sup> A 'blood feud' is defined as a (long-term) feud between social groups (such as families), in which various acts of violence, including murder, can take place in the context of retaliatory actions. Violence spurred on by an 'honour motive' is associated with certain social and gender norms in a community. For this reason, international organisations often include 'honour-based violence' under the heading of (sexual and) gender-based violence.

<sup>583</sup> Confidential source, 7 February 2022.

<sup>&</sup>lt;sup>584</sup> Confidential source, 22 February 2022.

 $<sup>^{\</sup>rm 585}$  Confidential source, 3 February 2022.

SGBV is an umbrella term that refers to practices based on gender norms and unequal power relations that are performed against people's will or that curtail their personal freedom. This includes physical, emotional, psychological and sexual threats and violence, such as violence in the home (domestic violence), forced marriage, female genital mutilation (FGM) and honour-related violence. Menaal Munshey, Access to SGBV Protection Services for Syrian and Palestinian Refugees in Lebanon, United Nations University Policy Report, 2018, p. 1, 6, 25; UN Women, An Assessment of GBV Services in Gaza, September 2017, p. 12; Razan Wazwaz, Oxfam Views and Voices, On the path the end violence against women: supporting civil society to combat GBV in the occupied Palestinian Territory, 18 December 2020; UNFPA, Gender-based violence, consulted on 4 March 2022. See also Error! Hyperlink reference not valid.

According to the UN Special Rapporteur on Violence against Women, the number of Palestinian women killed in the context of honour-related violence has increased in recent years. The WCLAC says that the Palestinian Central Bureau of Statistics (PCBS) conducted research into gender-based violence (GBV) in 2005, 2011 and 2019. It reported that in 2019, 29% of (currently or previously) married women have been subjected to some kind of violence by their partner. The percentage in Gaza (38%) was higher than in the West Bank (24%). Government agencies in the Family Protection Units of the Palestinian Civil Police (PCP), the Prosecution Office and the Ministry of Women's Affairs and Social Development are also said to publish data on GBV and femicide. The WCLAC attributes the high suicide rate among women and girls compared to men to 'involuntary' suicides, as they are the result of family and/or social and cultural pressure. 587 According to one source, reports of gender-based violence came increased during the corona outbreak. For a long time, organisations could only offer psychosocial help by telephone or record cases. 588

### Femicide

According to the WCLAC, nineteen women in the West Bank and eighteen women in Gaza died as a result of gender-based violence in 2019. The WCLAC recorded a total of 149 femicide cases between 2015 and 2020. The highest percentage of femicide was recorded during the lockdown periods in 2020 (25% of all cases). <sup>589</sup> In 2021, 26 cases of femicide had been registered by the end of November. <sup>590</sup> According to one source, this number had risen to 34 at the end of 2021. <sup>591</sup>

## Legislation

Various sources say that women and girls are not sufficiently protected – both legally and in practice – against sexual and gender-based violence, including honour-based violence and femicide. Women did not report abuse partly for fear of damaging their reputation. <sup>592</sup> It is said that the authorities have also failed to investigate the root cause of killings classified as femicide or gender-based violence. According to the WCLAC, this is contributing to a culture of impunity. <sup>593</sup> Palestinian women are also said to experience problems when reporting these crimes in areas under Israeli jurisdiction. <sup>594</sup> In the Palestinian Territories, religious groups are tried differently according to Personal Status Laws. For Muslims, this is done through Islamic courts and for Christians through ecclesiastical courts. <sup>595</sup>

589 WCLAC and Al-Muntada, Femicide in the Palestinian Society, April 2021, p. 3.

<sup>591</sup> Confidential source, 26 January 2022.

593 Human Rights Council, Report of the Special Rapporteur on violence against women, its causes and consequences, on her mission to the Occupied Palestinian Territory/State of Palestine, 12 July 2017, p. 6; Amnesty International, Report: State of Palestine, 2020, consulted on 15 November 2021; WCLAC and Al-Muntada, Femicide in the Palestinian Society, April 2021, p. 5.

 <sup>587</sup> Human Rights Council, Report of the Special Rapporteur on violence against women, its causes and consequences, on her mission to the Occupied Palestinian Territory/State of Palestine, 12 July 2017, p. 6; UNDP, State of Palestine: Gender Justice & the Law, 2018, p. 24; Amnesty International, Report: State of Palestine, 2020, consulted on 15 November 2021; WCLAC and Al-Muntada, Femicide in the Palestinian Society, April 2021, 3-5.
 588 Confidential source, 26 January 2022.

<sup>590</sup> Al Jazeera, Palestine: Femicide highlights need for domestic violence law, 29 November 2021, consulted on 30 November 2021.

<sup>&</sup>lt;sup>592</sup> Human Rights Council, Report of the Special Rapporteur on violence against women, its causes and consequences, on her mission to the Occupied Palestinian Territory/State of Palestine, 12 July 2017, p. 6; Amnesty International, Report: State of Palestine, 2020, consulted on 15 November 2021; WCLAC and Al-Muntada, Femicide in the Palestinian Society, April 2021, p. 5.

<sup>594</sup> Human Rights Council, Report of the Special Rapporteur on violence against women, its causes and consequences, on her mission to the Occupied Palestinian Territory/State of Palestine, 12 July 2017, p. 6; Amnesty International, Report: State of Palestine, 2020, consulted on 15 November 2021; WCLAC and Al-Muntada, Femicide in the Palestinian Society, April 2021, p. 2.

<sup>&</sup>lt;sup>595</sup> Freedom House, *Freedom in the World 2021: West Bank*, consulted on 12 November 2021

According to one source, women are discriminated against under the current legislation, such as criminal law and the Personal Status Laws. <sup>596</sup> Although an amendment to the law in 2018 (law number 5) was said to counteract the reduction in convictions for honour-related violence, sources say that this was scarcely noticeable in practice. Judges are said to have a conservative attitude towards social and cultural gender roles, according to the WCLAC. <sup>597</sup> A UNDP report states that legislative amendments from 2011 and 2018 to reduce punishment for honour-related violence were implemented in the West Bank but not in Gaza. Judges can still reduce sentences based on Articles 99 and 100 of the (Jordan) Penal Code and on Palestinian legislation according to which victims can pardon perpetrators. In the latter case, a reduction of the sentence was said to take place based on Article 99 of law number 5 of 2018. Following a 2018 amendment, however, this pardon does not apply to serious crimes such as the murder of women and children. <sup>598</sup>

According to the WCLAC, women seeking justice in the judicial system were discriminated against and often withdrew at the beginning of a court case because of social pressures or a loss of trust in the justice system. Several civil society organisations that champion women's rights, including WCLAC, have been trying for years to pass a draft law, the *Family Protection Bill*. This law is aimed at combatting violence against women and children.

According to Freedom House and Human Rights Watch, after the death of Palestinian woman Israa Ghrayeb in August 2019 more attention was paid to sexual and gender-based violence among the Palestinian population. In this particular case, three male relatives were suspected of murdering Ghrayeb. The crime was allegedly committed after Ghrayeb posted a photo of herself and her fiancé online. After a protest in Ramallah, where protesters called for an investigation into the cause of Ghrayeb's death, the PA revealed that she had died of injuries sustained during a physical attack. Under Article 330 of the Criminal Code, three persons were charged with involuntary manslaughter with a maximum sentence of five years' detention but had not yet been tried in 2020. 601 The murder of Sabreen Yasser Khweira in November 2021 also raised awareness in Palestinian society about the need for better legislation in relation to gender-based violence. Her husband, Amer Rabee, was arrested by PA police and charged with her murder. It is not known how this case has progressed. 602

## Alternative solutions

It is not clear whether there are alternative solutions to prevent violent incidents, such as murder – in 'matters of blood and honour' – and whether the authorities are also involved in developing such solutions. During the reporting period, one example came to light of an informal agreement after an incident of SGBV violence had taken place. Amnesty cites the example of Razan Moqbel, who was murdered by her fiancé in Ramallah on 28 July 2020, after which alternative arrangements were not prevented by the authorities. In this case, the husband-to-be was arrested by the

<sup>596</sup> Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>597</sup> Freedom House, Freedom in the World 2021: West Bank, consulted on 12 November 2021; WCLAC and Al-Muntada, Femicide in the Palestinian Society, April 2021, p. 2.

 <sup>&</sup>lt;sup>598</sup> UNDP, State of Palestine: Gender Justice & the Law, 2018, p. 9.
 <sup>599</sup> WCLAC and Al-Muntada, Femicide in the Palestinian Society, April 2021, p. 7.

November 2021; Confidential source, 26 January 2022.

<sup>&</sup>lt;sup>601</sup> Human Rights Watch, The Deadly Toll for Palestinian Women, 19 September 2019, consulted on 24 February 2022; Freedom House, Freedom in the World 2021: West Bank, consulted on 12 November 2021.

<sup>602</sup> Al Jazeera, Palestine: Femicide highlights need for domestic violence law, 29 November 2021, consulted on 30 November 2021.

Palestinian security services (it is not clear which) and the prosecution instigated a court action against him. On 3 August 2021, the families of the victim and the perpetrator agreed to withdraw the charges in exchange for a sum of NIS 487,000 (EUR 140,440<sup>603</sup>). According to Amnesty, the Palestinian authorities had not yet responded to this arrangement. 604

#### 3.3.8 Corruption

According to various sources, citizens regard government corruption as a major problem in the Palestinian Territories. 605 In recent years, the PA has been accused of embezzling public and development funds, cronyism and favouritism. 606 In Gaza, residents are said to doubt whether government agencies are independent. However, among respondents in Gaza the perception of corruption in Gaza was lower than that in the West Bank. 607 In a December 2021 poll, respondents said they believed there was corruption inside PA institutions (84%) as well as inside Hamas institutions (69%).608 A report by Transparency International (TI) states that 51% of Palestinians polled have little or no trust in the government, 52% trust the justice system and 59% trust the police. 609 In the PA, there an Anti-Corruption Commission that is responsible for implementing the anti-corruption strategy. The majority of the people surveyed are said to be unaware of the existence of this commission. 610 A survey by the Coalition for Accountability and Integrity (AMAN Coalition), the office of TI in the Palestinian Territories, states that 63% of the Palestinian respondents believe that the level of corruption in PA institutions had increased in 2021. Respondents referred to the system of wasta (use of personal contacts) to access public services and nepotism as the most common types of corruption. This was particularly the case in Gaza. More than half of those surveyed believe that reporting corruption puts them at risk of retaliation. They also believe that there is a lack of adequate protection for whistleblowers or witnesses. It was not clear who would be responsible for these retaliatory actions. 611

According to one source, in 2021 the PA was accused of keeping some of the corona vaccines for its own use, of concluding a controversial vaccine deal with Israel and of sending corona vaccines to Jordan although very few Palestinians had been vaccinated (see also PA critic Nizar Banat 3.1.1).612

604 Amnesty International, Report: State of Palestine, 2020, consulted on 15 November 2021.

<sup>603</sup> Exchange rate on 13 April 2022.

<sup>605</sup> Transparency International, Global Corruption Barometer Middle East & North Africa 2019 Citizens' Views and Experiences of Corruption, December 2019, p. 13-15; Freedom House, Freedom in the World 2021: West Bank, 2021, consulted on 10 January 2022.

<sup>606</sup> Al Shabaka, Corruption in Palestine: A self-enforcing system, 18 August 2015, consulted on 11 January 2022; Council on Foreign Relations, Corruption in the Palestinian Authority, 5 April 2018, consulted on 11 January 2022.

<sup>607</sup> Aman Transparency Palestine, Public Opinion Poll: The State of corruption and Anti-corruption in Palestine in 2021 Executive summary, 2021, p. 2-3.

<sup>608</sup> Palestinian Center for Policy and Survey Research, Public Opinion Poll No (82), 14 December 2021, p. 4.

<sup>&</sup>lt;sup>609</sup> Transparency International, Global Corruption Barometer Middle East & North Africa 2019 Citizens' Views and Experiences of Corruption, December 2019, p. 13-15.

610 Freedom House, Freedom in the World 2021: West Bank, 2021, consulted on 10 January 2022.

<sup>611</sup> Aman Transparency Palestine, Public Opinion Poll: The State of corruption and Anti-corruption in Palestine in 2021 Executive summary, 2021, p. 3-4.

The New York Times, Activist Dies in Palestinian Authority's Custody, Inciting Widespread Outrage, 24 June 2021, consulted on 22 February 2022; Amnesty International, Dood van Palestijnse activist in hechtenis moet transparent en effectief worden onderzocht, 25 June 2021, consulted on 22 February 2022; NPR, Palestinians dissidents rally against corruption in the Palestinian Authority, 10 December 2021, consulted on 11 January 2022.

#### 4 Refugees and displaced persons

#### 4.1 Refugee flows

Palestinians who were displaced during the 1948 and 1967 conflicts and who reside inside Israel's borders are not dealt with in this report. Displaced persons from these conflicts (and their descendants) who fled what is now Israel to the Palestinian Territories are often registered as refugees with UNRWA, but this is not always the case according to the Badil organisation. The displaced persons from these conflicts who became displaced inside the Palestinian Territories are internally displaced persons. This mostly involves Palestinians displaced by the 1967 conflict. 613 This group also includes persons who became refugees in 1948 and are therefore both refugees and internally displaced persons.<sup>614</sup> Most of the Palestinian refugees from these refugee flows who fled abroad were given shelter in Jordan (2,307,011), Lebanon (479,537) and Syria (568,730). These numbers relate to the current population, including descendants. The numbers do not include the non-refugees who are registered with UNRWA.615

Many Palestinians are registered as refugees with UNRWA. At the end of 2020, in the Palestinian Territories there were 871,537 registered refugees in the West Bank and 1,476,706 registered refugees in Gaza. This is the group that fled what is now Israel and that resides in the Palestinian Territories. This number is increasing due to the birth and registration of descendants of the original group. 616 There are nineteen official refugee camps in the West Bank and eight in Gaza. 617 The total population of the West Bank was almost 3 million in July 2021 and of Gaza almost 2.1 million (see also 2.4).

#### 4.2 Internally displaced persons

Most people were displaced in the past decade as a result of offensives on Gaza (2012, 2014 and 2021) and evictions and house demolitions in the West Bank and East Jerusalem (2016, 2019). Before that, people were also displaced by the construction of the Israeli wall around the West Bank. According to the Internal Displacement Monitoring Centre (IDMC), conflict and violence had displaced new groups in the Palestinian Territories by 2020, leaving a total of 131,000 internally displaced persons due to disasters. Natural disasters, such as wildfires and floods, created dozens of internally displaced persons in 2021.618 Most internally displaced persons are not registered and generally live among the rest of the population in the Palestinian Territories in rented housing or with relatives/host families. 619 Estimates of the number of internally displaced persons differ. The Badil organisation says that the group of persons who have been internally displaced since 1967, including their descendants in the Palestinian Territories, numbered around 344,599 people in 2018. According to the IDCM, the group of internally displaced persons numbered 243,000 people in 2019.620

<sup>613</sup> Badil, Q&A: What you need to know about Palestinian Refugees and Internally Displaced Persons, May 2011, p. 5-614 Badil, Survey of Palestinian Refugees and Internally Displaced Persons 2016-2018 Volume IX, 2018, p. 21.

<sup>615</sup> UNRWA, UNWRA in figures as of 31 December 2020, 2021, p. 1. 616 UNRWA, UNWRA in figures as of 31 December 2020, 2021, p. 1.

<sup>617</sup> UNRWA, Fields of Operations, 1 January 2017. 618 IDMC, *Palestine*, 2021, consulted on 3 January 2022.

<sup>&</sup>lt;sup>619</sup> Confidential source, 24 January 2022; Confidential source, 2 February 2022.

<sup>620</sup> IDMC, Internal Displacement in the Middle East and North Africa Region 2010-2019, p. 43; Badil, Survey of Palestinian Refugees and Internally Displaced Persons 2016-2018 Volume IX, 2018, p. 23, 48-49.

## Gaza Strip

In Gaza, around 8,250 displaced persons were said to be staying with host families or in rental properties in August 2021.<sup>621</sup> According to one source, displaced persons in Gaza are often temporarily displaced due to escalations of violence or flooding. Most of the homes destroyed in 2014 have been rebuilt. The homes destroyed by the escalation of violence in May 2021 have not yet been rebuilt. After receiving temporary shelter in UNRWA schools, people are said to be renting houses themselves or living with relatives (see also 1.3.3).<sup>622</sup>

## West Bank and East Jerusalem

317 Palestinians are believed to have been displaced by evictions in East Jerusalem in 2021.623 According to one source, people in East Jerusalem who are evicted usually cannot find a new home in Jerusalem due to the high housing/rental prices. They sometimes stay with family first, but usually end up in the West Bank or in the Kufr 'Aqab district, which officially belongs to Jerusalem but is actually behind the wall, just like the West Bank. This district is known for its poor living conditions. Also in the West Bank, people are at risk of being displaced because of evictions and the demolition of houses due to the development and expansion of construction projects, including those of Israeli settlers, or because they do not have building permits (see also Chapter 1, Evictions section). It also happens that people become temporarily displaced and are obliged to leave their homes from several days to a few months when they are close to military fire zones and the Israeli army is conducting exercises. 624 It is not known whether temporary shelter is arranged for these persons. The data in VN OCHA's Demolition and displacement in the West Bank database does not include these temporarily displaced persons. According to one source, there is a technical reason for this and it does not constitute a legal provision with regard to this group. 625 Another source says that this group does not have access to (temporary) shelter. 626

The following examples are illustrative but are by no means exhaustive. In July 2021, approximately 65 people – including 35 children – from the (sedentary) Bedouin community of Humsa al-Baqai'a in the Jordan Valley were displaced due to evictions in the West Bank and East Jerusalem by Israeli forces. A group of international aid organisations is said to have helped all the internally displaced persons in this group. 627 According to one source, some people in this group are still trying to return to their land to tend to their livestock (mostly goats and sheep). Organisations are only permitted to provide emergency aid in Area C and East Jerusalem and sustainable aid is only permitted in Area A and in approximately 1% of Area C. Internally displaced persons, including those who have been displaced multiple times, often end up in Area A, according to one source. 628 In the period from October to December 2021, dozens of people were displaced due to the destruction of houses by Israeli troops. These buildings are often built without Israeli

<sup>621</sup> OCHA, oPT Situation Report no. 9, August 2021, consulted on 7 January 2022.

<sup>622</sup> Confidential source, 2 February 2022.

<sup>623</sup> Al Jazeera, Sheikh Jarrah: Palestinian family faces forced displacement, 5 January 2021, consulted on 7 January 2022.

<sup>&</sup>lt;sup>624</sup> Confidential source, 2 February 2022; OCHA oPt, Protection of Civilians Report 8-21 February 2022, 28 February 2022, consulted on 1 March 2022.

<sup>&</sup>lt;sup>625</sup> Confidential source, 21 March 2022; OCHA oPt, Data on demolition and displacement in the West Bank, consulted on 11 April 2022. See also Error! Hyperlink reference not valid.

<sup>626</sup> Confidential source, 14 April 2022.

<sup>627</sup> Al Jazeera, Palestinian community in West Bank demolished for seventh time, 7 July 2021, consulted on 7 January 2022.

<sup>628</sup> Confidential source, 2 February 2022.

planning permission because it is difficult for Palestinians to obtain this permission (see also 3.1.4). Due to the destruction and evictions, people were forced to leave the area and thousands of people were denied access to water, farmland and schools. This happened in the Bedouin community in Hammamat al Malah-Al Meiteh in the Jordan Valley, Hebron, Nablus, Bethlehem, Khibet Atuf (area C) and Sur Bahir (area A) and in East Jerusalem. 629

# 5 Palestinians returning to their homeland

Palestinians from the Palestinian Territories who are voluntarily or forcibly returning to Gaza travel through Israel and Jordan or through Egypt. Israel has control over who returns through Israel and whether these persons are permitted to settle in the Palestinian Territories. Egypt has control over who can return to Gaza via Egypt. There were no known cases of the forced return of Palestinians to Gaza, according to one source. Gaza Israel does not recognise the right of Palestinian refugees to return to Israel inside the 1948 borders. It is not known whether there were any cases of forced or voluntary return to the West Bank.

## 5.1 Re-entry

To re-enter the Palestinian Territories, a Palestinian must have an identity card issued by Israel after 1995. With an identity card issued before 1995, the person concerned must have left the Palestinian Territories after 1995 in order to be able to return. The Israeli authorities are said to have control over whether a person can return and live in the Palestinian Territories. 631 It is not clear on which basis the Israeli authorities determine whether or not a person is permitted to settle in the Palestinian Territories after returning from abroad. Palestinians who want to return to Gaza face a number of obstacles. In addition to an identity card issued by Israel, they must also be in possession of an Israeli and Jordanian travel permit to enter Gaza via Jordan and Israel. 632 According to one source, a Palestinian who has left can also return to Gaza via the Rafah border crossing. The Israeli authorities only permit persons to return via the border crossing by which they left Gaza. The Rafah border crossing can sometimes be closed and the Egyptian authorities may refuse to allow people to cross the border for security reasons. To enter or leave Egypt as a transit country from or to Gaza, persons from Gaza do not require an Egyptian visa (see also 1.4.4). Palestinians returning to Gaza with an expired travel permit may encounter difficulties when applying for a new travel permit. 633

<sup>630</sup> Confidential source, 3 February 2022.

<sup>&</sup>lt;sup>631</sup> Immigration and Refugee Board of Canada, Israel and Palestine: Whether a Palestinian who has lived outside of the Occupied Palestinian Territory (OPT) for a number of years can return to reside in the OPT; whether the returnee requires an Israeli-issued identity card and whether it is possible to obtain the card from abroad; whether someone born in Gaza faces particular challenges returning to the OPT, 26 April 2016, consulted on 1 February 2022.

<sup>6932</sup> Immigration and Refugee Board of Canada, Israel and Palestine: Whether a Palestinian who has lived outside of the Occupied Palestinian Territory (OPT) for a number of years can return to reside in the OPT; whether the returnee requires an Israeli-issued identity card and whether it is possible to obtain the card from abroad; whether someone born in Gaza faces particular challenges returning to the OPT, 26 April 2016, consulted on 1 February 2022.

<sup>&</sup>lt;sup>633</sup> Confidential source, 24 January 2022.

# 6 Appendices

## 6.1 Public resources used

- +972 Magazine, 'To sing is not a right in Gaza', 6 June 2019, See also <a href="https://www.972mag.com/hamas-gaza-freedoms-music/">https://www.972mag.com/hamas-gaza-freedoms-music/</a>
- —, 'I'd rather die in the West Bank': LGBTQ Palestinians find no safety in Israel, 17 September 2021. See also Error! Hyperlink reference not valid.
- 7amleh, The Arab Center for the Advancement of Social Media, *The Attacks on Palestinian Digital Rights, Progress report*, May 6-19, 2021.
- ACLED, Dashboard Gaza and West Bank, 2022.
- ACTED, Gaza: ACTED's Emergency Assistance for families to meet their needs, 23 December 2021. See also Error! Hyperlink reference not valid.
- Addameer, Administrative detainees, December 2019. See also Error! Hyperlink reference not valid.
- -, Annual Report 2019, 2019.
- Key Issues: Isolation, Restrictions on family visits, July 2017. See also Error!
   Hyperlink reference not valid.
- —, Key Issues: Medical Negligence, July 2016. See also Error! Hyperlink reference not valid.
- —, Prisoners Statistics, 7 November 2021. See also Error! Hyperlink reference not valid.
- Al Haq, Field Report on Human Rights Violations in 2020, 2 March 2021.
- , National Preventive Mechanism against Torture and Ill-Treatment, 16 August 2018.
- –, Prevention of Torture: Monitoring, Documentation and means of Confrontation, 2020.
- Al Jazeera, 450 settler attacks on Palestinians in two years: Israeli NGO, 14 November 2021. See also Error! Hyperlink reference not valid.
- —, Abbas accused of power grab after Palestinian appointments, 10 February 2022.
   See also Error! Hyperlink reference not valid.
- --, Building materials allowed into Gaza after Israeli assault in May, 1 September 2021. See also Error! Hyperlink reference not valid.
- Gone but not forgotten: The case of missing Palestinians, 12 June 2018. See also Error! Hyperlink reference not valid.
- —, Israel attacks solidarity protests for escaped Palestinians, 9 September 2021.
   See also Error! Hyperlink reference not valid.
- —, Israel completes 'iron wall' underground Gaza barrier, 7 December 2021. See also Error! Hyperlink reference not valid.
- Israeli law tears Palestinian families apart, 19 July 2013. See also Error! Hyperlink reference not valid.
- —, Jerusalem: Israel forces Palestinians to self-demolish own homes, 31 January 2022. See also Error! Hyperlink reference not valid.
- Killing of Palestinian man by Israel an 'extrajudicial execution', 24 February 2021. See also Error! Hyperlink reference not valid.
- Nizar Banat family to seek international justice 'if necessary', 24 October 2021.
   See also Error! Hyperlink reference not valid.
- —, 'Love for a free life': Why Palestinian prisoners hunger strike, 10 March 2022. See also Error! Hyperlink reference not valid.
- -, Not a life: Israel keeps many Palestinians without legal status, 26 October 2021.
   See also Error! Hyperlink reference not valid.
- –, Palestine: Femicide highlights need for domestic violence law, 29 November 2021. See also Error! Hyperlink reference not valid.
- —, Palestinian Authority struggles to pay public employees, 2 December 2021. See also Error! Hyperlink reference not valid.
- –, Palestinian community in West Bank demolished for seventh time, 7 July 2021.
   See also Error! Hyperlink reference not valid.
- –, Palestinian prisoner from Gaza dies in Israeli custody, 18 November 2021. See also Error! Hyperlink reference not valid.

- —, Palestinian prisoners launch boycott of Israeli military courts, 4 January 2022.
   See also Error! Hyperlink reference not valid.
- –, Palestinians demand answers for PA shooting at student protest, 3 February 2022. See also Error! Hyperlink reference not valid.
- —, Palestinians, rights group condemn Jerusalem 'execution', 5 December 2021. See also Error! Hyperlink reference not valid.
- Sheikh Jarrah families reject 'unjust' deal with Israeli settlers, 2 November 2021.
   See also Error! Hyperlink reference not valid.
- —, Sheikh Jarrah: Palestinian family faces forced displacement, 5 January 2021. See also Error! Hyperlink reference not valid.
- —, Teen's death fuels Palestinian Authority corruption claims, 8 February 2022. See also Error! Hyperlink reference not valid.
- —, The colour coded ID system for Palestinians, 18 November 2017. See also Error! Hyperlink reference not valid.
- —, Why is the Palestinian Authority cracking down on opposition?, 25 January 2022.
   See also Error! Hyperlink reference not valid.
- Al Mezan Center for Human Rights, Fact Sheet on Obstacles to Accessing Education in the Gaza Strip, August 2021.
- Al Monitor, Bicycling program for women finds both supporters, critics in Gaza, 10 May 2021. See also Error! Hyperlink reference not valid.
- —, Gaza's female softball players in league of their own, 4 March 2020. See also Error! Hyperlink reference not valid.
- Hamas appoints first women to political bureau, 12 March 2021. See also Error!
   Hyperlink reference not valid.
- Hamas visits Gaza schools to promote hijab, 31 October 2018. See also Error!
   Hyperlink reference not valid.
- Al Muntada, Parallel Follow-up Report Prepared by the Women's Center for Legal Aid and Counselling (WCLAC) and the Palestinian Non-Governmental Organizations Forum to Combat Violence Against Women, September 2020.
- Al Shabaka, *Corruption in Palestine: A self-enforcing system*, 18 August 2015. See also Error! Hyperlink reference not valid.
- Rethinking Palestine Podcast, Palestine under Surveillance with Mona Shtaya, 25
   November 2021. See also Error! Hyperlink reference not valid.
- —, The Conditional Right to Health in Palestine, 30 June 2019. See also Error! Hyperlink reference not valid.
- —, The Palestinian Security Sector: Entrenching State Repression, 14 November 2021. See also Error! Hyperlink reference not valid.
- —, The political marginalization of Palestinian women in the West Bank. 28 July 2019. See also Error! Hyperlink reference not valid.
- Aman Transparency Palestine, *Public Opinion Poll: The State of corruption and Anti-*corruption in *Palestine in 2021 Executive summary*, 2021.
- Amnesty International, *Dood van Palestijnse activist in hechtenis moet transparent* en effectief worden onderzocht, 25 June 2021. See also Error! Hyperlink reference not valid.
- —, Neither dead nor alive, August 2018. See also Error! Hyperlink reference not valid.
- -, Palestine (State of) 2020, 2021. See also Error! Hyperlink reference not valid.
- —, Palestine (State of) 2021, 2022. See also Error! Hyperlink reference not valid.
- Anera, Health Care System in Palestine, 21 September 2020. See also Error! Hyperlink reference not valid.
- AP News, Report: NSO spyware found on 6 Palestinians activists' phones, 8 November 2021. See also Error! Hyperlink reference not valid.
- —, The deepening costs of 4 Gaza wars in 13 years, in 4 charts, 25 August 2021. See also Error! Hyperlink reference not valid.
- Arab News, *Gaza police block women from attending football match*, 28 January 2018. See also Error! Hyperlink reference not valid.
- Artikel XIV, Israeli-Palestinian Interim Agreement on the West Bank and the Gaza Strip, 28 September 1995.
- B'Tselem and Palestinian Centre for Human Rights, *Unwilling and Unable: Israel's Whitewashed Investigations of the Great March of Return Protests*, December 2021.
- —, Summary of joint report with PCHR, December 2021.

- B'Tselem, Death Penalty in the Palestinian Authority and Under Hamas Control, 16 October 2013. See also Error! Hyperlink reference not valid.
- -, East Jerusalem, 27 January 2019. See also Error! Hyperlink reference not valid.
- —, Fatalities: All data. See also Error! Hyperlink reference not valid.
- —, Minors in Jeopardy, Violations of the Rights of Palestinian Minors by Israel's Military Courts, March 2018. See also Error! Hyperlink reference not valid.
- -, Restrictions on Movement, 11 November 2017. See also Error! Hyperlink reference not
- —, State Business: Israel's misappropriation of land in the West Bank through settler violence, November 2021.
- —, Statistics on Palestinians in the custody of the Israeli security forces, 24 November 2021. See also Error! Hyperlink reference not valid.
- —, Statistics on the death penalty in the Palestinian Authority and under Hamas control in Gaza, 13 September 2021. See also Error! Hyperlink reference not valid.
- -, The Military Courts, 11 November 2017. See also Error! Hyperlink reference not valid.
- —, The Separation Barrier, 11 November 2017. See also https://www.btselem.org/separation\_barrier
- -, West Bank roads on which Israel forbids Palestinian vehicles, 31 January 2017. See also Error! Hyperlink reference not valid.
- Badil, Q&A: What you need to know about Palestinian Refugees and Internally Displaced Persons, May 2011.
- , Survey of Palestinian Refugees and Internally Displaced Persons 2016-2018 Volume IX, 2018.
- BBC Monitoring, Briefing: Israel limits entry from West Bank for Ramadan prayers, 6 April 2022.
- -, Briefing: Israel votes to re-ban Palestinian family unification, 11 March 2022.
- —, Briefing: Israeli army kills Palestinian in West Bank chase, 2 March 2022.
- -, Briefing: Israeli officer kills Palestinian after reported stabbing attempt, 12 April
- —, Briefing: Palestinian woman killed as surge of violence continues, 10 April 2022.
- -, Briefing: Surge in settler violence in West Bank follows deadly attack, 31 March 2022.
- —, Briefing: Tensions flare in East Jerusalem at start of Ramadan, 3 April 2022.
- —, Egypt resumes opening Rafah crossing with Gaza, 6 December 2021.
- -, Hamas warns of 'escalation' in Gaza, blaming Egypt, 7 December 2021.
- -, IS claims responsibility for deadly Israel attack, 28 March 2022.
- -, Israeli police shoot Palestinian during suspected stabbing attack, 19 March 2022.
- -, Israeli president's visit to West Bank shrine draws protest, 29 November 2021.
- -, Israeli troops kill five Palestinians in West Bank clashes, 14 April 2022.
- -, Palestinian death toll rises to five in occupied West Bank, 14 April 2022.
- -, Palestinian girl held in stabbing of settler in Jerusalem, 8 December 2021.
- —, Palestinian killed as Israeli forces raid Jenin refugee camp, 9 April 2022.
- -, Palestinian killed in Hebron as clashes erupt in West Bank, 1 April 2022.
- –, Palestinian local elections in West Bank, 11 December 2021.
- -, Palestinian officers extricate Israelis attacked in Ramallah, 2 December 2021.
- —, Palestinian president opens Central Council session, 7 February 2022.
- —, Palestinian prisoners on hunger strike reach release deal, 23 November 2021.
- —, Palestinians hold second round of local elections, 27 March 2022.
- —, Palestinians rally behind prisoners on hunger strike in Israel, 5 November 2021.
- —, Three Palestinians wounded by Israeli tank fire on Gaza, 29 December 2021.
- -, Two Israeli teens arrested over stoning attack in Sheikh Jarrah, 11 December 2021.
- BBC News, Palestinians vent against Hamas in rare online event, 29 January 2022. See also Error! Hyperlink reference not valid.
- -, Sheikh Jarrah: Why could Palestinians lose their homes in Jerusalem, 3 August 2021. See also Error! Hyperlink reference not valid.
- -, Two Palestinians killed in gunfight during Israeli West Bank raid, 1 April 2022. See also Error! Hyperlink reference not valid.
- Biometric update, Biometrics shipment to Palestinian Authority mysteriously released after being confiscated, 22 January 2021. See also Error! Hyperlink reference not valid.

- CIA The World Factbook, *Gaza Strip*, 12 April 2022. See also Error! Hyperlink reference not valid.
- -, West Bank, 12 April 2022. See also Error! Hyperlink reference not valid.
- Council on Foreign Relations, *Corruption in the Palestinian Authority*, 5 April 2018. See also Error! Hyperlink reference not valid.
- —, Global Conflict Tracker: Israeli-Palestinian Conflict, 7 January 2022. See also Error! Hyperlink reference not valid.
- Danish National ID Centre, *Jordan: Issuance of Jordanian documents to stateless Palestinians*, 16 April 2020.
- Daraj News, Mustafa Ibrahim, *Gaza: Yellow Passports, Blue Cards and Lost Rights*, 7 October 2021. See also Error! Hyperlink reference not valid.
- DCAF, Palestinian Women and Penal Law: Policy Brief Ramallah and Geneva, May 2012.
- Defense for Children International Palestine (DCIP), *Child Recruitment*, 2022. See also Error! Hyperlink reference not valid.
- Israeli forces raid DCIP office, confiscate computers and client files, 29 July 2021.
   See also Error! Hyperlink reference not valid.
- —, DCIP, Juvenile Justice, 2022. See also Error! Hyperlink reference not valid.
- —, Military detention: Number of Palestinian Children (12-17) in Israeli Administrative Detention, 2 October 2020. See also Error! Hyperlink reference not valid.
- –, Military detention: Number of Palestinian Girls in Israeli Military Detention, 2
   October 2020. See also Error! Hyperlink reference not valid.
- —, No way to treat a child: Palestinian children in the Israeli military detention system, April 2016.
- EASO, COI Query Palestine, 10 January 2020.
- European Council on Foreign Relations (ECFR), *Mapping Palestinian Politics*, 2022. See also Error! Hyperlink reference not valid.
- EO, Van Nablus naar Ninevé: Aflevering 1 De Samaritanen (Westelijke Jordaanoever), 2 July 2018. See also Error! Hyperlink reference not valid.
- Euro-Mediterranean, Human Rights Monitor, The Gaza Strip: Undocumented Citizens, March 2021.
- European Union, *Palestine Factsheet*, 23 September 2021. See also **Error! Hyperlink** reference not valid.
- Facebook, UN Human Rights Palestine, 4 December 2021. See also <a href="https://www.facebook.com/permalink.php?story-fbid=pfbid0qNRnkviZVCuFBjvyY-5483Mvzf1B8o6rxSKyPgm3VL3FdZ4E7vRNWcuVn3VK5TZvKl&id=5780144290535-53&cft-[0]=AZUIOZlpQlrcYqO-</a>
  - 4e5uZ1zPCxuvd1hDsMIitVUNGwIqBO9tfXkpqZx2gl aTS0WdIgLuGj3c7rJD0lcrZuP MZ4FBUZ0xa3VfO4sDTIdaszXl1FWOecHGwN9OngpjKUj8mG3PUl8pzgZH2ptKDGR 2ge7q9ePyh4DqhrTtQoE9jLHIQ& tn =%2CO%2CP-R
- Francesca P. Albanese and Lex Takkenberg, *Palestinian refugees in international law*, Oxford University Press: 2020.
- Freedom House, *Gaza Strip: Freedom in the World 2020 Country Report*, 2020. See also Error! Hyperlink reference not valid.
- —, Gaza Strip: Freedom in the World 2021 Country Report, 2021. See also Error! Hyperlink reference not valid.
- —, West Bank Freedom in the World 2021 Country report, 2021. See also Error!
   Hyperlink reference not valid.
- —, West Bank: Freedom in the World 2022 Country Report, 2022. See also Error!
   Hyperlink reference not valid.
- Gisha, Exits by Palestinians via Erez Crossing to Israel, the West Bank, and abroad, 30 June 2021. See also Error! Hyperlink reference not valid.
- -, Gaza Up Close, 1 September 2021. See also Error! Hyperlink reference not valid.
- Israel continues to restrict movement to and from Gaza, harming residents, 23
   May 2021. See also Error! Hyperlink reference not valid.
- —, One year of "coronavirus closure" at Erez crossing, 18 March 2021. See also Error! Hyperlink reference not valid.
- —, The illusion of disengagement, 12 September 2019. See also Error! Hyperlink reference not valid.
- Unemployment rate in Gaza at 44,7% in Q2 of 2021, 21 October 2021. See also Error! Hyperlink reference not valid.

- Haaretz, Bennett rejects Peace Process with Abbas, focuses on Palestinians' practical needs, 1 September 2021. See also Error! Hyperlink reference not valid.
- —, East Jerusalem residents will be able to renew travel documents at any other Interior Ministry offices, 14 May 2018. See also Error! Hyperlink reference not valid.
- –, Months later, Israeli military backlog holds up probe into dozens of Gaza border deaths, 29 June 2021. See also Error! Hyperlink reference not valid.
- —, Pride and Prejudice: The Hellish Life of Gaza's LGBTQ Community, 25 June 2019.
   See also Error! Hyperlink reference not valid.
- Shots fired at Israeli forces during West Bank protest in support of escaped prisoners, 8 September 2021. See also Error! Hyperlink reference not valid.
- —, Who are East Jerusalem's permanent residents, 10 April 2018. See also Error!
   Hyperlink reference not valid.
- —, Who would you be allowed to marry in Israel today?, 10 April 2018. See also Error! Hyperlink reference not valid.
- Hamoked, Childhood in Chains: The detention and interrogation of Palestinian Teenagers in the West Bank, April 2018.
- —, Under Cover of Darkness: Night arrests of Palestinian minors, by Israeli Security Forces in the West Bank, October 2020.
- Home Office, Report of a Home Office Fact-Finding Mission Occupied Palestinian Territories: freedom of movement, security and human rights situation, March 2020.
- Human Rights Watch, *Abusive System: Failures of Criminal Justice in Gaza*, September 2012. See also Error! Hyperlink reference not valid.
- —, Israel and Palestine: Events of 2021, 2022. See also Error! Hyperlink reference not valid.
- –, Palestine: No let-up in arbitrary arrests, torture, 29 May 2019. See also Error!
   Hyperlink reference not valid.
- The Deadly Toll for Palestinian Women, 19 September 2019. See also Error!
   Hyperlink reference not valid.
- Two Authorities, One Way, Zero Dissent: Arbitrary Arrest and Torture under the Palestinian Authority and Hamas, 23 October 2018. See also Error! Hyperlink reference not valid.
- Spyware used to hack Palestinian Rights Defenders, 8 November 2021. See also Error! Hyperlink reference not valid.
- —, West Bank: Free Suspected Blogger, 5 December 2010. See also Error! Hyperlink reference not valid.
- ICHR, The Report of the Independent Commission for Human Rights (ICHR) On The Initial Report of the State of Palestine Presented to The United Nations Committee on The Elimination of All Forms of Discrimination against Women, July 2018.
- —, The Status of Human Rights in Palestine Executive Summary 2020, 2021. International Displacement Monitoring Centre (IDMC), Internal Displacement in the Middle East and North Africa Region 2010-2019, 2020.
- —, Palestine, 2021. See also Error! Hyperlink reference not valid.
- Immigration and Refugee Board of Canada, Query response on Israel and Palestine: Travel documents issued by the Israeli government to residents of the West Bank and Gaza Strip (2017 March 2020), 27 April 2020. See also Error! Hyperlink reference not valid.
- —, Israel and Palestine: Whether a Palestinian who has lived outside of the Occupied Palestinian Territory (OPT) for a number of years can return to reside in the OPT; whether the returnee requires an Israeli-issued identity card and whether it is possible to obtain the card from abroad; whether someone born in Gaza faces particular challenges returning to the OPT, 26 April 2016. See also Error! Hyperlink reference not valid.
- —, Palestine: Procedures to obtain a Palestinian passport in the West Bank; the documents to be submitted and whether the Palestinian Authority requires original identity documents or accepts photocopies; whether the passport applicant has to appear in person to apply for or receive a passport; the features of a passport; the name of the signature on page three of the passport; the duration between the application for and the reception of a passport; whether a person may have two valid passports that were issued on different dates; the prevalence of fraudulent Palestinian passports (January 2008 April 2010), 30 April 2010. See also Error! Hyperlink reference not valid.

- International Labour Organisation (ILO), The situation of workers of the occupied Arab territories, Report of the Director-General Appendix 2021, 2021.
- International Planned Parenthood Federation, *Palestinian Family Planning and Protection Association (PFPPA)*. See also Error! Hyperlink reference not valid.
- International Trade Union Confederation (ITUC), Workers Rights in Crisis: Palestinian workers in Israel and the settlements, 12 April 2021.
- Islamic Relief, *International charity launches campaign to change lives of Palestinian orphans*, 5 February 2018. See also Error! Hyperlink reference not valid.
- Israel Ministry of Foreign Affairs, *Acquisition of Israeli Nationality*, 1 January 2010. See also Error! Hyperlink reference not valid.
- KavLaoved, Policy brief January 2022: Who is protecting the rights of Palestinian workers in Israel, January 2022.
- Lawyers for Justice, *Trial postponed for those accused of killing activist Nizar Banat*, 13 February 2022.
- —, Trial postponed for those accused of killing activist Nizar Banat, 11 April 2022. Luna Saadeh, 'The Israeli Occupation and Palestinians' Right to Choice in Marriage' in Kohl Journal Vol 4 (2): 2018.
- Menaal Munshey, Access to SGBV Protection Services for Syrian and Palestinian Refugees in Lebanon, United Nations University Policy Report, 2018.
- Middle East Eye, 'The Coordinator': Israel instructs Palestinians to download app that tracks their phones, 8 April 2020. See also Error! Hyperlink reference not valid.
- —, Israel: Lawyers call on Shit Bet to halt text message threats targeting Palestinians, 13 May 2021. See also Error! Hyperlink reference not valid.
- —, Palestinian prisoners to go on hunger strike against Israel's punitive measures, 1
   March 2022. See also Error! Hyperlink reference not valid.
- Middle East Institute, After 18 years, Palestinian families can finally live together: The end of Israel's 2003 citizenship law, 10 August 2021. See also Error! Hyperlink reference not valid.
- Middle East Monitor, *US backs down from reopening consulate in occupied East Jerusalem*, 16 December 2021. See also Error! Hyperlink reference not valid.
- —, PA resume economic dialogue after 5-year pause, 15 December 2021. See also Error! Hyperlink reference not valid.
- MIFTAH, Booklet on Institutions Providing Emergency Protection Services to Battered Women: West Bank and Gaza Strip, December 2015.
- Military Court Watch, *The UNICEF Report 9 years on*, 31 March 2022. See also Error! Hyperlink reference not valid.
- Ministry of Foreign Affairs, Letter to Parliament: Ontwikkelingen in Jeruzalem, Gaza en Israël, 17 May 2021. See also Error! Hyperlink reference not valid.
- Ministry of Foreign Affairs, *Thematisch Ambtsbericht Palestijnen in Libanon*, January 2021.
- Musawah, Thematic report on Muslim family law and Muslim women's rights in Palestine, July 2018.
- The Palestinian National Authority Ministry of Women's Affairs. *National Strategy to Combat Violence Against Women 2011-2019*, 2011.
- Mutaz M. Qafisheh, 'Who has the right to become a Palestinian Citizen? An International law analysis' in *Yearbook of Islamic and Middle Eastern Law Online*, June 2017.
- Nederlands Dagblad, *Meer christenen uit Gaza welkom in Jeruzalem*, 15 April 2022. See also Error! Hyperlink reference not valid.
- Norwegian Refugee Council, 10 things you should know about the evictions in East Jerusalem, 11 November 2021. See also Error! Hyperlink reference not valid.
- Norwegian Refugee Council, *Undocumented and Stateless: The Palestinian Population Registry and Access to Residency and Identity Documents in the Gaza Strip*, January 2012.
- NOS, Geweld bij al-Aqsamoskee Oost-Jeruzalem: ruim 150 gewonden gemeld, 15 April 2022. See also Error! Hyperlink reference not valid.
- —, Israëlische ondergrondse muur langs Gazastrook na bijna vier jaar voltooid, 7
   December 2021. See also Error! Hyperlink reference not valid.
- NPR, Hamas: Gaza women learning to drive must have a chaperone, 1 June 2016. See also Error! Hyperlink reference not valid.

- NPR, Palestinians dissidents rally against corruption in the Palestinian Authority, 10 December 2021. See also Error! Hyperlink reference not valid.
- NRC, Israël zet Palestijnse sociale organisaties op de terreurlijst, 25 October 2021. See also Error! Hyperlink reference not valid.
- OCHA, Data on casualties. See also See also Error! Hyperlink reference not valid.
- —, Data on demolition and displacement in the West Bank, 12 January 2021.
- Gaza Strip Access and Movement, October 2020. See also Error! Hyperlink reference not valid.
- -, Gaza Strip: Escalation of hostilities, 21 May 2021.
- —, GAZA: Internally Displaced people, 20 May 2021. See also Error! Hyperlink reference not valid.
- -, Highlights from the reporting period 18 21 maart 2022, 25 March 2022.
- -, Humanitarian needs overview OPT, December 2021.
- -, Humanitarian Response Plan: Working Document, December 2021.
- Longstanding access restrictions continue to undermine the living conditions of West Bank Palestinians, 8 June 2020.
- –, Most Palestinian plans to build in Area C not approved, 22 June 2021. See also Error! Hyperlink reference not valid.
- —, Overview November 2021, 3 November 2021. See also https://www.ochaopt.org/content/overview-november-2021
- —, Palestinian strive to access water in the Jordan Valley, 22 June 2021. See also Error! Hyperlink reference not valid.
- -, Protection of civilians report 5 8 October 2021, October 2021.
- -, Protection of civilians report 9 October 1 November 2021, November 2021
- -, Protection of Civilians Report 16 November 6 December 2021, December 2021.
- -, Protection of Civilians Report 7-20 December 2021, December 2021.
- —, Protection of Civilians Report 8-21 February 2022, 28 February 2022.
- -, Protection of Civilians Report 22 February 7 March 2022, 11 March 2022.
- -, Protection of Civilians Report 22 March 4 April 2022, 11 April 2022.
- -, Response to the escalation in the oPt, Situation Report No. 6, July 2021.
- -, Response to the escalation in the oPt Situation Report No. 9, August 2021.
- -, Response to the escalation in the oPt Situation Report No. 10, September 2021.
- —, Response to the escalation in the oPt Situation Report 19 Oct 1 Nov 2021, November 2021.
- —, West Bank access restrictions, July 2020. See also Error! Hyperlink reference not valid.
- —, UN experts alarmed by rise in settler violence in occupied Palestinian territory, 10 November 2021. See also Error! Hyperlink reference not valid.
- OHCHR, The Contribution of the State of Palestine to the Special Rapporteur's Thematic Report on Racism, Racial Discrimination, and Xenophobia in the Contest of Laws, Policies and Practices Relating to Citizenship, Nationality and Immigration. See also Error! Hyperlink reference not valid.
- Open Democracy, *Israel's digital apartheid is silencing Palestinians*, 20 May 2021. See also Error! Hyperlink reference not valid.
- oPt Education Cluster, *Background on schools under threat of demolition in West Bank*, December 2020.
- Oxford Analytica, *Palestinian Hamas may respond to Gaza aid delays*, 8 December 2021. See also Error! Hyperlink reference not valid.
- Palestine Children's Relief Fund (PCRF), Annual report 2020, 2020.
- Palestine Food Security Sector & Palestinian Central Bureau of Statistics, Socio-Economic & Food Security Survey 2020: State of Palestine, 2020.
- Palestinian Center for Policy and Survey Research, *Public Opinion Poll No (82)*, 14 December 2021.
- Palestinian Central Bureau of Statistics, *PCBS: Labour Force Survey, Q4 2020*, 2020.
- Palestinian Centre for Human Rights, *IOF Raid Yabous Cultural Centre and Edward Said National Conservatory of Music in Occupied East Jerusalem*, 23 July 2020. See also Error! Hyperlink reference not valid.
- Palestinian Centre for Human Rights, On World Day against the death penalty, death penalty is not a solution, only an illusion of deterrence to avoid liability, 10 October 2020. See also Error! Hyperlink reference not valid.

- Palestinian Police, Family Protection Unit and Juvenile Unit. See also Error! Hyperlink reference not valid.
- PAX, Executive Summary Discrimination Combating Mechanisms: Freedom of Religion and Belief in Palestine, 2020.
- PCATI, Documenting torture while providing legal aid: a handbook for lawyers, June 2016
- Razan Wazwaz, Oxfam Views and Voices, On the path the end violence against women: supporting civil society to combat GBV in the occupied Palestinian Territory, 18 December 2020. See also Error! Hyperlink reference not valid.
- Refugee Legal Aid Information, *Occupied Palestinian Territory LGBTI Resources*. See also Error! Hyperlink reference not valid.
- Reporters Without Borders, At least 35 journalists attacked while covering West Bank protests, 5 July 2021. See also Error! Hyperlink reference not valid.
- —, Israel now holding 13 Palestinian journalists, 28 May 2021. See also Error!
   Hyperlink reference not valid.
- —, Palestine: Harassed journalists, 2021. See also Error! Hyperlink reference not valid.
- —, WhatsApp blocks accounts of at least seven Gaza Strip journalists, 7 July 2021. See also Error! Hyperlink reference not valid.
- Reuters, Israel-Gaza conflict rages on despite U.S. regional diplomacy, 17 May 2021. See also Error! Hyperlink reference not valid.
- Israeli electric company ends power cuts to West Bank after Palestinians pay debt, 22 January 2020. See also Error! Hyperlink reference not valid.
- —, Palestinian Authority cut wages in prisoner, tax dispute with Israel, 6 December 2021. See also Error! Hyperlink reference not valid.
- —, With most Gaza homes wrecked by war still in ruins, smiles for the lucky few, 13
  January 2022. See also Error! Hyperlink reference not valid.
- Rijksoverheid, Nederland ondersteunt vredesproces Midden-Oosten. See also Error! Hyperlink reference not valid.
- Save the Children, Hope under the rubble: The impact of Israel's home demolition policy on Palestinian children and their families, 2021.
- SOS Children's Villages Palestine, Annual Report 2020, 2020.
- The Association for Civil Rights in Israel, *Sheikh Jarrah: Discrimination, Evictions and Dispossession under the auspices of the law*, June 2021. See also Error! Hyperlink reference not valid.
- The Danish Immigration Service, *Palestinian Refugees: Access to registration and UNRWA services, document and entry to Jordan*, June 2020.
- The Guardian, *Israel under pressure to conclude flawed case against aid worker*, 28 February 2022. See also **Error! Hyperlink reference not valid.**
- —, Israeli police in standoff with Palestinians over Sheikh Jarrah eviction, 17 January 2022. See also Error! Hyperlink reference not valid.
- –, Musicians decry Hamas ban on co-ed school concerts in Gaza, 22 December 2019. See also Error! Hyperlink reference not valid.
- Palestinian baby dies after treatment delayed by Israeli blockade of Gaza, 1 April 2022. See also Error! Hyperlink reference not valid.
- UN Palestine refugee aid agency 'close to collapse' after funding cuts, 5
  November 2021. See also Error! Hyperlink reference not valid.
- The Independent, AP photographer beaten by Israeli police in east Jerusalem, 17 December 2021. See also Error! Hyperlink reference not valid.
- —, Defiant in war and isolation, Hamas plays long game in Gaza, 20 December 2021. See also Error! Hyperlink reference not valid.
- —, Gaza's children resilient or traumatized, 15 January 2022. See also Error!
   Hyperlink reference not valid.
- —, Gaza's old battery pileups pose risk to health, environment, 22 December 2021. See also Error! Hyperlink reference not valid.
- Hamas: Qatar paying public salaries by sending fuel to Gaza, 30 November 2021.
   See also Error! Hyperlink reference not valid.
- -, Israel approves measures for Palestinians after leaders meet, 29 December 2022. See also Error! Hyperlink reference not valid.
- —, Israel return ex-prisoner weeks after ending hunger strike, 5 December 2021.
   See also Error! Hyperlink reference not valid.

- Israel to give Gaza Christians permits for Christmas holiday, 24 November 2021.
   See also Error! Hyperlink reference not valid.
- —, Israel upholds use of surveillance technology on protesters, 2 February 2022.
   See also Error! Hyperlink reference not valid.
- Israel's security forces are 'complicit' in drastic surge in settler violence, report finds, 21 July 2021. See also Error! Hyperlink reference not valid.
- —, Palestinians freed after hunger strikes have lifelong damage, 10 December 2021.
   See also Error! Hyperlink reference not valid.
- —, Palestinians say dozens of cars wrecked in settler vandalism, 9 November 2021.
   See also Error! Hyperlink reference not valid.
- Rubble brings opportunity, and risk, in war-scarred Gaza, 13 January 2022. See also Error! Hyperlink reference not valid.
- –, Some 400,000 Palestinians vote in rare municipal elections, 11 December 2021.
   See also Error! Hyperlink reference not valid.
- Some Palestinians get legal status after years in Gaza limbo, 10 January 2022.
   See also Error! Hyperlink reference not valid.
- —, UN agency for Palestinian refugees faces funding crisis, 30 November 2021. See also Error! Hyperlink reference not valid.
- The Jerusalem Post, *Palestinian workers in Israel: Opportunity or exploitation?*, 1 August 2021. See also Error! Hyperlink reference not valid.
- —, World Bank: Israel must give Palestinians 4G so economy can thrive, 9 February 2022. See also Error! Hyperlink reference not valid.
- The New Arab, *Palestinian Authority outlaws child marriage in occupied West Bank*, 7 November 2019. See also Error! Hyperlink reference not valid.
- The New York Times, Activist Dies in Palestinian Authority's Custody, Inciting Widespread Outrage, 24 June 2021. See also Error! Hyperlink reference not valid.
- Critic's Death Puts Focus on Palestinian Authority's Authoritarianism, 7 July 2021.
   See also Error! Hyperlink reference not valid.
- -, Latest Attack In Israel Kills At Least Two In Tel Aviv, 8 April 2022.
- —, Palestinian American died handcuffed in Israeli custody, witnesses say, 20
   January 2022. See also Error! Hyperlink reference not valid.
- —, Palestinian Authority Bans Activities by Gay Rights Group, 19 August 2019. See also Error! Hyperlink reference not valid.
- —, Palestinian Gunman Kills 5 in Israel's Fifth Attack in Recent Days, 29 March 2022. See also Error! Hyperlink reference not valid.
- –, Palestinian Prisoner's Hunger Strike Leads to Deal for Release, 4 January 2022.
   See also Error! Hyperlink reference not valid.
- The Representative Office of Norway to the Palestinian Authority, *Information to Gaza Citizens*. See also Error! Hyperlink reference not valid.
- The Times of Israel, Israel to authorize 2,000 more entry permits for Gaza workers, 10 March 2022. See also <a href="https://www.timesofisrael.com/liveblog\_entry/israel-to-authorize-2000-more-entry-permits-for-gaza-workers/#:~:text=Israel%20is%20set%20to%20issue,in%20the%20country%20to%2012%2C000">https://www.timesofisrael.com/liveblog\_entry/israel-to-authorize-2000-more-entry-permits-for-gaza-workers/#:~:text=Israel%20is%20set%20to%20issue,in%20the%20country%20to%2012%2C000</a>.
- Israeli electrical company to cut power to West Bank Palestinians over debts, 28
   October 2021. See also Error! Hyperlink reference not valid.
- Some Palestinians get legal status after years in Gaza limbo, 10 January 2022.
   See also Error! Hyperlink reference not valid.
- New 'Citizenship Law' advances, months after ban on Palestinian spouses lapsed,
   January 2022. See also Error! Hyperlink reference not valid.
- The Washington Post, Israel escalates surveillance of Palestinians with facial recognition program in West Bank, 8 November 2021. See also Error! Hyperlink reference not valid.
- Tone Anderson, IKON, When the boys return, 2012.
- Transparency International, Global Corruption Barometer Middle East & North Africa 2019 Citizens' Views and Experiences of Corruption, December 2019.
- U.S. Embassy in Israel, Marriage.
- UN, OHCHR Condemns in the strongest terms the execution of three men in Gaza UNOG Press Briefing Note (Excerpts), 7 April 2017.

- —, Two Years On: People injured and traumatized during the "Great March of Return" are still struggling, 6 April 2020. See also Error! Hyperlink reference not valid.
- UN, EU and UN Human Rights Office in Palestine mark the World Day Against the Death Penalty - Press Release, 10 October 2021. See also Error! Hyperlink reference not valid.
- UNDP, Everyday hero: Access to justice for Palestinian women, 4 December 2018. See also Error! Hyperlink reference not valid.
- UNDP, State of Palestine: Gender Justice & the Law, 2018.
- UNFPA, Gender-based violence. See also Error! Hyperlink reference not valid.
- UNFPA, Seeking protection: Survivors of sexual violence and their access to services in Palestine, February 2020.
- UN General Assembly, Economic costs of the Israeli occupation for the Palestinian people: poverty in the West Bank between 2000 and 2019, 30 August 2021.
- UNHCR, Country of Origin Information on the Situation in the Gaza Strip, Including on Restrictions on Exit and Return, 2018.
- Ensuring accountability and justice for all violations of international law in the Occupied Palestinian Territory, including East Jerusalem, 15 February 2021.
- —, UNHCR position on returns to Gaza, March 2022.
- UN Human Rights Council, Report of the Special Rapporteur on violence against women, its causes and consequences, on her mission to the Occupied Palestinian Territory/State of Palestine, 12 July 2017.
- UNICEF, State of Palestine Country Report on out-of-school children, July 2018. UN News, UNRWA seeks \$1.6 billion to support Palestinian refugees in 2022, 18 January 2021. See also Error! Hyperlink reference not valid.
- UNRWA, Consolidated Eligibility and Registration Instructions (CERI), 2006.
- -, Dashboard, UNRWA-mandaatgebied. See also Error! Hyperlink reference not valid.
- -, Demolition Watch. See also Error! Hyperlink reference not valid.
- -, Eligibility and registration Q&A, 2020.
- —, Emergency appeal, emergency appeal for the occupied Palestinian territory. 2 March 2021.
- -, Fields of Operations, 1 January 2017. See also Error! Hyperlink reference not valid.
- -, Gaza situation report 87, 9 April 2015.
- Letter from UNRWA to UNHCR describing UNRWA's mandate and services, 22 September 2021.
- —, NGO partners. See also Error! Hyperlink reference not valid.
- Regional Information Relating to Women's Nationality. See also Error! Hyperlink reference not valid.
- -, UNWRA in figures as of 31 December 2020, 2021.
- -, Were We Work: Gaza Strip. See also Error! Hyperlink reference not valid.
- -, What we do: Microfinance. See also Error! Hyperlink reference not valid.
- —, Working at UNRWA. See also Error! Hyperlink reference not valid.
- UN Women, An Assessment of GBV Services in Gaza, September 2017.
- USDoS, Israel (includes West Bank and Gaza) 2020 International Religious Freedom Report, 2021.
- –, West bank and Gaza 2020 Human Rights Report, 2021.
- WCLAC and Al-Muntada, Femicide in the Palestinian Society, April 2021.
- WCLAC, Access to justice for women victims of violence in Palestine, 2021.
- West Bank Protection Consortium, *Palestinian access to water& attacks on WASH structures in Area C*, April 2021.
- Women's Learning Partnership, *Palestine (Gaza) Penal code, British Mandate Criminal code*. See also Error! Hyperlink reference not valid.
- World Bank Group, *Economic Monitoring Report to the Ad Hoc Liaison Committee*, 17 November 2021.

## 6.2 List of abbreviations

AMAN - Coalition for Accountability and Integrity

AI/Amnesty - Amnesty International

CEDAW – Convention on the Elimination of All Forms of Discrimination against Women

CLA - Coordination Liaison Administration

COGAT - Israeli Coordinator of Government Activities in the Territories

DCIP - Defense for Children International Palestine

ECL - Electronic Crimes Law

FPU - Family Protection Units

FH - Freedom House

GAPAR - General Administration for Palestine Arab Refugees

HRW - Human Rights Watch

ICRC - International Committee of the Red Cross

IDF - Israeli Defence Forces

IDMC - Internal Displacement Monitoring Centre

IHCR - Independent Commission for Human Rights

ILO - International Labour Organisation

IOM - International Organisation for Migration

IPS - Israeli Prison Service

LGBTI - Lesbian, Homosexual, Bisexual, Transgender, Intersexual

MoWA - Ministry of Women's Affairs

OCHA - United Nations Office for the Coordination of Humanitarian Affairs

PASF - Palestinian Authority Special Forces

PCAC - Palestinian Civil Affairs Committee

PCBS - Palestinian Central Bureau of Statistics

PCHR - Palestinian Centre for Human Rights

PSL - Personal Status Laws

PLO - Palestinian Liberation Organisation (Palestijnse Bevrijdingsorganisatie)

PPS - Palestinian Prisoners Society

RSF - World Press Freedom Index van Reporters Without Borders

SGBV - Sexual and Gender Based Violence

TI - Transparency International

UNDP - United Nations Development Programme

UNHCR - United Nations High Commissioner for Refugees

UNICEF - United Nations Children's Emergency Fund

UNRWA – United Nations Relief Works Agency for Palestinian Refugees

USDoS- United States Department of State

**UN - United Nations** 

WCLAC - Women's Centre for Legal Aid and Counselling

# 6.3 Map of the Palestinian Territories



Source: Managementboek.nl